### **Counting Bases**

by

Kerri Pyper Webb

A thesis presented to the University of Waterloo in fulfilment of the thesis requirement for the degree of Doctor of Philosophy in

Combinatorics and Optimization

Waterloo, Ontario, Canada, 2004

©Kerri Webb 2004

I hereby declare that I am the sole author of this thesis. This is a true copy of the thesis, including any required final revisions, as accepted by my readers.

I understand that my thesis may be made electronically available to the public.

#### Abstract

A theorem of Edmonds characterizes when a pair of matroids has a common basis. Enumerating the common bases of a pair of matroid is a much harder problem, and includes the #P-complete problem of counting the number of perfect matchings in a bipartite graph. We focus on the problem of counting the common bases in pairs of regular matroids, and describe a class called *Pfaffian matroid pairs* for which this enumeration problem can be solved. We prove that when a pair of regular matroids is non-Pfaffian, there is a set of common bases which certifies this, and that the number of bases in the certificate is linear in the size of the ground set of the matroids. When both matroids in a pair are series-parallel, we prove that determining if the pair is Pfaffian is equivalent to finding an edge signing in an associated graph, and in the case that the pair is non-Pfaffian, we obtain a characterization of this associated graph. Pfaffian bipartite graphs are a class of graphs for which the number of perfect matchings can be determined; we show that the class of series-parallel Pfaffian matroid pairs is an extension of the class of Pfaffian bipartite graphs.

Edmonds proved that the polytope generated by the common bases of a pair of matroids is equal to the intersection of the polytopes generated by the bases for each matroid in the pair. We consider when a similar property holds for the binary space, and give an excluded minor characterization of when the binary space generated by the common bases of two matroids can not be determined from the binary spaces for the individual matroids. As a result towards a description of the lattice of common bases for a pair of matroids, we show that the lattices for the individual matroids determine when all common bases of a pair of matroids intersect a subset of the ground set with fixed cardinality.

#### Acknowledgements

I thank the faculty, staff and graduate students in the Department of Combinatorics and Optimization for all their support and encouragement during my time in Waterloo. Discussions with John Irving towards the end of my Masters convinced me to begin this program and he provided a reliable source of empathy and humor throughout. James Muir was always available for technical support, lunch, and friendship. Aaron Williams wrote several computer programs which were invaluable to me, as was his friendship. I would also like to thank my committee members for their many helpful suggestions.

Above all I sincerely thank Jim Geelen for all his time, patience, and support. It was extremely rewarding to work with him, and I am honored to have done so.

### Contents

1	Introduction				
<b>2</b>	Preliminaries				
	2.1	Matroids	7		
	2.2	Matroid intersection	9		
	2.3	Fundamental graphs	0		
	2.4	Pivoting	2		
	2.5	Totally unimodular graphs	3		
	2.6	Connectivity	6		
3	Pfaf	fian Matroid Pairs 2	0		
	3.1	Counting common bases	0		
	3.2	Properties of Pfaffian matroid pairs	3		
	3.3	Pfaffian bipartite graphs	4		
	3.4	Certifying non-Pfaffian pairs	6		
	3.5	Examples	0		
	3.6	Unique signing	4		
	3.7	Exact basis problem 3	5		
<b>4</b>	Binary Spaces 38				
	4.1	The binary space of a matroid	9		
	4.2	Cycle spaces in directed graphs	0		
	4.3	The binary space for a pair of matroids	5		
	4.4	An excluded minor characterization	0		
<b>5</b>	Lat	tices 5	8		
	5.1	Matroid polytopes	9		
	5.2	Separators and constant-size intersecting sets	1		
6	Series-Parallel Matroids 6				
	6.1	Fundamental graphs of series-parallel matroids	0		
	6.2	Contributing circuits and decomposing circuits	4		

6.3	Vertex signings of the fundamental graph	76
6.4	Proper edge signings of the fundamental graph	78
6.5	Accordions	83
6.6	Neutral circuits and a certificate of constant size	87
6.7	Odd double cycles	100

### Bibliography

## Chapter 1 Introduction

Given two matrices  $A_1$  and  $A_2$  whose columns are indexed by a common set S, we are interested in counting the number of common column bases of  $A_1$  and  $A_2$ . We may therefore assume that both matrices have rank r and, by possibly removing redundant rows, we may assume that  $A_1$  and  $A_2$  each have r rows. In this thesis we explore the Cauchy-Binet Formula as a tool for counting the common bases of  $A_1$  and  $A_2$ . The formula states that:

$$\det A_1 A_2^{\top} = \sum_{X \subseteq S: \, |X| = r} \det A_1[X] A_2[X].$$

Here  $A_i[X]$  denotes the submatrix of  $A_i$  indexed by the columns X. Observe that if det  $A_1[X]A_2[X] = 1$  for each common basis X of  $A_1$  and  $A_2$ , then det  $A_1A_2^{\top}$  counts the number of common bases of  $A_1$  and  $A_2$ . This condition is highly restrictive, and we will place the additional restriction that  $A_1$  and  $A_2$  are totally unimodular. That is, we will restrict the counting bases problem to the case that each square nonsingular submatrix of  $A_1$  and  $A_2$  has determinant  $\pm 1$ . We call the pair  $(A_1, A_2)$  of matrices *Pfaffian* if  $A_1$  and  $A_2$  are totally unimodular and either

- (i) det  $A_1[X]A_2[X] = 1$  for each common basis X of  $A_1$  and  $A_2$ , or
- (ii) det  $A_1[X]A_2[X] = -1$  for each common basis X of  $A_1$  and  $A_2$ .

We say that the pair  $(A_1, A_2)$  of totally unimodular matrices has a *Pfaffian signing* if we can scale some of the columns of  $A_1$  and  $A_2$  by -1 so that  $(A_1, A_2)$  becomes Pfaffian. Note that in case (ii), scaling a row of  $A_2$  by -1 results in a pair of matrices satisfying (i), and note also that if  $(A_1, A_2)$  has a Pfaffian signing, then it suffices to re-sign the columns of  $A_2$ , since re-signing a column of  $A_1$  and the corresponding column of  $A_2$  does not affect whether or not the pair  $(A_1, A_2)$  is Pfaffian.

The use of the term "Pfaffian" highlights the connection between Pfaffian matrix pairs and *Pfaffian bipartite graphs*, discussed in Section 3.3. Let G = (V, E) be a bipartite graph with bipartition (U, W) and assume that |U| = |W| and that G has no isolated vertices. Let  $A_1$  and  $A_2$  be the  $U \times E$  and  $W \times E$  incidence matrices of G. If  $X \subseteq E$ , then X is a basis of  $A_i$  if each vertex in  $A_i$  is incident with exactly one edge in X, and thus the common bases of  $A_1$  and  $A_2$  are the perfect matchings of G. Since each column of  $A_1$  and  $A_2$  has exactly one nonzero entry,  $A_1$  and  $A_2$  are totally unimodular and moreover,  $A_1$  and  $A_2$  remain totally unimodular if we replace some of their +1 entries with -1. In Section 3.3 we show that G admits a Pfaffian orientation if and only if  $(A_1, A_2)$  has a Pfaffian signing. The notion of Pfaffian matroid pairs and their relationship with Pfaffian bipartite graphs is due to Dirk Vertigan, personal communication.

While we are explicitly interested in matrices, many of the concepts are matroidal and matroid theory provides a convenient forum for the discussion. We provide the necessary background in Chapter 2, but will assume that the reader is somewhat familiar with matroids during this introduction. A matroid pair is a pair  $P = (M_1, M_2)$  of matroids  $M_1$  and  $M_2$ , where  $M_1$  and  $M_2$  have equal rank and are defined on the same ground set S. We are interested in the common bases of  $M_1$  and  $M_2$ , which we call bases of P. Edmonds' Matroid Intersection Theorem [10] states that P has a basis if and only if

$$\operatorname{rank}_{M_1}(S_1) + \operatorname{rank}_{M_2}(S_2) \ge \operatorname{rank}_{M_1}(S)$$

for all partitions  $\{S_1, S_2\}$  of S.

A matroid pair  $P = (M_1, M_2)$  is called *Pfaffian* if there exists a Pfaffian pair  $(A_1, A_2)$  of matrices such that  $A_1$  and  $A_2$  are representations over the reals of  $M_1$  and  $M_2$  respectively; we call such a pair  $(A_1, A_2)$  a *Pfaffian representation* of *P*. Since  $A_1$  and  $A_2$  are totally unimodular,  $M_1$  and  $M_2$  are necessarily regular matroids. A theorem of Camion [3] shows that a regular matroid has an essentially unique totally unimodular representation, which implies that if  $A_1$  and  $A_2$  are totally unimodular representations of regular matroids  $M_1$ and  $M_2$ , then  $(M_1, M_2)$  is Pfaffian if and only if  $(A_1, A_2)$  has a Pfaffian signing.

If G is a bipartite graph, then we will denote the associated matroid pair by P(G); such pairs are called graphic matroid pairs, and thus a graphic matroid pair is simply a pair of partition matroids on a common ground set. A matching covered graph is a connected graph with the property that each edge of the graph is in a perfect matching and it is natural to restrict the study of Pfaffian bipartite graphs to matching covered graphs. We introduce an analogous notion for matroid pairs: a matroid pair  $P = (M_1, M_2)$  is connected if  $\operatorname{rank}_{M_1}(S_1) + \operatorname{rank}_{M_2}(S_2) > \operatorname{rank}_{M_1}(S)$  for each proper partition  $\{S_1, S_2\}$  of S. In Theorem 3.8 we show that if G is a bipartite graph with a perfect matching, then G is matching covered if and only if the graphic matroid pair P(G) is connected.

The study of Pfaffian graphs began in 1963 when, in the context of crystal physics, Kastelyn [20] proved that every planar graph is Pfaffian. More recently, Robertson, Seymour, and Thomas [35] and, independently, McCuaig [28] proved that all Pfaffian bipartite graphs can be constructed from planar graphs and copies of a particular nonplanar graph called the Heawood graph, by certain compositions across 4-element sets. In addition to providing a good characterization for Pfaffian bipartite graphs, these results also provide an efficient algorithm for recognizing Pfaffian bipartite graphs. The study of Pfaffian pairs is still in its infancy, and we do not yet know if these remarkable results for bipartite graphs extend to matroids.

Preceding the decomposition results mentioned above, Little [22] proved a beautiful partial characterization of Pfaffian bipartite graphs as those bipartite graphs which do not "contain" an odd subdivision of  $K_{3,3}$  (see Section 3.3). In the context of matroid pairs, this characterization of Pfaffian bipartite graphs can be viewed as an excluded minor characterization. Let  $P = (M_1, M_2)$  be a matroid pair and let X and Y be disjoint subsets of S. If there exists a basis B of P such that  $Y \subseteq B$  and  $B \cap X = \emptyset$ , then we call the pair  $P \setminus X/Y = (M_1 \setminus X/Y, M_2 \setminus X/Y)$  a minor of P. The dual of a matroid pair  $P = (M_1, M_2)$  is the matroid pair  $P^* = (M_1^*, M_2^*)$  and Theorems 3.4 and 3.5 show that the class of Pfaffian matroid pairs is closed under duality and minors. With this description of a minor a matroid pair, Little's Theorem then says that a bipartite graph G is non-Pfaffian if and only if the graphic matroid pair P(G) contains a minor P(H)where H is an odd subdivision of  $K_{3,3}$ . Thus Little's Theorem provides a succinct method for proving that a matroid pair is non-Pfaffian, but it does not help to prove that a pair is Pfaffian.

The problem of finding an excluded minor characterization for Pfaffian matroid pairs remains open. There are already infinitely many excluded minors coming from the class of graphic matroid pairs, and the existence of many other excluded minors for the class of Pfaffian matroid pairs, such as those discussed in Sections 6.6 and 6.7, suggests that characterizing the excluded minors for Pfaffian matroid pairs may be complicated.

The goals of this thesis are two-fold:

- (1) Understand how the different Pfaffian signings of a given Pfaffian matroid pair are related.
- (2) Find succinct certificates for proving that a given matroid pair is non-Pfaffian.

#### Equivalent Pfaffian signings

It is well known that any one Pfaffian orientation of a matching covered bipartite graph G can be obtained from any other Pfaffian orientation by reversing the orientation across some cut in G. This property does not extend to general (non-bipartite) Pfaffian graphs, and despite considerable effort, Little's Theorem and the decomposition of Robertson, Seymour, Thomas, and McCuaig have resisted generalization to general graphs. The existence of "inequivalent" Pfaffian orientations is often cited as the root of the difficulty; similar difficulties arise for Pfaffian matroid pairs.

Let  $P = (M_1, M_2)$  be a Pfaffian matroid pair on the ground set S and let  $(A_1, A_2)$  be a Pfaffian representation of P. By resigning  $(A_1, A_2)$  on X we mean scaling by a factor of -1 the columns of  $A_2$  indexed by X. A set  $X \subseteq S$  is called a *constant-parity intersecting*  set of P if either  $|X \cap B|$  is even for all bases B of P, or  $|X \cap B|$  is odd for all bases B of P. Note that resigning a Pfaffian representation on  $X \subseteq S$  gives another Pfaffian representation if and only if X is a constant-parity intersecting set of P.

Suppose that  $X_1 \subseteq S$  is a separator of  $M_1$ ; that is,  $\operatorname{rank}_{M_1}(X_1) + \operatorname{rank}_{M_1}(S \setminus X_1) = \operatorname{rank}(M_1)$ . Then for each basis B of  $M_1$ , we have  $|B \cap X_1| = \operatorname{rank}_{M_1}(X_1)$ , and therefore each separator of  $M_1$  or of  $M_2$  is a constant-parity intersecting set of P. Moreover, the symmetric difference of two constant-parity intersecting sets gives another constant-parity intersecting set for P can be expressed as the symmetric difference of separators of  $M_1$  and  $M_2$ , as the following example shows: let  $A_1$  and  $A_2$  be the rank 3 totally unimodular matrices given by

$$A_{1} = \begin{pmatrix} 1 & 2 & 3 & 4 & 5 & 6 \\ 1 & 0 & 0 & 1 & 0 & 0 \\ 0 & 1 & 0 & 0 & 1 & 0 \\ 0 & 0 & 1 & 0 & 0 & 1 \end{pmatrix} \text{ and } A_{2} = \begin{pmatrix} 1 & 2 & 3 & 4 & 5 & 6 \\ 1 & 0 & 0 & 0 & 1 & 1 \\ 0 & 1 & 0 & -1 & 0 & 1 \\ 0 & 0 & 1 & -1 & -1 & 0 \end{pmatrix}$$

If  $M_1$  and  $M_2$  are the matroids represented by  $A_1$  and  $A_2$  respectively, then it is easy to check that  $X = \{4, 5, 6\}$  is a constant-parity intersecting set of the matroid pair  $P = (M_1, M_2)$ . However, each separator of  $M_1$  and  $M_2$  has even size, and therefore Xcannot be written as the symmetric difference of separators of  $M_1$  and  $M_2$ . This matroid pair is an example of a "twined  $K_4$ " pair given in Example 2 of Section 3.5 and discussed further in Section 4.4. The following theorem is one of the main contributions of the thesis.

**Theorem 1.1.** Let  $P = (M_1, M_2)$  be a connected matroid pair on the ground set S where  $M_1$  and  $M_2$  are binary. If P has no twined  $K_4$  minor, then  $X \subseteq S$  is a constant-parity intersecting set of P if and only if there exist separators  $X_1$  of  $M_1$  and  $X_2$  of  $M_2$  such that  $X = X_1 \Delta X_2$ .

Theorem 1.1 is proved in Section 4.4 as a consequence of a characterization of the binary space generated by the bases of P.

In Chapter 5 we discuss a closely related property to constant-parity intersecting sets called *constant-size intersecting sets*: a subset X of the ground set for the matroid pair P is a constant-size intersecting set of P if there exists an integer k such that  $|X \cap B| = k$ for each basis B of P. In Theorem 5.4 we characterize the constant-size intersecting sets of any connected matroid pair  $P = (M_1, M_2)$  in terms of the separators of  $M_1$  and  $M_2$ . Theorem 5.4, unlike Theorem 1.1, does not require that  $M_1$  and  $M_2$  are binary and does not require the exclusion of twined  $K_4$ 's.

#### Certifying non-Pfaffian pairs

Let  $P = (M_1, M_2)$  be a matroid pair on the ground set S. We would like to determine whether or not P is a Pfaffian matroid pair. By Seymour's decomposition of regular matroids [39], we can efficiently check that  $M_1$  and  $M_2$  are regular and can then efficiently find totally unimodular representations  $A_1$  and  $A_2$  of  $M_1$  and  $M_2$  respectively. It remains to determine whether  $(A_1, A_2)$  has a Pfaffian signing. This problem can, abstractly, be reformulated using linear algebra.

Let  $\mathcal{B}(P)$  denote the set of all bases of P and let K be the  $\mathcal{B}(P) \times S$  matrix over GF(2) whose rows are the characteristic vectors for the bases of P. For each  $B \in \mathcal{B}(P)$  we define  $\alpha_B$  and  $\alpha'_B$  in GF(2) such that det  $A_1[B] \det A_2[B] = (-1)^{\alpha_B}$  and  $\alpha'_B = 1 + \alpha_B$ . The matrix pair  $(A_1, A_2)$  then has a Pfaffian signing if an only if there exists a vector x over GF(2) such that either  $Kx = \alpha$  or  $Kx = \alpha'$ . Unfortunately this is not a practical reformulation since K may have exponentially many rows and we do not explicitly know the set of bases of P. However, since K has linear rank, this formulation can be used to succinctly certify that a matroid pair is non-Pfaffian.

A set  $C \subseteq \mathcal{B}(P)$  is called *inconsistent* with respect to  $\alpha$  for P if there is no set  $X \subseteq S$  such that  $|X \cap B| \equiv \alpha_B \mod 2$  for all  $B \in C$ . It follows that if P has an inconsistent set with respect to  $\alpha$  and P has an inconsistent set with respect to  $\alpha'$ , then P is non-Pfaffian. The union of two such inconsistent sets is called a *non-Pfaffian bases certificate*. The following theorem is proved in Section 3.4:

**Theorem 1.2.** If  $P = (M_1, M_2)$  is a matroid pair on the ground set S and  $M_1$  and  $M_2$  are both regular matroids, then either P is Pfaffian or P has a non-Pfaffian bases certificate containing at most 2|S| + 2 bases.

For graphic matroid pairs, Little's Theorem [22] (Section 3.3) implies the much stronger result that the number of bases required in a non-Pfaffian bases certificate is six. We conjecture that this general linear bound for certifying that a matroid pair is non-Pfaffian can be improved to a constant bound. The constant in our conjecture must be at least six, and as of yet we do not have an example of a non-Pfaffian matroid pair which shows that the constant must be greater than six.

**Conjecture 1.3.** There exists a constant c such that if P is the matroid pair  $(M_1, M_2)$  and  $M_1$  and  $M_2$  are regular, then either P is Pfaffian or there is a non-Pfaffian bases certificate for P with at most c bases.

Aesthetically, Little's Theorem is preferable to Theorem 1.2, and our progress toward a better characterization for non-Pfaffian matroid pairs has been hampered by inequivalent signings. In light of Theorem 1.1, it is natural to restrict our focus to the class of matroid pairs with no twined  $K_4$  minor. We impose the stronger condition that neither  $M_1$  nor  $M_2$  contains an  $M(K_4)$  minor. That is, we consider the special case where  $M_1$  and  $M_2$  are series-parallel matroids. This class still contains the graphic matroid pairs and is closed with respect to duality and minors, and for this class we reformulate the problem of determining whether or not P is Pfaffian to questions concerning the cycle space of a certain directed graph. We then apply an unpublished theorem of Gerards to obtain a nicer characterization involving "odd double-cycles"; see Theorem 6.22. (Gerards theorem is a generalization of a well-known theorem of Seymour and Thomasson [38] on even cycles in directed graphs.) While we are unfortunately unable to prove Conjecture 3.12 for the series-parallel class, we make significant progress; see Chapter 6.

# Chapter 2 Preliminaries

Most of the matroids considered in this thesis are *representable* matroids, particularly the matroid pairs in the class we call Pfaffian matroids, and thus most of our results are results about matrices. The language of matroids is therefore not required, however the use of matroid terminology allows the application of matroid defined operations like duality, minors, and pivoting. Here we formally define matroids and the matroid intersection problem. Using the *fundamental graph*, a well known graphical representation for a single matroid, we describe a graphical representation of a pair of matroids. This graphical representation of a matroid pair will be used extensively in Chapters 3, 4, and 6.

We explain the connection between totally unimodular matrices and the matroid class we are restricting the counting bases problem to, namely *regular* matroids, and we state some fundamental theorems of Tutte which highlight the essential role of regular matroids within matroid theory. The final section of this chapter is devoted to the notion of connectivity in matroids and the relationship between matroid connectivity and connectivity in the directed graph we use to represent pairs of matroids; in Theorem 2.8 we show that this directed graph is strongly connected if and only if the matroid pair is connected.

#### 2.1 Matroids

A matroid M consists of a ground set S and a nonempty collection  $\mathcal{I}$  of subsets of S satisfying the following two axioms:

- (i) If  $I_1 \subseteq I_2$  and  $I_2 \in \mathcal{I}$  then  $I_1 \in \mathcal{I}$ .
- (ii) If  $I_1, I_2 \in \mathcal{I}$  and  $|I_1| < |I_2|$  then there exists  $e \in I_2 \setminus I_1$  such that  $I_1 \cup \{e\} \in \mathcal{I}$ .

The elements of  $\mathcal{I}$  are called the *independent* sets of M and a subset of S which is not independent is *dependent*.

A common example of a matroid comes from linear algebra. Let A be a matrix over a fixed field and let S be the set of column indices of A. If  $\mathcal{I}$  is defined to be all subsets of columns of A that form an independent set, then  $(S, \mathcal{I})$  is a matroid. A matroid that can be represented by a matrix in this way is called *representable* and the representable matroids are a fundamental class in matroid theory. We will be working mainly with *binary* matroids (matroids representable over GF(2)) and with *regular* matroids, which are matroids representable over the reals by a *totally unimodular* matrix. A matrix is totally unimodular if and only if each square nonsingular submatrix has determinant +1 or -1; totally unimodular matrices are discussed in Section 2.5.

Another fundamental class of matroids comes from graph theory: if G is a graph and a subset of the edges of G is called independent if and only if it induces an acyclic subgraph in G, then the independent sets of edges satisfy axioms (i) and (ii). A matroid defined on the edges of a graph G in this way is called the *cycle matroid* for G, and is denoted M(G). The graphic matroids are those matroids which can be represented by a cycle matroid for a graph, and the graphic matroids are a subclass of the regular matroids (see Oxley [31]). The maximal independent sets of a graphic matroid correspond to spanning forests of the graph, and an inclusion minimal dependent set in a graphic matroid corresponds to a circuit in the graph.

Matroids were first considered by Whitney [46] in 1935 as a generalization of this notion of independence in graphs and linear independence. Terms from linear algebra and graph theory transfer to matroid theory: a *basis* of a matroid is an independent set of maximum size, the *rank* of a subset S' of the ground set of a matroid is the maximum size of an independent set in S', and a *circuit* is an inclusion minimal dependent set. If M is a matroid on the ground set S and  $S' \subseteq S$ , then we will denote the rank of S' by rank<sub>M</sub>(S'). The rank of S is called the rank of the matroid and is denoted rank(M). An element with rank 0 is a *loop*. Axiom (ii) implies that all bases of a matroid have the same number of elements, and since the independent sets in a matroid are all subsets of a basis of the matroid, a matroid is completely determined by its bases.

The operations of duality and minors from graph theory also apply to matroid theory.

**Duality:** If G is a planar graph and  $G^*$  is a planar dual of G, then there is a natural bijection between the edges of G and the edges of  $G^*$ . A set F of edges in G is a spanning forest in G if and only if the complement of the corresponding edges in  $G^*$  is a spanning forest in  $G^*$ . Similarly, if M is a matroid on the ground set S and  $B \subseteq S$ , then  $S \setminus B$  is a basis of the matroid dual  $M^*$  for M if and only if B is a basis for M.

**Deletion:** If e is an element in the ground set for the matroid M, then the bases for the matroid formed by deleting e are exactly those bases of M that do not contain the element e. This matroid is denoted  $M \setminus \{e\}$ . If A is a matrix and M is the matroid represented by A, then deleting an element of M corresponds to deleting the corresponding column

of A. If M = M(G) then  $M \setminus \{e\}$  is equivalent to the cycle matroid for  $G \setminus \{e\}$ .

**Contraction:** If e is a loop of the matroid M then  $M/\{e\} = M \setminus \{e\}$ . Otherwise, the notion of contraction in a matroid is defined via duality and deletion: if e is not a loop of M then  $M/\{e\} = (M^* \setminus \{e\})^*$ . It follows that B is a basis of  $M/\{e\}$  if and only if  $B \cup \{e\}$  is a basis of M. When M = M(G) then M/e = M(G/e). Let M be the matroid represented by the matrix A and let e index a column of A. If e has one nonzero entry, then contracting e from M corresponds to deleting the row of A containing the nonzero entry for column e and deleting column e. Row operations on A do not affect column dependencies, and thus if A' is obtained from A by a series of row operations then A and A' represent the same matroid. It follows that if e is not a loop of M then to obtain a representation of  $M/\{e\}$  we may first perform row operations such that column e in A has exactly one non-zero entry. These row operations are formalized by the matroid operation called *pivoting* which is discussed in Section 2.4.

#### 2.2 Matroid intersection

Given two matroids M and N on the same ground set S, the matroid intersection problem is to find a common independent set of maximum size. If  $\{S_1, S_2\}$  is a partition of S and  $X \subseteq S$  is independent in both M and N, then

$$|X| = |X \cap S_1| + |X \cap S_2| \le \operatorname{rank}_M(S_1) + \operatorname{rank}_N(S_1).$$
(2.1)

Thus the maximum size of a common independent set is at most the minimum of  $\operatorname{rank}_M(S_1) + \operatorname{rank}_N(S_1)$  over all partitions  $\{S_1, S_2\}$  of S. A beautiful theorem of Edmonds [10] proves that there is an independent set  $X \subseteq S$  and a partition  $\{S_1, S_2\}$  of S for which (2.1) is met with equality.

**Theorem 2.1 (Edmonds' Matroid Intersection Theorem).** If M and N are matroids on the ground set S then the maximum size of a common independent set in M and N is equal to the minimum of  $\operatorname{rank}_M(S_1) + \operatorname{rank}_N(S_2)$  over all partitions  $\{S_1, S_2\}$  of S.

Corollaries of this theorem include König's theorem for the maximum size of a matching in a bipartite graph and Menger's theorem for the maximum number of vertex disjoint directed paths between two vertex subsets of a directed graph. Assuming independence can be checked in polynomial time with respect to the size of the ground set, results of Edmonds [9], [10] prove that there is a polynomial algorithm for finding a common independent set of maximum size in two matroids. If a weight is assigned to each element in the ground set, then the weighted version of the matroid intersection problem is to find a common independent set of maximum weight. This has a similarly elegant solution and will be discussed in Section 3.7. If M and N are matroids with the same rank on the same ground set S then we call P = (M, N) a matroid pair. The rank of P is the common rank of M and N and the bases of P are the common bases of M and N. Edmonds' Matroid Intersection Theorem clearly determines if two matroids have a common basis and the problem of deciding if P has a basis can be efficiently solved.

#### 2.3 Fundamental graphs

We present a well known graphical representation of a matroid, called the fundamental graph, and give a method for combining the fundamental graphs for two matroids on the same ground set into one graphical representation of a matroid pair. The transfer of properties of a matroid pair to properties of the graphical representation for the pair are discussed, and fundamental results required for later chapters are given.

Let M be a matroid on the ground set S and let B be a basis of M. The partial representation of M with respect to B is the  $\{0,1\}$  matrix R with rows indexed by the elements of B, columns indexed by the elements of  $S \setminus B$ , and  $[R]_{i,j} = 1$  if and only if i belongs to the unique circuit in  $B \cup \{j\}$ . The bipartite graph with adjacency matrix R is called the *fundamental graph* for M, and we denote this graph by G(M, B). If  $v \in B$  then deleting row v from R leaves the partial representation of M/v with respect to  $B \setminus v$  and if  $v \in S \setminus B$  then deleting column v from R leaves the partial representation for  $M \setminus v$  with respect to B. Thus

$$G(M,B) \setminus \{v\} = \begin{cases} G(M/v, B \setminus \{v\}), & \text{if } v \in B; \\ G(M \setminus v, B), & \text{if } v \in S \setminus B. \end{cases}$$
(2.2)

If H is an induced subgraph of G(M, B) then it follows from (2.2) that there exists a minor M' of M and a basis B' of M' such that H = G(M', B'). If  $M^*$  is the dual of M and R is the partial representation of M with respect to the basis B, then  $R^T$  is the partial representation of  $M^*$  with respect to  $S \setminus B$ . Therefore the fundamental graph G(M, B) is isomorphic to  $G(M^*, S \setminus B)$ .

When M is a binary matroid, the matrix [I|R] is a representation of M over GF(2). If S is the ground set for the binary matroid M and  $F \subseteq S$ , it follows that F is a basis of M if and only if |F| = |B| and  $R[B \setminus F; F \setminus B]$  is nonsingular. Here  $R[B \setminus F; F \setminus B]$  denotes the submatrix of R indexed by rows  $B \setminus F$  and columns  $F \setminus B$ ; the union of  $B \setminus F$  and  $F \setminus B$  is denoted  $B \Delta F$ . If A is the adjacency matrix for the bipartite graph G then there is a one-to-one correspondence between perfect matchings in G and non-zero terms in the permutation expansion of the determinant of A and thus the following observation is immediate: **Observation 2.2.** If M is a binary matroid on the ground set S and B is a basis of M, then  $F \subseteq S$  is a basis of M if and only if the subgraph of G(M, B) induced by  $B\Delta F$  has an odd number of perfect matchings.

Given a matroid pair P = (M, N), and a basis B of P, the directed bipartite graph G(P, B) is constructed as follows: direct the edges of the fundamental graph G(M, B) from B to  $S \setminus B$ , direct the edges of the fundamental graph G(N, B) from  $S \setminus B$  to B, and identify the vertices of G(M, B) and G(N, B). The resulting graph is called the fundamental graph for the matroid pair P. For example, let P = (M, N) be the matroid pair with basis B and let M and N have partial representations  $R_M$  and  $R_N$  with respect to B, as shown below:

$$R_M = \begin{bmatrix} 1 & 0 & 0 & 0 & 1 & 0 \\ 0 & 1 & 1 & 0 & 0 & 0 \\ 0 & 0 & 0 & 1 & 0 & 1 \end{bmatrix}, \qquad R_N = \begin{bmatrix} 0 & 1 & 0 & 0 & 0 & 1 \\ 1 & 0 & 0 & 1 & 0 & 0 \\ 0 & 0 & 1 & 0 & 1 & 0 \end{bmatrix}.$$

The fundamental graph G(P, B) is isomorphic to the graph in Figure 2.1, where the black vertices correspond to the elements of B.



Figure 2.1: G(P, B)

Given a matroid pair P on the ground set S, a subset F of S is called *contributing* with respect to the basis B if  $B\Delta F$  is a basis of P. A subgraph H of the fundamental graph G(P, B) for P is contributing if V(FH) is contributing with respect to B. Thus there is a bijection between induced contributing subgraphs of G(P, B) and bases of P. When the basis B of P is arbitrary we will denote G(P, B), G(M, B), and G(N, B) by G(P), G(M), and G(N) respectively.

Let P be a matroid pair and let D be a set of vertex disjoint directed circuits in G(P). Then E(D) induces a perfect matching in G(M)[V(D)] and E(D) induces a perfect matching in G(N)[V(D)]. Denote these matchings by  $T_M(D)$  and  $T_N(D)$  respectively. If D' is a set of vertex disjoint directed circuits in G(P) with V(D) = V(D'), then  $D \neq D'$ if and only if either  $T_M(D) \neq T_M(D')$  or  $T_N(D) \neq T_N(D')$ . Thus D is the unique set of vertex disjoint directed circuits in G(P)[V(D)] if and only if G(M)[V(D)] and G(N)[V(D)] have unique perfect matchings. It follows from Observation 2.2that if D is the unique set of vertex disjoint directed circuits in G(P)[V(D)], then D is contributing. When D is the unique set of vertex disjoint directed circuits in G(P)[V(D)], then D is contributing. call D a simple contributing subgraph of G(P). In Chapter 6 we will consider a class of matroid pairs P for which every contributing circuit in G(P) is simple.

If  $M^*$  and  $N^*$  are the dual matroids of M and N respectively then the matroid pair  $(M^*, N^*)$  is called the *dual* of P and is denoted  $P^*$ . Since  $G(M^*, S \setminus B)$  is isomorphic to G(M, B) and  $G(N^*, S \setminus B)$  is isomorphic to G(N, B), the graph G(P, B) is isomorphic to the graph obtained by reversing the orientation of the edges in  $G(P^*, S \setminus B)$ .

If P is the matroid pair (M, N) and  $A_M$  and  $A_N$  are representations of M and N respectively, then  $(A_M, A_N)$  is called a representation of P. If  $A_M$  does not have full row rank then a row of  $A_M$  can be deleted without affecting the column dependencies in M; thus if  $A_M$  is a representation of M then we will assume that  $A_M$  has full row rank. Similarly, if  $(A_M, A_N)$  is a representation of a matroid pair then we assume that both  $A_M$  and  $A_N$  have full row rank. If B is a basis of P and the matrices  $R_M$  and  $R_N$  are the partial representations of M and N with respect to B, then  $(R_M, R_N)$  is referred to as the partial representation of P with respect to B.

#### 2.4 Pivoting

Given bases B and B' of a matroid M, we would like to know the relationship between G(M, B) and G(M, B'). If R is the partial representation of M with respect to B and R' is the partial representations of M with respect to B', then determining the relationship between G(M, B) and G(M, B') is equivalent to determining the relationship between R and R'. The operation that moves from one partial representation of a matroid to another is called *pivoting* and is discussed in this section.

Suppose that M is a binary matroid on the ground set S with basis B. Let  $X \subseteq B$  and  $Y \subseteq S \setminus B$  be such that  $B' = B\Delta(X \cup Y)$  is a basis of M. If R is the partial representation of M with respect to B and W = R[X; Y], then up to permuting rows and columns, we may assume that R has the block decomposition

$$R = \begin{bmatrix} W & C \\ E & D \end{bmatrix}.$$

Pivoting R on  $X \cup Y$  is the matrix operation which results in the matrix R', where

$$R' = \begin{bmatrix} W^{-1} & W^{-1}C \\ -EW^{-1} & D - EW^{-1}C \end{bmatrix}.$$

If A = [I|R] then A is a representation of M over GF(2) and pivoting A on  $X \cup Y$  results

in the matrix A', where

$$A' = \begin{bmatrix} W^{-1} & 0 \\ -EW^{-1} & I \end{bmatrix} A$$
$$= \begin{bmatrix} W^{-1} & 0 \\ -EW^{-1} & I \end{bmatrix} \begin{bmatrix} I & 0 & W & C \\ 0 & I & E & D \end{bmatrix}$$
$$= \begin{bmatrix} W^{-1} & 0 & I & W^{-1}C \\ -EW^{-1} & I & 0 & D - EW^{-1}C \end{bmatrix}$$

Note that the columns of A' indexed by B' are a permutation of the identity matrix, and that a permutation of the columns of A' results in the matrix [I|R']. Since A' was obtained by multiplying A by a nonsingular matrix, A' is a representation of M over GF(2) and thus R' is the partial representation of M with respect to B'. We now consider how this pivoting operation affects G(P, B). It suffices to consider the case of pivoting R on  $\{u, v\}$ for some  $u \in B, v \in S \setminus B$  with  $R[u; v] \neq 0$ .

Let  $u \in B$  index the first row of R and let  $v \in S \setminus B$  index the first column of R and assume that

$$R = \begin{bmatrix} 1 & c^T \\ e & D \end{bmatrix}.$$
 (2.3)

If R' is the matrix obtained by pivoting R on  $\{u, v\}$ , then over GF(2),

$$R' = \begin{bmatrix} 1 & c^T \\ e & D + ec^T \end{bmatrix}.$$
 (2.4)

Since  $R[u; v] \neq 0$ , the set  $B' = B\Delta\{u, v\}$  is a basis of M and R' is the partial representation of M with respect to B'.

If x is a vertex in the graph G then let N(x) denote the vertices adjacent to x in G. By considering Equations (2.3) and (2.4) we determine that G(M, B') is obtained from G(M, B) by complementing the edges between  $N(u) \setminus \{v\}$  and  $N(v) \setminus \{u\}$  and swapping the labels on u and v. Figure 2.2 shows the graphs G and G' where G' is obtained by pivoting on the edge uv in G. Note that the graph obtained by pivoting twice on the edge uv in G.

#### 2.5 Totally unimodular graphs

An edge signing of a graph is a  $\{0, 1\}$  function on the edges of the graph. In this section we consider the relationship between edge signings of the graph G(P) for a matroid pair P and totally unimodular representations of P. These edge signings play a major role in Chapters 3 and 6. We also give some well known and fundamental theorems for regular matroids as a motivation for restricting the counting bases problem to a subclass of regular matroids.



Figure 2.2: Pivoting G on the edge uv

A matrix A over the reals is *totally unimodular* if every square submatrix of A has determinant 0, +1, or -1. A signing of a  $\{0, 1\}$  matrix A is a matrix A' obtained from A by changing some of the nonzero entries to -1; a  $\{0, 1\}$  matrix is called *regular* if it has a totally unimodular signing. A scaling of a is obtained by scaling some of its rows and columns by -1; Camion [3] observed that a totally unimodular signing of a regular matrix is unique up to scaling.

A matroid is regular if it can be represented over GF(2) by a regular matrix. Equivalently, a matroid is regular if it can be represented over the reals by a totally unimodular matrix. The property of being totally unimodular is closed under pivoting (see Cornuéjouls [5]) and since the matrix R is regular if and only if [I|R] is regular, determining if a binary matroid M is regular is equivalent to determining if a partial representation of M is regular. If P is the matroid pair (M, N) then we say that P is regular if M and N are regular. If  $(A_M, A_N)$  is a representation of P then  $(A_M, A_N)$  is called a totally unimodular representation of P if both  $A_M$  and  $A_N$  are totally unimodular matrices. Regular matroids are a fundamental class in matroid theory, as the next theorem indicates.

**Theorem 2.3 (Tutte** [43]). Let M be a matroid. The following are equivalent:

- 1. M is regular.
- 2. M is representable over every field.
- 3. M is representable over GF(2) and GF(3).

An edge signing s of the graph G is even if  $s_e = 0$  for all edges e in G. There is a natural bijection between edge signings s of G and signings A' of the adjacency matrix for G given by  $s_e = 1$  if and only if the entry for e in A' is -1. If A' is a totally unimodular signing of the adjacency matrix for G then the associated edge signing of G is called a totally unimodular signing of G. If P = (M, N) is a matroid pair with basis B, then an edge signing of G(P, B) is totally unimodular if the edge signing restricted to G(M, B) is totally unimodular and the edge signing restricted to G(N, B) is totally unimodular. By Camion's result on the uniqueness of totally unimodular signings, a totally unimodular edge signing of a graph is unique up to resigning across cuts. Thus if the graph G has a totally unimodular edge signing and F is a forest of G, then the totally unimodular edge signing can be fixed on E(F).

Let F be a maximal forest of the graph G and let  $\{e_1, e_2, \ldots, e_k\}$  be an ordering of  $E(G) \setminus E(F)$  such that for  $i = 1, \ldots, k$ , there is a circuit  $C_i$  through  $e_i$  satisfying both of the following:

- $E(C_i) \subseteq E(F) \cup \{e_1, e_2, \dots, e_i\}$ , and
- $C_i$  is an induced circuit in G.

This can be done, for example, by selecting  $e_1$  such that the path in F connecting the end vertices of  $e_1$  is minimal, and for  $j \ge 2$  selecting  $e_j$  such that the path in  $E(F) \cup \{e_1, e_2, \ldots, e_{i-1}\}$  is minimal. If all the edges of  $C_i \setminus \{e_i\}$  are signed, then there is a unique sign for  $e_i$  that results in a totally unimodular signing of  $C_i$ . Thus if G has a totally unimodular signing then we can find such a signing of G by arbitrarily signing the edges of F and then signing each  $e_i$  in turn such that the edge signing restricted to  $C_i$  is totally unimodular. This algorithm is called *Camion's signing algorithm* and a  $\{0, 1\}$  matrix A is regular if and only if Camion's signing algorithm returns a totally unimodular signing of A.

The smallest example of a  $\{0, 1\}$  matrix that is not regular comes from the *Fano* matroid, denoted  $F_7$ . A partial representation for the Fano matroid is the matrix R, given below:

$$R = \begin{bmatrix} 1 & 1 & 0 & 1 \\ 1 & 0 & 1 & 1 \\ 0 & 1 & 1 & 1 \end{bmatrix}$$

The bipartite graph G with adjacency matrix R is shown in Figure 2.3. Camion's signing



Figure 2.3:  $G(F_7, B)$ 

algorithm applied to G and R results in a signing which is not totally unimodular, and thus R is not a regular matrix and the Fano matroid is not a regular matroid. A theorem of Tutte states that  $F_7$  and the dual of  $F_7$  are the only obstructions for binary matroids to be regular. **Theorem 2.4 (Tutte [43]).** A binary matroid M is regular if and only if M does not contain a minor isomorphic to the Fano or the dual of the Fano.

There is a short proof of Theorem 2.4 due to Gerards [14].

#### 2.6 Connectivity

We define the well known notion of connectivity for matroids and matroid pairs, and prove that the fundamental graph for a matroid pair (see Section 2.3) is strongly connected if and only if the matroid pair is connected.

A rank r matroid M on the ground set S is *connected* if

 $\operatorname{rank}_M(S_1) + \operatorname{rank}_M(S_2) > r$ 

for all partitions  $\{S_1, S_2\}$  of S with  $S_1 \neq \emptyset$  and  $S_1 \neq S$ . If  $\operatorname{rank}_M(S_1) + \operatorname{rank}_M(S_2) = r$  for some partition  $\{S_1, S_2\}$  of S, then  $S_1$  and  $S_2$  are separators of M. Clearly both the empty set and S are separators of M, and a separator  $S_1$  of M is called a *nontrivial separator* if  $S_1 \neq \emptyset$  and  $S_1 \neq S$ . If restricting M to a nontrivial separator gives a connected matroid, then the separator is a *component* of M. The components of the graphic matroid M(G) correspond to the blocks (2-connected components) of the graph G, and if M is a representable matroid with matrix representation A then the components of Mcorrespond to the blocks of A. When a matroid is not connected then there is a unique partition of its ground set into components. A common technique in proving results about matroids is to assume the matroids are connected. Here we give an analogous connectivity definition for matroid pairs and we show that for the problem of counting common bases in a pair of matroids, we may assume that the matroid pair is connected.

If P is the matroid pair (M, N), then by Edmonds' Matroid Intersection Theorem (Theorem 2.1), P has a basis if and only if  $\operatorname{rank}_M(S_1) + \operatorname{rank}_N(S_2) \ge r$  for all partitions  $\{S_1, S_2\}$  of S. We say that P is connected if

$$\operatorname{rank}_M(S_1) + \operatorname{rank}_N(S_2) > r$$

for all nontrivial partitions  $\{S_1, S_2\}$  of S. If  $\{S_1, S_2\}$  is a partition of S such that

$$\operatorname{rank}_M(S_1) + \operatorname{rank}_N(S_2) = r,$$

then  $S_1$  and  $S_2$  are called *separators* of P. In particular, if  $S_1 \subseteq S$  is a separator of M and a separator of N, then  $S_1$  is a separator of P, and if  $S_1$  is non-trivial then P is not connected. This notion can be generalized for higher connectivity: the matroid pair P = (M, N) is k-connected if

$$\operatorname{rank}_M(S_1) + \operatorname{rank}_N(S_2) > r + k - 1$$

for all partitions  $\{S_1, S_2\}$  satisfying  $|S_1| \ge k$  and  $|S_2| \ge k$ .

The next lemma shows that if we are interested in counting bases of a matroid pair and the pair is not connected, then we can restrict the counting problem to two smaller matroid pairs.

**Theorem 2.5.** Let P = (M, N) be a rank r matroid pair on the ground set S, and let  $\{S_1, S_2\}$  be a partition of S such that

$$\operatorname{rank}_M(S_1) + \operatorname{rank}_N(S_2) = r.$$

If  $U_1 \subseteq S_1$  and  $U_2 \subseteq S_2$ , then  $U_1 \cup U_2$  is a basis of P if and only if  $U_1$  is a basis of the pair  $(M \setminus S_2, N/S_2)$  and  $U_2$  is a basis of the pair  $(M/S_1, N \setminus S_1)$ .

*Proof.* Assume that  $U_1$  is a basis of  $M \setminus S_2$  and  $U_2$  is a basis of  $M/S_1$ , and let  $r_1 = \operatorname{rank}_M(S_1)$  and  $r_2 = \operatorname{rank}_{M/S_1}(S_2)$ . Then  $r_1 + r_2 = r$ , and  $|U_1 \cup U_2| = |U_1| + |U_2| = r$ . To show that  $U_1 \cup U_2$  is a basis of M it therefore suffices to show that  $U_1 \cup U_2$  has rank r in M.

Let  $X = S_1 \setminus U_1$  and let  $x \in X$ . Since

$$\operatorname{rank}_{M/U_1}(x) = \operatorname{rank}_M(U_1 \cup \{x\}) - \operatorname{rank}_M(U_1) = 0,$$

every element of X is a loop of  $M/U_1$  and therefore

$$\operatorname{rank}_{M/U_1}(U_2 \cup X) = \operatorname{rank}_{M/U_1}(U_2).$$
 (2.5)

Furthermore,

$$\operatorname{rank}_{M/U_1}(U_2 \cup X) = \operatorname{rank}_{(M/U_1)/X}(U_2) + \operatorname{rank}_{M/U_1}(X)$$
(2.6)  
=  $\operatorname{rank}_{M/S_1}(U_2) + 0$   
=  $r_2$ .

Combining Equations (2.5) and (2.6) gives that  $\operatorname{rank}_{M/U_1}(U_2) = r_2$ . Thus

$$\operatorname{rank}_{M}(U_{1} \cup U_{2}) = \operatorname{rank}_{M}(U_{1}) + \operatorname{rank}_{M/U_{1}}(U_{2})$$
$$= r_{1} + r_{2}$$
$$= r,$$

and therefore  $U_1 \cup U_2$  is a basis of M. By symmetry, if  $U_2$  is a basis of  $N \setminus S_1$  and  $U_1$  is a basis of  $N/S_2$ , then  $U_1 \cup U_2$  is a basis of N. Therefore if  $U_1$  is a basis of  $(M \setminus S_2, N/S_2)$  and  $U_2$  is a basis of  $(M/S_1, N \setminus S_1)$ , then  $U_1 \cup U_2$  is a basis of P.

Conversely, suppose that  $U_1 \cup U_2$  is a basis of P with  $U_1 \subseteq S_1$ ,  $U_2 \subseteq S_2$  and let  $r_1 = \operatorname{rank}_M(S_1)$  and  $r_2 = \operatorname{rank}_N(S_2)$ . Since  $\operatorname{rank}_M(S_1) + \operatorname{rank}_N(S_2) = e$  by assumption,  $r_1 + r_2 = r$ . If B is a basis of P, then  $|B \cap S_1| \leq r_1$  and  $|B \cap S_2| \leq r_2$ . Since

 $|B \cap S_1| + |B \cap S_2| = r$ , it follows that  $|B \cap S_1| = r_1$  and  $|B \cap S_2| = r_2$  for all bases B of P. In particular,  $|(U_1 \cup U_2) \cap S_i| = r_i$  and therefore  $|U_i| = r_i$  for i = 1, 2.

Since  $U_1$  is independent in  $M \setminus S_2$ ,  $\operatorname{rank}_{M \setminus S_2}(U_1) = |U_1| = r_1$  and thus  $U_1$  is a basis of  $M \setminus S_2$ . Furthermore,  $\operatorname{rank}_{M/S_1}(U_2) = \operatorname{rank}_M(U_2 \cup S_1) - \operatorname{rank}_M(S_1) = r - r_1 = r_2$ , and thus  $U_2$  is a basis of  $M/S_1$ . By symmetry,  $U_1$  is a basis of  $N/S_2$  and  $U_2$  is a basis of  $N \setminus S_1$ .

Let P = (M, N) be a rank r matroid on the ground set S and assume that  $\{S_1, S_2\}$  is a partition of S with rank<sub>M</sub> $(S_1) + \operatorname{rank}_N(S_2) \le r$ . If  $P^* = (M^*, N^*)$  is the dual matroid pair of P then  $P^*$  has rank  $r^* = |S| - r$ , and

$$\operatorname{rank}_{M^*}(S_2) + \operatorname{rank}_{N^*}(S_1) = (|S_2| - r + \operatorname{rank}_M(S_1)) + (|S_1| - r + \operatorname{rank}_N(S_2))$$
$$= |S| - 2r + \operatorname{rank}_M(S_1) + \operatorname{rank}_N(S_2)$$
$$\leq r^*.$$

**Observation 2.6.** The matroid pair P is connected if and only if the dual pair  $P^*$  is connected.

When considering whether a bipartite graph is Pfaffian, an edge which is not in any perfect matching can be deleted. In particular, if an edge is in every perfect matching then we can delete both vertices of the edge and consider the remaining graph. The matroid analog to an edge which is not in any perfect matching is a *loop* of a matroid pair: a loop of the matroid pair P is an element e of the ground set for P such that no basis of P contains e. Similarly, the matroid analog to an edge in every perfect matching of a graph is a *coloop*: an element e of the ground set is a coloop of P if every basis of P contains e. Note that a loop of P is a coloop of  $P^*$ .

Let S be the ground set for the rank r matroid pair P = (M, N) and suppose that  $e \in S$  is a loop of P. The maximum size of a common independent set of M/e and N/e is then r-2, and by Edmonds' Matroid Intersection Theorem (Theorem 2.1), there exists a partition  $\{S_1, S_2\}$  of  $S \setminus e$  such that  $\operatorname{rank}_{M/e}(S_1) + \operatorname{rank}_{N/e}(S_2) \leq r-2$ . Since  $\operatorname{rank}_M(S_1 \cup \{e\}) \leq \operatorname{rank}_{M/e}(S_1) + 1$  and  $\operatorname{rank}_N(S_2) \leq \operatorname{rank}_N(S_2 \cup \{e\}) \leq \operatorname{rank}_{N/e}(S_2) + 1$ , the partition  $\{S_1 \cup \{e\}, S_2\}$  of S satisfies

 $\operatorname{rank}_{M}(S_{1} \cup \{e\}) + \operatorname{rank}_{N}(S_{2}) \leq \operatorname{rank}_{M/e}(S_{1}) + 1 + \operatorname{rank}_{N/e}(S_{2}) + 1 \leq r,$ 

and therefore P is not connected. By duality, if  $e \in S$  is a coloop of P then P is not connected, and thus we have the following observation:

**Observation 2.7.** If a matroid pair is connected, then it has no loops and no coloops.

For an undirected graph G and a set  $R \subseteq V(G)$ , the *edge cut* of G induced by R is denoted  $\delta(R)$  and consists of all edges e of G such that exactly one end of e is in R. If G is a directed graph and  $R \subseteq V(G)$ , then we let

$$\delta(R) = \{ vw \in E : v \in R, w \notin R \}.$$

That is,  $\delta(R)$  denotes the set of edges directed out of R. The directed graph G is strongly connected if  $\delta(R) \neq \emptyset$  for all non-empty  $R \subset V(G)$ , or equivalently, if there exists a directed path from u to v for every  $u, v \in V(G)$ . Determining if a matroid pair P is connected is equivalent to determining if G(P) is strongly connected, as the next theorem shows.

**Theorem 2.8.** The matroid pair P is connected if and only if the fundamental graph G(P) is strongly connected. In particular, if P = (M, N) is a rank r matroid pair and  $\{S_1, S_2\}$  is a partition of the ground set S, then  $\operatorname{rank}_M(S_1) + \operatorname{rank}_N(S_2) = r$  if and only if  $\delta(S_2)$  is empty.

Proof. Assume that P = (M, N) is a rank r matroid pair on the ground set S, and let  $(R_M, R_N)$  be the partial representation of P with respect to the basis B of P. Suppose P is not connected and let  $\{S_1, S_2\}$  be a partition of S such that  $\operatorname{rank}_M(S_1) + \operatorname{rank}_N(S_2) \leq r$ . Since M and N have a common basis, it follows from Edmonds' Matroid Intersection Theorem that  $\operatorname{rank}_M(S_1) + \operatorname{rank}_N(S_2) = r$ . Let  $U_i = B \cap S_i$  for i = 1, 2 and let  $W_i = S_i \setminus B$ . By Theorem 2.5,  $U_1$  is a basis for the matroid pair  $(M \setminus S_2, N/S_2)$  and  $U_2$  is a basis for the matroid pair  $(M/S_1, N \setminus S_1)$ .

Suppose there exists  $u \in U_2, v \in W_1$  such that  $R_M[u; v] = 1$ . Then  $B \cup \{v\} \setminus \{u\}$  is a basis for M and therefore  $(B \cup \{v\} \setminus \{u\}) \cap S_1 = U_1 \cup \{v\}$  is independent in  $M \setminus S_2$ , which contradicts that  $U_1$  is a basis of  $M \setminus S_2$ . Thus every entry of  $R_M[U_2; W_1]$  is zero, which implies that there are no edges directed from  $U_2$  to  $W_1$  in G(P, B). By symmetry, every entry of  $R_N[U_1; W_2]$  is zero, and therefore there are no edges directed from  $W_2$  to  $U_1$  in G(P, B). Since  $\{U_1, U_2\}$  is a partition of B and  $\{W_1, W_2\}$  is a partition of  $S \setminus B$ , it follows that  $\delta(W_2 \cup U_2) = \emptyset$  and thus G(P, B) is not strongly connected. (See Figure 2.4.)



Figure 2.4: A directed cut in G(P, B)

Conversely, suppose G(P, B) is not connected and let  $\{U_1, U_2\}$  be a partition of Band  $\{W_1, W_2\}$  be a partition of  $S \setminus B$  such that  $\delta(U_2 \cup W_2) = \emptyset$ . If  $u \in U_2$  and  $v \in W_1$ then  $R_M[u; v] = 0$  and therefore  $B \cup \{v\} \setminus \{u\}$  is not a basis. Thus  $\operatorname{rank}_M(U_1 \cup W_1) =$  $\operatorname{rank}_M(U_1) \leq |U_1|$ . Symmetrically,  $\operatorname{rank}_N(U_2 \cup W_2) = \operatorname{rank}_N(U_2) \leq |U_2|$ . Thus

$$\operatorname{rank}_{M}(U_{1} \cup W_{1}) + \operatorname{rank}_{N}(W_{2}) \cup U_{2}) \le |U_{1}| + |U_{2}| = |B| = r$$

and P is not connected.

# Chapter 3 Pfaffian Matroid Pairs

In this chapter we define *Pfaffian matroid pairs*, which are the primary focus of this thesis. We begin by considering the problem of counting the number of common column bases between two matrices with the same set of columns, and show how the classical Cauchy-Binet formula solves this problem in some cases. The pairs of matrices for which the formula solves the counting problem are called *Pfaffian matrix pairs*. We then formulate the problem of counting bases in terms of matroids, and define Pfaffian matroid pairs in terms of Pfaffian matrix pairs. We give some characterizations of Pfaffian matroid pairs, and we show that the class of Pfaffian matroid pairs contains the class of Pfaffian bipartite graphs.

#### 3.1 Counting common bases

Let  $A_1$  and  $A_2$  be matrices with r rows and k columns. If the columns of  $A_1$  and  $A_2$  are indexed by S and  $A_i[X]$  denotes the column submatrix of  $A_i$  indexed by the columns  $X \subseteq S$ , then the classical Cauchy-Binet formula states that

$$\det A_1 A_2^{\top} = \sum_{X \subseteq S: |X| = r} \det A_1[X] \det A_2[X].$$
(3.1)

Since det  $A_1[X]$  det  $A_2[X] \neq 0$  if and only if X is a common basis of  $A_1$  and  $A_2$ , the non-zero terms in the right hand side of Equation (3.1) correspond to the common bases of  $A_1$  and  $A_2$ . If  $A_1$  and  $A_2$  are totally unimodular then

$$\det A_1[X] \det A_2[X] \in \{0, \pm 1\}$$

for all  $X \subseteq S$  with |X| = r and thus  $|\det A_1 A_2^{\top}|$  gives a lower bound on the number of common bases of  $A_1$  and  $A_2$ . If  $A_1$  and  $A_2$  totally unimodular and  $|\det A_1 A_2^{\top}|$  is equal to the number of common bases of  $A_1$  and  $A_2$ , then we call the pair  $(A_1, A_2)$  a *Pfaffian pair of matrices*. That is, the pair  $(A_1, A_2)$  of matrices is Pfaffian if both  $A_1$  and  $A_2$  are totally unimodular and either

- (i) det  $A_1[B] = \det A_2[B]$  for all common bases B of  $A_1$  and  $A_2$ , or
- (ii) det  $A_1[B] = -\det A_2[B]$  for all common bases B of  $A_1$  and  $A_2$ .

Let  $A_1$  be obtained from  $A_1$  by scaling some of the columns of  $A_1$  by a factor of -1, and let  $\tilde{A}_2$  be obtained from  $A_2$  by scaling some of the columns of  $A_2$  by -1. If  $(\tilde{A}_1, \tilde{A}_2)$  is a Pfaffian pair of matrices then we say that the matrix pair  $(A_1, A_2)$  has a *Pfaffian signing*.

Let P be the matroid pair (M, N). If  $(A_M, A_N)$  is a representation of P such that  $(A_M, A_N)$  is a Pfaffian matrix pair then  $(A_M, A_N)$  is a Pfaffian representation of P. By the Cauchy-Binet formula given in Equation (3.1), if  $(A_M, A_N)$  is a Pfaffian representation of P, then  $|\det A_M A_N^{\top}|$  is the number of bases of P. We say that a matroid pair is *Pfaffian* whenever it has a Pfaffian representation. If  $(A_M, A_N)$  is a Pfaffian representation of P satisfying det  $A_M[B] = -\det A_N[B]$  for all bases B of P, and  $A'_N$  is the representation obtained from  $A_N$  by scaling a row of  $A_N$ , then det  $A'_N[B] = \det A_M[B]$  for all bases B of P. Unless otherwise stated, we therefore assume that if  $(A_M, A_N)$  is a Pfaffian representation of P.

Let  $(A_M, A_N)$  be a totally unimodular representation of P. By Camion's theorem [3] for totally unimodular signings, given any other representation  $(A'_M, A'_N)$  with the same non-zero entries as  $(A_M, A_N)$ , the two representations are the same up to resigning across rows and columns. Let  $(A^i_M, A^i_N)$  be obtained from  $(A_M, A_N)$  by scaling the  $i^{th}$  columns of  $A_M$  and  $A_N$  by -1. If X indexes a square column submatrix in  $A_M$  and  $A_N$ , then

$$\det A_M^i[X] \det A_N[X] = \det A_M[X] \det A_N^i[X].$$

Thus if  $(A'_M, A'_N)$  is a scaling of  $(A_M, A_N)$  such that  $(A'_M, A'_N)$  is a Pfaffian representation of P, then a Pfaffian representation of P can be obtained from  $(A_M, A_N)$  by scaling  $A_N$ only.

Suppose  $A'_N$  is obtained from  $A_N$  by pivoting (Section 2.4). Then there exists a nonsingular matrix D such that  $A'_N = DA_N$  and therefore det  $A'_N[B] = \det D \det A_N[B]$ for all bases B. Furthermore,  $A'_N$  is a totally unimodular representation of N and thus if  $(A_M, A_N)$  is a Pfaffian representation of P then either det  $A_M[B] = \det A'_N[B]$  for all bases B of P or det  $A_M[B] = -\det A'_N[B]$  for all bases B of P. Therefore if  $A'_N$  is obtained from  $A_N$  by pivoting, then  $(A_M, A_N)$  is a Pfaffian representation of P if and only if  $(A_M, A'_N)$  is a Pfaffian representation of P.

If B is a basis of P and det  $A_M[B] = \det A_N[B]$ , then we say that B is correctly signed with respect to  $(A_M, A_N)$ . A basis which is not correctly signed with respect to  $(A_M, A_N)$  is incorrectly signed. We make the following observation:

**Observation 3.1.** If  $(A_M, A_N)$  is a totally unimodular representation of the matroid pair P, then P is Pfaffian if and only if there is a scaling  $A'_N$  of  $A_N$  such that every basis of P is correctly signed with respect to  $(A_M, A'_N)$ .

If  $(R'_M, R'_N)$  is a totally unimodular partial representation of P with respect to the basis B, then  $(R'_M, R'_N)$  is called a Pfaffian representation of P with respect to B if  $([I|R'_M], [I|R'_N])$  is a Pfaffian representation of P. The edge signing of G(P, B) corresponding to  $(R'_M, R'_N)$  is a called Pfaffian signing of G(P, B) if  $(R'_M, R'_N)$  is a Pfaffian representation of P, and thus P is Pfaffian if and only if G(P) has a Pfaffian signing. Note that  $(R'_M, R'_N)$  is a Pfaffian representation of P with respect to B if and only if det  $R'_M[U; W]$  det  $R'_N[U; W] \in \{0, 1\}$  whenever  $U \subseteq B$ ,  $W \subseteq S \setminus B$  and |U| = |W|. If s is the edge signing of G(P, B) corresponding to  $(R'_M, R'_M)$  and  $X \subseteq S$  is such that  $B\Delta X$ is a basis, then we say that G(P, B)[X] is correctly signed with respect to s if the basis  $B\Delta X$  is correctly signed with respect to  $(R'_M, R'_N)$ . This leads to the next observation.

**Observation 3.2.** If P is a matroid pair and s is a totally unimodular edge signing of G(P), then s is a Pfaffian signing of G(P) if and only if all of the contributing subgraphs of G(P) are correctly signed.

For Observation 3.2 to be useful, a characterization of a Pfaffian signing of the matroid pair P with respect to an edge signing of G(P) is required. This appears to be a difficult problem in general, with the exception of the graphic matroid pairs discussed in Section 3.3.

When s is an edge signing of the graph G and  $E' \subseteq E(G)$ , let

$$s(E') = \sum_{e \in E'} s_e$$

where the sum is taken over GF(2). In particular, if C is a circuit in G then s(C) is the parity of the number of edges in C that are signed 1. Recall from Section 2.3 that a simple contributing circuit in a fundamental graph is a directed circuit C in G(P) which is the unique circuit cover of V(C) in G(P). When s is a Pfaffian signing of G(P) and C is a simple contributing circuit in G(P) then we observe that s(C) depends only on |C|.

**Lemma 3.3.** If P is a matroid pair and C is a simple contributing circuit in G(P), then C is correctly signed with respect to the edge signing s if and only if

$$s(C) \equiv \frac{|C|}{2} + 1 \pmod{2}.$$

*Proof.* Assume P = (M, N) is a matroid pair on the ground set S with basis B and partial representation  $(R_M, R_N)$  with respect to B. Let s be an edge signing of G(P, B) and let  $(R'_M, R'_N)$  be the corresponding signing of  $(R_M, R_N)$ .

Let C be a simple contributing circuit in G(P, B) and define  $T_M(C)$  and  $T_N(C)$ to be the unique perfect matchings in G(M, B)[V(C)] and G(N, B)[V(C)]. Let  $A_M$ and  $A_N$  be the adjacency matrices for  $T_M(C)$  and  $T_N(C)$  respectively, and let  $A'_M$  and  $A'_N$  be the signings of  $A_M$  and  $A_N$  corresponding to s. Assume  $U = V(C) \cap B$  and  $W = V(C) \setminus B$ . Since there is a one-to-one correspondence between perfect matchings in  $G(M, B)[U \cup W]$  and terms in det  $R'_M[U; W]$ , the determinant of  $R'_M[U; W]$  is equal to det  $A'_M$ . Similarly, det  $R'_N[U; W] = \det A'_N$  and therefore C is correctly signed if and only if det  $A'_M = \det A'_N$ .

Assume that |C| = 2k and let  $U = \{u_1, u_2, \ldots, u_k\}$  and  $W = \{w_1, w_2, \ldots, w_k\}$ , where the cyclic order of the vertices on C is  $\{u_1, w_1, u_2, w_2, \ldots, u_k, w_k\}$ . By simultaneously permuting rows and columns in  $A_M$  and  $A_N$  we may assume that the  $i^{th}$  rows of  $A_M$  and  $A_N$  are indexed by  $u_i$  and that the  $i^{th}$  columns of  $A_M$  and  $A_N$  are indexed by  $w_i$ . Since  $T_M(C) = \{u_1w_1, u_2w_2, \ldots, u_kw_k\}$ ,  $A_M$  is the k by k identity matrix, and since  $T_N =$  $\{w_1u_2, w_2u_3, \ldots, w_ku_1\}$ ,  $A_N$  is a permutation matrix with determinant  $(-1)^{k+1}$ . Thus det  $A'_M = (-1)^{s(T_M)}$  and det  $A'_N = (-1)^{s(T_N)}(-1)^{(k+1)}$  and therefore det  $A'_M = \det A'_N$ if and only if  $s(T_M) = s(T_N) + (k+1)$ . Since  $s(T_M) + s(T_N) = s(C)$  and  $k = \frac{|C|}{2}$ , we conclude that C is correctly signed if and only if  $s(C) = \frac{|C|}{2} + 1$ .

If P is a matroid pair then Lemma 3.3 implies that it is easy to determine if an edge signing correctly signs the simple contributing circuits of G(P). However, there does not appear to be a similarly easy definition of the sign of a contributing circuit when the circuit in question is not simple.

#### **3.2** Properties of Pfaffian matroid pairs

It is well known that the class of graphic matroids is closed under taking minors, and that the class of regular matroids is closed under duality and under taking minors. In this section we show the class of Pfaffian matroid pairs to be closed under duality and under taking minors.

**Theorem 3.4.** If P is a matroid pair then P is Pfaffian if and only if  $P^*$  is Pfaffian.

Proof. Let M and N be matroids on the ground set S and let P be the matroid pair (M, N). Suppose that  $(A_M, A_N)$  is a Pfaffian representation of P and let B be a basis of P. We may assume that  $A_M = [I|R_M]$  and  $A_N = [I|R_N]$ , where the rows of  $A_M$  and  $A_N$  are indexed by B and the columns of  $R_M$  and  $R_N$  are indexed by  $S \setminus B$ . The matrices  $A_M^* = [R_M^\top | I]$  and  $A_N^* = [R_N^\top | I]$  are then totally unimodular representations of  $M^*$  and  $N^*$  respectively. If  $B^*$  is a basis of  $P^*$  and  $\overline{B^*} = S \setminus B^*$ , then  $\overline{B^*}$  is a basis of P and

$$\det A_M^*[B^*] = \det R_M^{\top}[(S \setminus B) \setminus B^*; B \cap B^*]$$
$$= \det R_M[B \cap B^*; (S \setminus B) \setminus B^*]$$
$$= \det R_M[B \setminus \overline{B^*}; \overline{B^*} \setminus B].$$

Similarly, det  $A_N^*[B^*] = \det R_N[B \setminus \overline{B^*}; \overline{B^*} \setminus B]$ . Since  $(A_M, A_N)$  is a Pfaffian representation of P and  $\overline{B^*}$  is a basis of P, det  $R_M[B \setminus \overline{B^*}; \overline{B^*} \setminus B] = \det R_N[B \setminus \overline{B^*}; \overline{B^*} \setminus B]$ . Thus  $(A_M^*, A_N^*)$  is a Pfaffian representation of  $P^*$ , and P is Pfaffian if and only if  $P^*$  is Pfaffian.

Let P be the matroid pair P = (M, N) on the ground set S and let  $B \subseteq S$  be a basis of P. If  $U \subseteq S \setminus B$  and  $V \subseteq B$ , then the matroid pair  $(M \setminus U/V, N \setminus U/V)$  is called a *minor* of P and is denoted  $P \setminus U/V$ . Since  $((M \setminus U/V)^*, (N \setminus U/V)^*) = (M^*/U \setminus V, M^*/U \setminus V)$ , the dual of  $P \setminus U/V$  is  $P^*/U \setminus V$ .

**Theorem 3.5.** All minors of a Pfaffian matroid pair are Pfaffian.

*Proof.* Let S be the ground set for the matroid pair P = (M, N) and assume  $u \in S$  is not a coloop of P. If  $(A_M, A_N)$  is a Pfaffian representation of P and  $A'_M$  and  $A'_N$  are obtained from  $A_M$  and  $A_N$  respectively by deleting the columns indexed by u, then for all bases B of  $(M \setminus u, N \setminus u)$ ,

$$\det A'_M[B] = \det A_M[B] = \det A_N[B] = \det A'_N[B].$$

Thus  $(A'_M, A'_N)$  is a Pfaffian representation for  $P \setminus u$ . By Theorem 3.4, the pair P/v is Pfaffian whenever v is not a loop of P and thus all minors of P are Pfaffian.

Let P be a non-Pfaffian matroid pair on the ground set S. If  $P \setminus u$  is Pfaffian whenever  $u \in S$  is not a coloop of P and P/v is Pfaffian whenever  $v \in S$  is not a loop of P, then P is minimally non-Pfaffian.

#### 3.3 Pfaffian bipartite graphs

Let G be a bipartite graph and let A be the adjacency matrix for G. Since there is a one-to-one correspondence between perfect matchings in G and nonzero terms in the permutation expansion of det A, the number of perfect matchings in G is at least  $|\det A|$ . The graph G is a *Pfaffian bipartite graph* if there exists a signing A' of A such that det A' is exactly the number of perfect matchings in G.

The complete bipartite graph  $K_{3,3}$  has exactly six perfect matchings, and the adjacency matrix A for  $K_{3,3}$  is the 3 by 3 matrix of all ones. Since it is not possible to replace some of the entries of A with -1 such that the new matrix has determinant  $\pm 6$ , the graph  $K_{3,3}$  is non-Pfaffian. An odd subdivision of  $K_{3,3}$  is obtained by possibly replacing some of the edges of  $K_{3,3}$  with paths of odd length, and every odd subdivision of  $K_{3,3}$ is also non-Pfaffian. If the subgraph H of G is an odd subdivision of  $K_{3,3}$  and  $G \setminus V(H)$ has a perfect matching, then G is said to contain  $K_{3,3}$ . It is straightforward to show that a bipartite graph that contains  $K_{3,3}$  is non-Pfaffian (see Robertson, Seymour, and Thomas [35]), and a theorem of Little [22] shows that the converse also holds.

**Theorem 3.6 (Little).** A bipartite graph is Pfaffian if and only if it does not contain  $K_{3,3}$ .

In this section we present the well known application of matroid intersection as a method of finding a perfect matching in a bipartite graph, and we show that a bipartite graph is Pfaffian if and only if the associated matroid pair is Pfaffian. In Chapter 6 we will show that the graphic matroid pairs for Pfaffian bipartite graphs are a subclass of series-parallel Pfaffian matroid pairs.

If G is a bipartite graph with vertex bipartition  $\{U, W\}$  then define the matroids  $M_U$ and  $M_W$  on the ground set E(G) such that  $E' \subseteq E(G)$  is independent in  $M_U$  if and only if each vertex in U is incident to at most one edge in E' and  $E' \subseteq E(G)$  is independent in  $M_W$  if and only if each vertex in W is incident to at most one edge in E'. The matroids  $M_U$  and  $M_W$  are referred to as the *partition matroids* defined by the bipartition of V(G). The matroid pair  $P = (M_U, M_W)$  is denoted P(G) and is called the *graphic matroid pair* for G.

If  $E' \subseteq E(G)$  is independent in  $M_U$  and in  $M_W$ , then E' is a matching of G, and thus the perfect matchings in G correspond to the bases of P(G). Let  $A_U$  be the  $U \times E(G)$ incidence matrix and let  $A_W$  be the  $W \times E(G)$  incidence matrix. If A is the adjacency matrix for G, then

$$A = A_U A_W^\top.$$

Furthermore, scaling column e of  $A_W$  by -1 corresponds to replacing the entry for e in A with -1, and thus A has a signing A' such that det A' is the number of perfect matchings in G if and only if  $A_W$  has a scaling  $A'_W$  such that the determinant of  $A_U(A'_W)^{\top}$  is the number of perfect matchings in G.

**Observation 3.7.** The bipartite graph G is Pfaffian if and only if the graphic matroid pair P(G) is Pfaffian, and thus the class of Pfaffian matroid pairs extends the class of Pfaffian bipartite graphs.

A graph is called *matching covered* if it is connected and every edge in the graph is contained in a perfect matching. It is natural to restrict the study of Pfaffian graphs to matching covered graphs, and the following theorem shows that this is equivalent to restricting graphic matroid pairs to connected matroid pairs.

**Theorem 3.8.** Let G be a bipartite graph with a perfect matching. The graphic matroid pair P(G) is connected if and only if G is matching covered.

*Proof.* Let  $V(G) = \{U, W\}$  be the vertex bipartition of the bipartite graph G, and let M and N be the partition matroids defined on U and W respectively such that P(G) is the graphic matroid pair (M, N). Since G has a perfect matching, |U| = |W| and P(G) has rank equal to |U|.

Suppose that P(G) is not connected, and let  $\{S_1, S_2\}$  be a non trivial partition of E(G) such that

$$\operatorname{rank}_{M}(S_{1}) + \operatorname{rank}_{N}(S_{2}) = |U|.$$
(3.2)

Since rank<sub>M</sub>(S<sub>1</sub>) is the number of vertices of U that are incident to the edges S<sub>1</sub> and rank<sub>N</sub>(S<sub>2</sub>) is the number of vertices of W that are incident to the edges S<sub>2</sub>, Equation (3.2) implies that there exist sets  $U_1 \subseteq U$  and  $W_1 \subseteq W$  such that  $|U_1| + |W_1| = |U|$  and

$$\delta(U_1) \cup \delta(W_1) = E(G).$$

If  $U_2 = U \setminus U_1$ , then since |U| = |W|,  $|U_2| = |W_1|$ , and thus every perfect matching of G induces a perfect matching of  $G[U_2 \cup W_1]$ . It follows that either  $\delta(U_2 \cup W_1) = \emptyset$  and G is not connected, or  $\delta(U_2 \cup W_1) \neq \emptyset$  and every edge in  $\delta(U_2 \cup W_1)$  is not in any perfect matching of G. Hence if P(G) is not connected, then G is not matching covered.

Conversely, suppose that G is not matching covered. If G has an edge e such that e is not in any perfect matching of G, then e is not in any basis of P(G), and thus by definition e is a loop of P(G). By Observation 2.7, P(G) is not connected in this case. Similarly, if G is not connected and  $S_1 \subset E(G)$  is such that the subgraph of G induced by  $S_1$  is a component of G, then

$$\operatorname{rank}_M(S_1) + \operatorname{rank}_N(S_2) = |U|$$

and again P(G) is not connected.

#### 3.4 Certifying non-Pfaffian pairs

Given a Pfaffian representation for a matroid pair, either every basis is correctly signed with respect to the representation, or every basis is incorrectly signed. Thus, if a set of bases of a matroid pair can neither be simultaneously correctly signed, nor simultaneously incorrectly signed, then the matroid pair is non-Pfaffian. We call such a collection of bases a *non-Pfaffian bases certificate*. Since a matroid pair can have an exponential number of bases with respect to the size of the ground set, a non-Pfaffian bases certificate need not be an efficient way to show that the matroid pair is non-Pfaffian. In this section we prove that if a matroid pair is non-Pfaffian, then it has a non-Pfaffian bases certificate where the number of bases in the certificate is linear with respect to the ground set for matroid pair.

Let P be a regular matroid pair on the ground set S and let  $(A_M, A_N)$  be a totally unimodular representation of P. Let  $X \subseteq S$  and let  $A'_N$  be the matrix obtained by scaling by a factor of -1 the columns of  $A_N$  indexed by X. If B is a basis of N, then  $\det A_N[B] = -\det A'_N[B]$  if and only if  $|X \cap B|$  is odd. Thus P is Pfaffian if and only if there exists  $X \subseteq S$  satisfying one of the following:

(i)  $|X \cap B|$  is even for all correctly signed bases B of P and  $|X \cap B'|$  is odd for all incorrectly signed bases B',

- (ii)  $|X \cap B|$  is odd for all correctly signed bases B of P and  $|X \cap B'|$  is even for all incorrectly signed bases B'.
- If P has odd rank and  $X \subseteq S$  satisfies (ii), then over GF(2),

$$|(S \setminus X) \cap B| = |X \cap B| + |B| = |X \cap B| + 1$$

and thus P is Pfaffian if and only if there exists  $X \subseteq S$  satisfying (i). However, a Pfaffian matroid pair P may not have a solution to (i) when P has even rank. For example, let P = (M, N) where M and N have totally unimodular representations  $A_M$  and  $A_N$  as given below

$$A_M = \begin{bmatrix} 1 & 0 & 1 & 1 & 0 \\ 0 & 1 & 1 & 0 & 1 \end{bmatrix}, \quad A_N = \begin{bmatrix} 1 & 0 & 1 & 1 & 0 \\ 0 & 1 & 0 & 1 & 1 \end{bmatrix}.$$
 (3.3)

The correctly signed bases of P with respect to the column indices are

$$\{1,2\}, \{1,5\}, \{2,3\}, \{2,4\}, \{3,5\}, \{4,5\}$$

and the only incorrectly signed basis is  $\{3, 4\}$ . Although there is no subset of  $\{1, 2, 3, 4, 5\}$  which intersects all correctly signed bases with even parity and intersects the incorrectly signed basis with odd parity, P is a Pfaffian matroid pair: scaling  $A_N$  across columns 3 and 4 and any row gives a Pfaffian representation of P.

Let  $(A_M, A_N)$  be a totally unimodular representation of the matroid pair P and suppose that there exists a set of correctly signed bases  $\{B_1, B_2, \ldots, B_{2k+1}\}$  of P and a set of incorrectly signed bases  $\{B'_1, B'_2, \ldots, B'_{2m+1}\}$  of P such that

$$B_1 \Delta B_2 \Delta \cdots \Delta B_{2k+1} \Delta B'_1 \Delta B'_2 \Delta B'_{2m+1} = \emptyset.$$
(3.4)

For  $S_1, S_2 \subseteq S$  and  $X \subseteq S$ ,

$$|X \cap S_1| + |X \cap S_2| = |X \cap (S_1 \Delta S_2)| + 2|X \cap (S_1 \cap S_2)|,$$

and therefore  $|X \cap S_1| + |X \cap S_2|$  has the same parity as  $|X \cap (S_1 \Delta S_2)|$ . Thus for all  $X \subseteq S$ , Equation (3.4) implies that over GF(2),

$$\sum_{i=1}^{2k+1} |X \cap B_i| + \sum_{i=1}^{2m+1} |X \cap B'_i| = |X \cap (B_1 \Delta B_2 \Delta \cdots \Delta B_{2k+1} \Delta B'_1 \Delta B'_2 \Delta B'_{2m+1})|$$
$$= |X \cap \emptyset|$$
$$= 0.$$

If  $X \subseteq S$  is such that  $|X \cap B|$  is even for all correctly signed bases B and  $|X \cap B'|$  is odd for all incorrectly signed bases B', then

$$\sum_{i=1}^{2k+1} |X \cap B_i| + \sum_{i=1}^{2m+1} |X \cap B'_i| = 1.$$
(3.5)

Equation (3.5) similarly holds if X intersects all correctly signed bases with odd cardinality and intersects all incorrectly signed bases with even cardinality. Thus the union of an odd number of correctly signed bases with an odd number of incorrectly signed bases which together satisfy (3.4) is a non-Pfaffian bases certificate for P. The following observation will be used to prove that a non-Pfaffian bases certificate of linear size with respect to the ground set exists for any non-Pfaffian matroid pair.

**Observation 3.9.** If A is an  $m \times n$  matrix over GF(2) and b is an m dimensional vector over GF(2), then exactly one of the following holds:

- 1. Ax = b has a solution.
- 2. There exists  $y \in \{0,1\}^m$  such that  $y^{\top}A = 0$  and  $y^{\top}b = 1$ .

Certainly if  $y^{\top}A = 0$  and  $y^{\top}b = 1$  then there is no solution to Ax = b, and thus a vector y satisfying  $y^{\top}A = 0$  and  $y^{\top}b = 1$  provides a certificate that Ax = b does not have a solution. The *support* of y, denoted  $\operatorname{supp}(y)$  is the set of indices i for which  $y_i$  is nonzero. The following lemma shows that if Ax = b has no solution over GF(2), then there exists a certifying vector y such that the size of the support of y is at most one more than the number of columns in A.

**Lemma 3.10.** If A is an  $m \times n$  matrix over GF(2) and  $b \in GF(2)^m$ , then either Ax = b has a solution over GF(2) or there exists  $y \in \{0,1\}^m$  such that  $y^{\top}A = 0, y^{\top}b = 1$ , and  $|\operatorname{supp}(y)| \leq n+1$ .

*Proof.* If B is a matrix with k rows and  $S \subseteq \{1, 2, ..., k\}$ , let  $B_S$  denote the restriction of B to the rows in S. Suppose Ax = b has no solution over GF(2). By Observation 3.9, there exists  $y \in \{0, 1\}^m$  such that  $y^{\top}A = 0$  and  $y^{\top}b = 1$ . If S = supp(y) and **1** is the vector of all ones, then  $\mathbf{1}^{\top}A_S = 0$  and  $\mathbf{1}^{\top}b_S = 1$ . Since  $A_S$  has n columns, the rank of  $A_S$  is at most n.

Suppose  $|\operatorname{supp}(y)| > n + 1$ . Then  $A_S$  has more than n + 1 rows and thus there is a nontrivial subset  $S_1$  of S such that the rows of  $A_{S_1}$  are dependent, and therefore  $\mathbf{1}^{\top}A_{S_1} = 0$ . If  $S_2 = S \setminus S_1$  then

$$\mathbf{1}^{\top}A_S = \mathbf{1}^{\top}A_{S_1} + \mathbf{1}^{\top}A_{S_2} = 0$$

and therefore  $\mathbf{1}^{\top} A_{S_2} = 0$ . Similarly,

$$\mathbf{1}^{\top}b_S = \mathbf{1}^{\top}b_{S_1} + \mathbf{1}^{\top}b_{S_2} = 1$$

and thus either  $\mathbf{1}^{\top}b_{S_1} = 1$  or  $\mathbf{1}^{\top}b_{S_2} = 1$ . Without loss of generality, assume  $\mathbf{1}^{\top}b_{S_1} = 1$ . If  $z \in \{0,1\}^m$  is defined by  $z_i = 1$  if and only if  $i \in S_1$ , then  $z^{\top}A = \mathbf{1}^{\top}A_{S_1} = 0$  and  $z^{\top}b = \mathbf{1}^{\top}b_{S_1} = 1$ . Thus  $|\operatorname{supp}(y)| > n + 1$  implies that there exists  $z \in \{0,1\}^m$  with  $z^{\top}A = 0, z^{\top}b = 1$ , and  $|\operatorname{supp}(z)| < |\operatorname{supp}(y)|$ . The lemma follows.  $\Box$  We apply Lemma 3.10 to prove the main theorem of this section: if P is a non-Pfaffian matroid pair then P has a non-Pfaffian bases certificate of linear size with respect to the ground set of P.

Let A be the matrix whose rows are the characteristic vectors for the bases of P. Given a totally unimodular representation  $(A_M, A_N)$  of P, let b be the  $\{0, 1\}$  vector indexed by the bases of P with

$$b_B = \begin{cases} 1, & \text{if } B \text{ is correctly signed with respect to } (A_M, A_N); \\ 0, & \text{else.} \end{cases}$$

If  $\bar{b} \equiv b + 1 \pmod{2}$ , then P is Pfaffian if and only if either Ax = b or  $Ax = \bar{b}$  has a solution over GF(2). By Lemma 3.10, if there is no solution to Ax = b then there exists a vector y such that  $y^{\top}A = 0$  and  $y^{\top}b = 1$ . This implies that there is a set of bases which can not be simultaneously correctly signed and the number of incorrectly signed bases in the set is odd. Lemma 3.10 further implies that if such a set exists, then there is such a set with at most |S| + 1 bases. If the number of correctly signed bases is also odd, then this set certifies that P is non-Pfaffian, and hence there is a non-Pfaffian bases certificate with at most |S| + 1 bases.

If Ax = b also has no solution then there is a set of at most |S| + 1 bases with an odd number of correctly signed bases whose symmetric difference is the empty set. Either this set has an odd number of incorrectly signed bases and therefore certifies that P is non-Pfaffian, or the set for Ax = b can be combined with the set for  $Ax = \overline{b}$  to get a non-Pfaffian bases certificate with at most 2|S| + 2 bases. Thus the following theorem holds:

**Theorem 3.11.** If P is a non-Pfaffian matroid pair on the ground set S, then P has a non-Pfaffian bases certificate with at most 2|S| + 2 bases.

It follows that the problem of determining if a matroid pair is Pfaffian is in Co-NP: for any matroid pair that is non-Pfaffian, there is a certificate which, in polynomial time with respect to the size of the ground set, verifies that the pair is non-Pfaffian. Theorem 3.11 can be compared with Little's theorem (see Section 3.3) and the problem of determining if a bipartite graph is Pfaffian. By Little's theorem, if a bipartite graph is non-Pfaffian then it has a  $K_{3,3}$  minor which certifies that it is non-Pfaffian. In 1999, a decomposition theorem of Robertson, Seymour, and Thomas showed that the problem of determining if a bipartite graph is Pfaffian is also in NP. We do not know if determining if the problem of determining if a matroid pair is Pfaffian is similarly in NP.

In Section 6.6 we give a class of matroid pairs for which the linear bound in the size of the non-Pfaffian certificate can be improved to a bound of constant size. We conjecture that the linear bound in Theorem 3.11 can be improved to a constant bound for all matroid pairs. **Conjecture 3.12.** There exists a constant c such that every non-Pfaffian regular matroid pair has a non-Pfaffian bases certificate containing at most c bases.

Conjecture 3.12 is true for graphic matroid pairs: if G is a bipartite graph, then Little's Theorem [22] implies that the graphic matroid P(G) has a non-Pfaffian bases certificate containing exactly six bases.

#### 3.5 Examples

We give three examples of Pfaffian matroid pairs and three examples of non-Pfaffian pairs, and note the significance of each. Each of the non-Pfaffian matroid pairs is minimally non-Pfaffian.

#### Example 1: trivial Pfaffian pair.

A simple example of a Pfaffian matroid pair is the matroid pair P = (M, M) where M is any regular matroid. If  $A_M$  is any totally unimodular representation of M then  $(A_M, A_M)$  is clearly a Pfaffian representation of P. It follows that the number of bases of a regular matroid pair can be easily determined.

#### Example 2: twined $K_4$ pair.

Let  $D_3$  be the graph composed of three circuits, each of length 2, and let  $K_4$  be the complete graph on four vertices. Consider the matroid pair  $P = (M(D_3), M(K_4))$ , where the edges in  $D_3$  and  $K_4$  are labeled as shown in Figure 3.1(a) and 3.1(b) respectively.



Figure 3.1: edge labels for  $D_3$  and  $K_4$ 

We refer to P as a *twined*  $K_4$  *pair*. The matrices  $A_M$  and  $A_N$  below are totally unimodular representations of  $M(D_3)$  and  $M(K_4)$  respectively:

$$A_M = \begin{bmatrix} 1 & 0 & 0 & 1 & 0 & 0 \\ 0 & 1 & 0 & 0 & 1 & 0 \\ 0 & 0 & 1 & 0 & 0 & 1 \end{bmatrix}, \qquad A_N = \begin{bmatrix} 1 & 0 & 0 & 0 & 1 & 1 \\ 0 & 1 & 0 & -1 & 0 & 1 \\ 0 & 0 & 1 & -1 & -1 & 0 \end{bmatrix}.$$

For the basis B of P corresponding to the first three columns of  $A_M$  and  $A_N$ , the graph G(P, B) is shown in Figure 3.2. Since G(P, B) is strongly connected, Theorem 2.8 implies
that P is a connected matroid pair. Note that P has four bases, namely  $B, B_1 = \{1, 5, 6\},\$ 



Figure 3.2: G(P, B)

 $B_2 = \{2, 4, 6\}$  and  $B_3 = \{3, 4, 5\}$ . The determinant of  $A_M A_N^T$  is 4 and therefore  $(A_M, A_N)$  is a Pfaffian representation of P. If

$$A'_N = \begin{bmatrix} 1 & 0 & 0 & 0 & -1 & -1 \\ 0 & 1 & 0 & 1 & 0 & -1 \\ 0 & 0 & 1 & 1 & 1 & 0 \end{bmatrix},$$

then  $A'_N$  is obtained from  $A_N$  by scaling columns 4, 5, and 6 by -1, and since the determinant of  $A_M A'^T_N$  is 4, the representation  $(A_M, A'_N)$  of P is also a Pfaffian representation. The twined  $K_4$  pair has a pivotal role in our analysis of unique signings (Section 3.6) and binary spaces (Chapter 4).

#### Example 3: skew-symmetric matroid pair.

A matrix A is *skew-symmetric* if

$$A^{\top} = -A.$$

Note that if A is an  $n \times n$  skew-symmetric matrix, then

$$\det A = \det A^T = \det(-A) = (-1)^n \det A$$

and thus skew-symmetric matrices have even rank.

Let M be the regular matroid with representation [I|I] and let N be a regular matroid with totally unimodular representation [I|A] for some skew-symmetric matrix A. We call the matroid pair P = (M, N) a skew-symmetric matroid pair. The bases of P correspond to those square submatrices of I and A that are nonsingular in both. Since the only nonsingular submatrices of I are principal submatrices, the bases of P correspond to principal nonsingular submatrices of A. It can be shown that skew-symmetric matrices have non-negative determinant, and since a principal submatrix of a skew-symmetric matrix is skew-symmetric, det  $A[X;X] \ge 0$  for all row and column subsets X. Furthermore, det I[X;X] = 1 for all row and column subsets X, and thus the representation ([I|I], [I|A]) is a Pfaffian representation of P. From the observation above that all nonsingular skew-symmetric matrices have even rank, it follows that all bases of P intersect the columns of A an even number of times, and therefore resigning across these columns gives another Pfaffian representation of P. We note that this example is a generalization of Example 2.

#### Example 4: smallest non-Pfaffian pair.

Let P be the regular matroid pair on the ground set  $\{1, 2, 3, 4, 5\}$  with totally unimodular representation  $(A_M, A_N)$  as shown below:

$$A_M = \begin{bmatrix} 1 & 0 & 1 & 0 & 1 \\ 0 & 1 & 0 & 1 & 1 \end{bmatrix}, \qquad A_N = \begin{bmatrix} 1 & 0 & 0 & 1 & 1 \\ 0 & 1 & 1 & 0 & 1 \end{bmatrix}.$$

Note that the submatrix indexed by columns 3 and 4 has determinant +1 in  $A_M$  and determinant -1 in  $A_N$ , and thus the basis  $\{3,4\}$  of P is incorrectly signed. The matroid pair P has six bases, three of which are incorrectly signed and three which are correctly signed.

Correctly signed Bases:	$\{1,2\},\{1,5\},\{2,5\}$
Incorrectly signed Bases:	$\{3,4\},\{3,5\},\{4,5\}$

Since the symmetric difference of the three correctly signed bases is the empty set and the symmetric difference of the three incorrectly signed bases is also the empty set, the set of all bases of P forms a non-Pfaffian bases certificate for P. With respect to rank and size of the ground set, this example and its dual are the smallest non-Pfaffian matroid pairs.

#### Example 5: $K_{3,3}$ .

If M and N are the two partition matroids defined by the vertex bipartition of  $K_{3,3}$ (Section 3.3) then  $A_M$  and  $A_N$  shown in (3.6) are totally unimodular representations of M and N respectively. Since  $K_{3,3}$  is a non-Pfaffian graph, the matroid pair P = (M, N)is non-Pfaffian by Observation 3.7. The six bases of P correspond to the six perfect matchings of  $K_{3,3}$ , and since each edge in  $K_{3,3}$  is in exactly two perfect matchings, the symmetric difference of these six bases is the empty set. Under the representation  $(A_M, A_N)$ , three of the bases are correctly signed and three are incorrectly signed.

$$A_{M} = \begin{bmatrix} 1 & 0 & 0 & 1 & 0 & 0 & 0 & 1 & 0 \\ 0 & 1 & 0 & 0 & 1 & 1 & 0 & 0 & 0 \\ 0 & 0 & 1 & 0 & 0 & 0 & 1 & 0 & 1 \end{bmatrix} \qquad A_{N} = \begin{bmatrix} 1 & 0 & 0 & 0 & 1 & 0 & 0 & 0 & 1 \\ 0 & 1 & 0 & 1 & 0 & 0 & 1 & 0 & 0 \\ 0 & 0 & 1 & 0 & 0 & 1 & 0 & 1 & 0 \end{bmatrix}$$
(3.6)

Correctly signed Bases:	$\{1, 2, 3\}, \{4, 6, 9\}, \{5, 7, 8\}$
Incorrectly signed Bases:	$\{1, 6, 7\}, \{2, 8, 9\}, \{3, 4, 5\}$

The graph of G(P) in Figure 3.3 is an example of a double cycle. Double cycles and



Figure 3.3: G(P) corresponding to  $K_{3,3}$ 

their relationship to the problem of counting bases are analyzed in Section 6.7.

#### Example 6: neutral circuit.

Let the matroid pair P have basis  $B = \{1, 2, 3, 4, 5, 6\}$  and partial representation  $(R_M, R_N)$  with respect to B, as shown in (3.7). Note that G(P) shown in Figure (3.4) has a Hamiltonian circuit C whose vertices can be covered with two disjoint induced directed circuits and whose vertices can also be covered with three disjoint induced directed circuits. A *neutral* circuit in a directed graph is a directed circuit whose vertices can be covered with an even number of disjoint contributing circuits. Thus the Hamiltonian circuit C in G(P) is neutral.

$$R_{M} = \begin{bmatrix} 1 & 0 & 0 & 0 & 0 & 0 \\ 0 & 1 & 0 & 0 & 0 & 0 \\ 0 & 0 & 1 & 0 & 0 & 0 \\ 0 & 0 & 0 & 1 & 0 & 0 \\ 0 & 0 & 0 & 0 & 1 & 0 \\ 0 & 0 & 0 & 0 & 0 & 1 \end{bmatrix} \qquad R_{N} = \begin{bmatrix} 0 & 0 & 1 & 0 & 0 & 1 \\ 1 & 0 & 0 & 0 & 1 & 0 \\ 0 & 1 & 0 & 1 & 0 & 0 \\ 0 & 1 & 0 & 1 & 0 & 0 \\ 1 & 0 & 0 & 0 & 1 & 0 \end{bmatrix}$$
(3.7)

Correctly signed Bases:	$\{1, 2, 3, 4, 5, 6\}, \{1, 2, 3, 10, 11, 12\}, \{4, 5, 6, 7, 8, 9\}$
Incorrectly signed Bases:	$\{1, 2, 5, 6, 9, 10\}, \{1, 3, 4, 6, 8, 11\}, \{2, 3, 4, 5, 7, 12\}$

In Theorem 6.20 we show that if M and N are series-parallel matroids and G(M, N) has a neutral circuit, then the matroid pair (M, N) is non-Pfaffian and has an non-Pfaffian bases certificate containing at most 32 bases. In this example, P has a non-Pfaffian bases certificate with six bases.



Figure 3.4: A neutral circuit

## 3.6 Unique signing

In Section 3.3 we defined a Pfaffian bipartite graph to be a bipartite graph G for which there exists a signing A' of the adjacency matrix A for G such that the determinant of A' is equal to the number of perfect matchings of G. This signing of A corresponds to a  $\{0,1\}$  signing of the edges of G, where an edge is signed 1 if and only if the corresponding entry of A is signed -1. Given a perfect matching T of G, and an edge signing of s of G, call a T-alternating circuit C of G odd with respect to s if the sum of s on the edges of C is odd. Pfaffian bipartite graphs can equivalently be defined as the bipartite graphs with an edge signing such that with respect to a fixed perfect matching T of G, all T alternating circuits in the graph are odd. Similar to Camion's [3] observation that a totally unimodular signing of a  $\{0,1\}$  matrix is unique up to resigning across rows and columns, it follows from this definition of a Pfaffian bipartite graph that resigning a Pfaffian signing across cuts in G creates another Pfaffian signing. The Pfaffian signing can therefore be fixed on a spanning tree of G, and this partial signing uniquely determines the sign on all of the remaining edges which are in an T alternating circuit. Edges not in such a circuit can be removed from the graph without affecting whether G is Pfaffian, and so we may assume that all edges in G are in a perfect matching.

Up to resigning across cuts, a Pfaffian bipartite graph therefore has a unique Pfaffian signing. Similarly, up to resigning across rows and columns, a regular matrix has a unique totally unimodular signing. This uniqueness in both cases is critical for constructing an algorithm to find either a Pfaffian signing of a bipartite graph or a totally unimodular signing of a regular matrix. For a Pfaffian matroid pair there are two types of trivial resignings of a Pfaffian representation of the pair, which we describe next. In this section we consider whether a Pfaffian representation is unique up to such trivial resignings.

Let P = (M, N) be a rank r Pfaffian matroid pair on the ground set S and let  $(R_M, R_N)$  be a Pfaffian representation of P with respect to the basis B. The first trivial resigning of  $(R_M, R_N)$  occurs when P is not connected.

Suppose that  $\{S_1, S_2\}$  is a non-trivial partition of S such that

$$\operatorname{rank}_M(S_1) + \operatorname{rank}_N(S_2) = r.$$

By Theorem 2.5, if  $B' \subseteq S$  is a basis of P then  $B' \cap S_1$  is a basis of  $M \setminus S_2$  and thus  $|B' \cap S_1| = \operatorname{rank}_M(S_1)$  for all bases B' of P. It follows that if  $R'_N$  is the representation obtained from  $R_N$  by scaling the rows and columns corresponding to  $S_1$ , then  $(R_M, R'_N)$  is a Pfaffian representation of P. The edge resigning of G(P) corresponding to the signing  $R'_N$  resigns the edges from G(N) in the cut  $\delta(S_1)$ . Theorem 2.5 proved that these edges correspond to a directed cut in G(P), and therefore there are no directed circuits in G(P) that intersect  $\delta(S_1)$ . Thus such a resigning does not affect the sign on any directed circuit in G(P).

The second trivial resigning occurs when at least one of the matroids in the pair is not connected. Let  $X \subset S$  index a component of M, and let  $R'_N$  be obtained from  $R_N$  by resigning across the rows and columns corresponding to X. Since  $|X \cap B'| = \operatorname{rank}_M(X)$ for all bases B' of P,

$$\det R_N[U;W] = (-1)^{\operatorname{rank}_M(X)} \det R_N[U;W]$$

1

whenever  $(B \setminus U) \cup W$  is a basis of P, and therefore  $(R_M, R'_N)$  is a Pfaffian representation of P. Graphically, this resigning corresponds to resigning the edges of G(N) across the cut  $\delta(X)$  in G(P). Since X is a component of G(M), all the edges of  $\delta(X) \cup \delta(S \setminus X)$  are edges of G(N) and thus all the edges with exactly one end in X are resigned. It again follows that such a resigning does not affect the sign on any directed circuit in G(P).

Unfortunately, there are Pfaffian matroid pairs for which a Pfaffian signing of the fundamental graph can be resigned across edges that do not correspond to such cuts; in particular, a skew-symmetric matroid pair as defined in Examples 3 of Section 3.5 can have resignings that do not correspond to cuts in its fundamental graph. We conjecture that if a Pfaffian matroid pair is sufficiently connected and does not contain a skew-symmetric minor, then up to resigning across cuts of the fundamental graph for the pair, the Pfaffian signing is unique. Recall from Section 2.6 that the matroid pair (M, N) is *k*-connected if

 $\operatorname{rank}_M(S_1) + \operatorname{rank}_N(S_2) > r + k - 1$ 

whenever the partition  $\{S_1, S_2\}$  satisfies  $|S_1| \ge k$  and  $|S_2| \ge k$ .

**Conjecture 3.13.** If the Pfaffian matroid pair P = (M, N) is 3-connected and does not contain a skew-symmetric pair as a minor, then up to resigning across separators of M and N, the Pfaffian representation of P is unique.

## 3.7 Exact basis problem

Given a weight function on the ground set for a matroid pair, the exact basis problem is to determine if the matroid pair has a basis of a particular weight. Although there are optimality conditions and an efficient algorithm by Edmonds [10] for finding a basis of maximum or minimum weight, there are no efficient algorithms to solve the exact basis problem. Here we present an algebraic formulation of the exact basis problem which leads to an efficient solution when the matroid pair is Pfaffian. An application of the exact basis problem is the corresponding exact matching problem in an edge weighted graph. Thus, given a weighting on the edges of a Pfaffian bipartite graph, the formulation presented here can be used to determine if the graph has a perfect matching of a specific weight.

The main tool of the formulation is a linear algebra identity. Let  $A_1$  and  $A_2$  be square matrices with rows and columns indexed by the integers 1 through m. If  $X = \{x_1, x_2, \ldots, x_k\}$  and  $Y = \{y_1, y_2, \ldots, y_k\}$  are subsets of  $\{1, 2, \ldots, m\}$  then define sign(X, Y) by

$$sign(X, Y) = (-1)^{\sum_{i=1}^{k} x_i + y_i}.$$

A standard identity from linear algebra states that the determinant of  $A_1 + A_2$  can be obtained by the following summation, where  $\bar{X}$  denotes the complement of X:

$$\det(A_1 + A_2) = \sum_{X \subseteq Z} \sum_{Y \subseteq Z: |X| = |Y|} \operatorname{sign}(X, Y) \det A_1[X; Y] \det A_2[\bar{X}; \bar{Y}].$$
(3.8)

Suppose that P is a rank r Pfaffian matroid pair on the ground set S with weights w(e) defined for all  $e \in S$ . Let  $(A_M, A_N)$  be a Pfaffian representation of P and assume that the rows of  $A_M$  and  $A_N$  are indexed by the basis B. Let A be the square matrix formed from  $A_M$  and  $A_N$  as shown in (3.9), such that the rows and columns of A are indexed by  $B \cup S$  to match the row and column indices of  $A_M$  and  $A_N$ .

$$A = \begin{bmatrix} 0 & A_M \\ -A_N^\top & 0 \end{bmatrix}$$
(3.9)

Let z be an indeterminant and let Z be the diagonal matrix with the same row and column indices as A, as shown in (3.10).

$$Z = \begin{bmatrix} 0 & 0 & & \\ & z^{w(e_1)} & & \\ 0 & z^{w(e_2)} & & \\ & & \ddots & \\ & & & z^{w(n)} \end{bmatrix}$$
(3.10)

Note that the only nonsingular submatrices of Z are principal submatrices Z[X; X] with  $X \subseteq S$ . Furthermore, for all  $X \subseteq S$ ,

$$\det Z[X;X] = \sum_{e \in X} z^{w(e)} = z^{w(X)}.$$

Since sign(X, X) = 1 for all  $X \subseteq S$ , identity (3.8) implies that

$$\det(A+Z) = \sum_{X \subseteq S} \det A[B \cup X; B \cup X] \det Z[S \setminus X; S \setminus X]$$
$$= \sum_{X \subseteq Z} \det A_M[X] \det A_N[X] z^{w(S)-w(X)}$$

The assumption that  $(A_M, A_N)$  is a Pfaffian representation of P therefore implies that the coefficient of  $z^{w(S)-k}$  in det(A + Z) is exactly the number of bases of P with weight k. It follows that the exact basis problem can be solved for a pair of Pfaffian matroids. In particular, when M is a regular matroid and w is a weight function on the ground set of M, then for all weights k this formulation determines if M has a basis of weight k. We note that the weights given by w must be unitary for this formulation to lead to an efficient algorithm, otherwise det(A + Z) can have exponentially many terms. For non unitary weights, the problem remains hard.

# Chapter 4 Binary Spaces

A major obstacle to characterizing Pfaffian matroid pairs is the existence of "inequivalent" Pfaffian signings: given two Pfaffian representations of the same matroid pair, we do not have a simple description of how these two signings differ. (See Section 3.6 and Examples 2 and 3 in Section 3.5.)

Given a matroid P on the ground set S, a set  $X \subseteq S$  is called a *constant-parity* intersecting set if either

- (i)  $|B \cap X|$  is even for all bases B of P, or
- (ii)  $|B \cap X|$  is odd for all bases B of P.

It follows that resigning a Pfaffian representation of P on  $X \subseteq S$  gives another Pfaffian representation if and only if X is a constant-parity intersecting set. Thus the problem of describing the relationship between multiple Pfaffian representations of the same matroid pair is equivalent to the problem of describing the constant-parity intersecting sets of the matroid pair.

In this chapter we approach the problem of describing the constant-parity sets for a matroid pair P through a characterization of the *binary space* generated by the bases of P; if A is a set of  $\{0, 1\}$  vectors then the binary space generated by A, denoted Bin(A), is all linear combinations over GF(2) of the vectors in A.

Let M be a matroid on the n element ground set S and let  $Q^M$  and  $R^M$  be the subsets of  $GF(2)^n$  given by

$$Q^{M} = \{ x \in GF(2)^{n} : x(S_{i}) = 0 \text{ for all } 1 \le i \le c \} \text{ and}$$
$$R^{M} = \{ x \in GF(2)^{n} : x(S_{i}) = \operatorname{rank}(S_{i}) \text{ for all } 1 \le i \le c \}.$$

If P is the connected matroid pair (M, N), then we show in Theorem 4.8 that

$$\operatorname{Bin}(\mathcal{B}(P)) \subseteq (Q^M \cap Q^N) \cup (R^M \cap R^N).$$
(4.1)

Our main result is Theorem 4.14, where we show that by excluding the twined  $K_4$  pair (Example 2 of Section 3.5) as a matroid minor, the inclusion in Equation (4.1) above is tight. This proof of this main theorem considers a binary space defined on the vertices of a directed graph. In Section 4.2 we determine the dimension of this binary space using a variant of ear decompositions, and in Section 4.3 this dimension is used to characterize  $Bin(\mathcal{B}(P))$  for a matroid pair P in terms of the fundamental graph for P.

### 4.1 The binary space of a matroid

If M is a matroid on the ground set S and S' is a component of M, then for all bases B of M,  $|B \cap S'| = \operatorname{rank}_M(S')$ , and thus the parity of  $|B \cap S'|$  is equal to the parity of the rank of S'. If  $x \in \operatorname{Bin}(\mathcal{B}(M))$  is the sum of k basis vectors, then it follows that  $|x(S')| = k \cdot \operatorname{rank}_M(S')$  over GF(2). This simple necessary condition completely characterizes the vectors in  $\operatorname{Bin}(\mathcal{B}(M))$ , as the next theorem shows.

**Theorem 4.1 (Rieder).** Let M be a matroid with c components on the ground set S and for  $1 \leq i \leq c$ , let  $S_i \subseteq S$  be the ground set for the  $i^{th}$  component of M. If

$$Q^{M} = \{ x \in GF(2)^{S} : x(S_{i}) = 0 \text{ for all } 1 \le i \le c \} \text{ and}$$
$$R^{M} = \{ x \in GF(2)^{S} : x(S_{i}) = \operatorname{rank}(S_{i}) \text{ for all } 1 \le i \le c \},$$

then  $\operatorname{Bin}(\mathcal{B}(M)) = Q^M \cup R^M$ .

*Proof.* If  $y \in Bin(\mathcal{B}(M))$  then there exist  $b_1, b_2, \ldots, b_k \in \mathcal{B}(M)$  such that  $y = \sum_{j=1}^k b_j$ , and therefore for  $1 \leq i \leq c$ ,

$$y(S_i) = \sum_{j=1}^k b_j(S_i) = k \operatorname{rank}(S_i) = \begin{cases} 0, & \text{if } k \text{ is even;} \\ \operatorname{rank}(S_i), & \text{if } k \text{ is odd.} \end{cases}$$

Thus if  $y \in Bin(\mathcal{B}(M))$ , then  $y \in Q^M \cup R^M$ .

Let  $1 \leq i \leq c$  be such that  $|S_i| \geq 2$  and let  $u, v \in S_i$ . Since M restricted to  $S_i$  is connected, there exists a cycle C in M with  $C \subseteq S_i$  and  $u, v \in C$ . The independent set  $C \setminus u$  can be extended to a basis  $B_v$  of M where  $u \notin B_v$ . The dependent set  $B_v \cup u$  then contains a unique cycle, namely C, and therefore  $B_u = (B_v \cup u) \setminus v$  is also a basis of M. Let  $e_j$  denote the  $j^{th}$  standard basis vector. The sum over GF(2) of the characteristic vector for  $B_v$  with the characteristic vector for  $B_u$  is then  $e_v + e_u$ , and since i, u and vwere arbitrary, it follows that  $e_u + e_v \in Bin(\mathcal{B}(M))$  whenever u and v are elements in the same component of M. Thus if x satisfies  $x(S_i) = 0$  for all  $1 \leq i \leq c$ , then x can be expressed as a linear combination of basis vectors, and therefore

$$Q^M \subseteq \operatorname{Bin}(\mathcal{B}(M)). \tag{4.2}$$

Let  $x \in \mathbb{R}^M$  and  $y \in \mathcal{B}(M)$ , and let z = x + y. Then  $z(S_i) = 0$  for  $1 \leq i \leq c$ , and thus by (4.2),  $z \in Bin(\mathcal{B}(M))$ . It follows that  $z + y = x \in Bin(\mathcal{B}(M))$ , and therefore  $\mathbb{R}^M \subseteq Bin(\mathcal{B}(M))$ . The theorem follows.

A description of the lattice generated by  $\mathcal{B}(M)$  contains a description of the binary space generated by  $\mathcal{B}(M)$  and thus Theorem 4.1 is a corollary of Rieder's [33] characterization of Lat( $\mathcal{B}(M)$ ) (Theorem 5.2). If dim(Bin( $\mathcal{B}(M)$ )) denotes the dimension of the binary space of Bin( $\mathcal{B}(M)$ ), then Theorem 4.1 implies that dim(Bin( $\mathcal{B}(M)$ )) is determined by the number of components of M and the parity of the rank of these components.

**Corollary 4.2.** If M is a matroid on n elements and M has c connected components, then

$$\dim(\operatorname{Bin}(\mathcal{B}(M))) = \begin{cases} n-c, & \text{if all components of } M \text{ have even rank;} \\ n-c+1, & \text{if a component of } M \text{ has odd rank.} \end{cases}$$

Proof. Let S be the ground set for M and for  $1 \leq i \leq c$  let  $S_i$  be the subset of S corresponding to the  $i^{th}$  component of M. Let A be the matrix over GF(2) such that the  $i^{th}$  row of A is the characteristic vector for  $S_i$ . Since A has n columns and c independent rows, the nullspace of A has dimension n-c. Since  $Q^M = \{x : Ax = 0\}$ , the dimension of  $Q^M$  is n-c. If all components of M have even rank then by Theorem 4.1,  $Bin(\mathcal{B}(M)) = Q^M$  and thus  $Bin(\mathcal{B}(M))$  has dimension n-c.

Suppose instead that a component of M has odd rank. Let  $r \in \{0,1\}^c$  be such that  $r_i$  is equivalent to  $\operatorname{rank}_M(S_i)$  over GF(2), and let  $y \in \mathcal{B}(M)$ . Then  $Ay = r \neq 0$ , and therefore y is not in the nullspace of A. It follows that the dimension of  $\operatorname{Bin}(\mathcal{B}(M))$  is at least n - c + 1. If  $\{z_1, z_2, \ldots, z_{n-c}\}$  is a basis for the nullspace of A and  $x \in \mathcal{B}(M)$ , then A(x + y) = r + r = 0 and thus there exist  $\alpha_i \in \{0, 1\}$  such that  $x + y = \sum_{i=1}^{n-c} \alpha_i z_i$ . Therefore x is in the span of  $\{y, z_1, \ldots, z_{n-c}\}$  whenever  $x \in \mathcal{B}(M)$ , and it follows that  $\operatorname{Bin}(\mathcal{B}(M))$  has dimension n - c + 1 whenever a component of M has odd rank.

For a matroid pair P = (M, N), we would like a description of  $Bin(\mathcal{B}(P))$  in terms of the intersection of  $Bin(\mathcal{B}(M))$  and  $Bin(\mathcal{B}(N))$ . Our approach requires some theorems about cycle spaces, which we present next.

# 4.2 Cycle spaces in directed graphs

Given a connected matroid pair P = (M, N) with sets  $Q^M, Q^N, R^M$ , and  $R^N$  defined as in Theorem 4.1, the binary space generated by  $\mathcal{B}(P)$  is characterized by the directed circuits in the fundamental graph for P, as the following theorem shows: **Theorem** If P = (M, N) is a connected matroid pair, then

$$\operatorname{Bin}(\mathcal{B}(P)) = (Q^M \cap Q^N) \cup (R^M \cap R^N)$$

if and only if  $V(C) \in Bin(\mathcal{B}(P))$  for every directed cycle C in G(P).

We prove this theorem in Section 4.3, and our proof uses the dimension of the vector space generated by the vertices of the directed circuits in the fundamental graph for the matroid pair. In this section we determine the dimension of that vector space for any strongly connected bipartite graph.

Let G = (V, E) be a directed graph. A *circulation* in G is a function on the edges of G such that the total flow out of any vertex is equal to the total flow into the vertex. That is, a circulation in G over the field  $\mathbb{F}$  is a vector  $f \in \mathbb{F}^E$  such that

$$f(\delta(v)) = f(\delta(V \setminus v)) \quad \forall v \in V.$$

The set of circulations of a graph G is called the *cycle space* of G and is denoted  $\mathcal{C}(G)$ . The cycle space is a well studied vector space and its dimension is well known. (See Bondy and Murty [2].)

**Theorem 4.3.** If G = (V, E) is a strongly connected directed graph and  $\mathbb{F}$  is a field, then the dimension of  $\mathcal{C}(G)$  over  $\mathbb{F}$  is |E| - |V| + 1.

Note that if C is a circuit in G then there is an associated circulation f for C where

 $f(e) = \begin{cases} 1, & \text{if } e \text{ is a forward edge in } C \text{ with respect to an orientation of } C; \\ -1, & \text{if } e \text{ is a backward edge in } C \text{ with respect to the same orientation;} \\ 0, & \text{otherwise.} \end{cases}$ 

Every circulation in  $\mathcal{C}(G)$  can be expressed as a weighted sum of circulations for circuits (see Bondy and Murty [2]).

If G is a directed graph then for each  $f \in \mathcal{C}(G)$  we define a function  $f_V$  on the vertices of G such that  $f_V(v)$  is the total amount of flow out of, and therefore the total amount of flow into, vertex v:

$$f_V(v) = \sum_{vu \in \delta(v)} f(vu) \qquad \forall v \in V(G).$$

The vertex cycle space of G, denoted  $\mathcal{C}_V(G)$ , is the set of  $f_V$  over all  $f \in \mathcal{C}(G)$ . In our characterization of the binary space for a connected matroid pair P we require the dimension of the vertex cycle space over GF(2) for G(P). In this section we therefore determine the dimension of  $\mathcal{C}_V(G)$  over GF(2) when G is a strongly connected and bipartite directed graph. When G has vertex bipartition  $\{X, Y\}$  we partition the edges of G into those directed out of X and those directed into X, and we consider the two subgraphs induced by these edges. We first determine the dimension of the subspace of  $C_V(G)$  containing those vectors which are zero both on every component of  $G[\delta(X)]$  and on every component of  $G[\delta(Y)]$ .

**Lemma 4.4.** Let G be a strongly connected bipartite directed graph with vertex bipartition  $\{X, Y\}$  and let  $c_X$  and  $c_Y$  be the number of components of  $G[\delta(X)]$  and  $G[\delta(Y)]$ respectively. Let  $\mathcal{W}$  be the set of  $W \subseteq X \cup Y$  such that G[W] is a component of either  $G[\delta(X)]$  or  $G[\delta(Y)]$ . If  $\mathbb{F}$  is a field and

$$R = \{ x \in \mathbb{F}^V : x(W) = 0 \quad for \ all \quad W \in \mathcal{W} \},\$$

then R has dimension  $|X \cup Y| - c_X - c_Y + 1$ .

Proof. Let A be the matrix over  $\mathbb{F}$  whose rows are the characteristic vectors for the subsets in  $\mathcal{W}$ . Then  $R = \{x : Ax = 0\}$ , and the dimension of R is equal to the dimension of the nullspace of A. Since G is strongly connected, every vertex of G is incident to an arc in  $\delta(X)$ . The components of  $G[\delta(X)]$  therefore partition the vertices of G, and summing the rows of A over all components of  $G[\delta(X)]$  gives the all ones vector. Similarly, the sum of the rows of A indexed by the components of  $G[\delta(Y)]$  is the all ones vector. Thus the rows of A are dependent and the rank of A is at most  $|\mathcal{W}| - 1$ . If a strict subset of the rows of A are dependent, then there is a set of the components of  $G[\delta(Y)]$  whose union covers the same proper subset of V as a set of components of  $G[\delta(Y)]$ . This contradicts that G is connected, and therefore the rank of A is equal to  $|\mathcal{W}| - 1 = c_X + c_Y - 1$ . Since A has |V| columns, the nullspace of A has dimension  $|V| - c_X - c_Y + 1$ , and the lemma follows.

Let G be a strongly connected directed graph and let H be a subgraph of G. An *ear* of H is a directed path Q in G with both ends of Q in V(H) and no interior vertices of Q in V(H). The ends of an ear need not be distinct. An *ear decomposition* of G is a sequence of strongly connected subgraphs  $G_0, G_1, \ldots, G_k$  of G such that  $G_0$  is a directed circuit,  $G_k = G$ , and for  $1 \leq i \leq k$ , there exists an ear  $Q_{i-1}$  of  $G_{i-1}$  such that  $G_i = G_{i-1} \cup Q_{i-1}$ . Hence

$$|E(G_i)| - |V(G_i)| = |E(G_{i-1})| - |V(G_{i-1})| + 1$$

for  $1 \leq i \leq k$ , and therefore

$$k = |E(G)| - |V(G)|.$$
(4.3)

It follows that all ear decompositions of G contain the same number of ears. Ear decompositions were first used by Hetyei [17] and are a common tool in graph theory.

For directed bipartite graphs we describe a variant of an ear decomposition using *acyclic ears*: if H is a subgraph of the strongly connected bipartite directed graph G, and G has vertex bipartition  $\{X, Y\}$ , then we call an ear Q of H acyclic if the subgraph of G induced by  $\delta(X) \cap E(H \cup Q)$  is acyclic and the subgraph of G induced by  $\delta(Y) \cap E(H \cup Q)$  is acyclic and the subgraph of G induced by  $\delta(Y) \cap E(H \cup Q)$  is acyclic and the subgraph of G induced by  $\delta(Y) \cap E(H \cup Q)$  is also acyclic. If H is the subgraph induced by the bold edges in Figure 4.1(a), then the directed path shown with dashed lines in Figure 4.1(b) is an acyclic ear of H.



Figure 4.1: An acyclic ear of a subgraph

Let G be a strongly connected directed bipartite graph and let  $d \ge 0$  be such that there exist subgraphs  $H_0, H_1, \ldots, H_d$  of G where  $H_0$  is a directed circuit in G and for  $1 \le i \le d-1, H_{i+1}$  is formed by the addition of an acyclic ear to  $H_i$ . Note that if a subgraph H has an acyclic ear then H restricted to  $G[\delta(X)]$  is a forest and H restricted to  $G[\delta(Y)]$  is a forest. However, when H restricted to  $G[\delta(X)]$  is a forest and H restricted to  $G[\delta(Y)]$  is a forest, it need not be that H has an acyclic ear in G. If  $H_d$  does not have an acyclic ear, then we call the sequence  $H_0, \ldots, H_d$  an acyclic ear decomposition of G.

Ear decompositions and acyclic ear decompositions share some similar properties. For example, just as all ear decompositions of a strongly connected graph have the same number of ears, all acyclic ear decompositions of a strongly connected directed bipartite graph have the same number of acyclic ears, as shown in the next lemma.

**Lemma 4.5.** Let G be a strongly connected bipartite directed graph with vertex bipartition  $\{X, Y\}$  and let  $c_X$  and  $c_Y$  be the number of components of  $G[\delta(X)]$  and  $G[\delta(Y)]$ respectively. If  $H_0, \ldots, H_d$  is an acyclic ear decomposition of G, then

$$d = |X \cup Y| - c_X - c_Y.$$

Proof. If  $H_d$  is not a spanning subgraph of G then there exists an edge uv of G such that  $u \in V(H_d)$  and  $v \notin V(H_d)$ . For any shortest path Q from v to  $V(H_d)$ , the ear  $Q_d = \{uv\} \cup Q$  is an acyclic ear for  $H_d$ . This contradicts the definition of an acyclic ear decomposition and thus  $H_d$  is a spanning subgraph of G.

Suppose the subgraph of  $H_d$  induced by  $E(H_d) \cap \delta(X)$  is not a spanning subgraph of  $G[\delta(X)]$ , and let  $v \in V(G)$  be such that v is not incident to an edge in  $E(H_d) \cap \delta(X)$ . Since G is strongly connected, v is incident to an edge  $e \in \delta(X)$ . The subgraph of G induced by  $\delta(X) \cap E(H_d \cup e)$  is therefore acyclic and by definition, the ear  $Q_d = e$  is an acyclic ear of  $H_d$ . Thus  $H_d$  must be a spanning forest of  $G[\delta(X)]$ , and it follows that

$$|E(H_d) \cap \delta(X)| = |V(H_d)| - c_X = |X \cup Y| - c_X.$$
(4.4)

Similarly, the subgraph of  $H_d$  induced by  $\delta(Y)$  is a spanning forest of  $G[\delta(Y)]$ , and therefore

$$E(H_d) \cap \delta(Y)| = |X \cup Y| - c_Y.$$

$$(4.5)$$

It follows from Equations (4.4) and (4.5) that

$$|E(H_d)| = 2|X \cup Y| - c_X - c_Y, \tag{4.6}$$

and since  $H_0, H_1, \ldots, H_d$  is an ear decomposition of  $H_d$ ,

$$d = |E(H_d)| - |X \cup Y|.$$
(4.7)

Equations (4.6) and (4.7) imply that  $d = |X \cup Y| - c_X - c_Y$ .

Given an ear decomposition  $G_0, G_1, \ldots, G_k$  of G and paths  $Q_{i-1}$  with  $G_i = G_{i-1} \cup Q_{i-1}$ for  $1 \leq i < k$ , let  $C_0 = G_0$  and for  $1 \leq i \leq k$  let  $C_i$  be a directed circuit in  $G_i$  with  $Q_{i-1} \subseteq C_i$ . For all  $0 \leq i \leq k$ , there exists  $e \in E(C_i)$  such that  $e \notin E(C_j)$  for all  $0 \leq j < i$ , and therefore the characteristic vectors for  $E(C_0), \ldots, E(C_k)$  are independent. Since k = |E(G)| - |V(G)|, Theorem 4.3 implies that  $\{C_0, \ldots, C_k\}$  is a basis for  $\mathcal{C}(G)$ . Such a basis is called a cycle basis of  $\mathcal{C}(G)$  with respect to the ear decomposition  $G_0, G_1, \ldots, G_k$ . An analogous set of independent vectors exists for the vertex cycle space  $\mathcal{C}_V(G)$  using acyclic ear decompositions, as we show next. Note that the cycle basis for  $\mathcal{C}(G)$  is defined with respect to the *edges* of G where as the vertex cycle space is defined with respect to the *vertices* of G.

**Lemma 4.6.** Let G be a strongly connected bipartite directed graph with acyclic ear decomposition  $H_0, \ldots, H_d$  and let  $C_0, \ldots, C_d$  be a cycle basis of  $C(H_d)$  with respect to  $H_0, \ldots, H_d$ . The characteristic vectors for  $V(C_0), \ldots, V(C_d)$  are independent over GF(2).

*Proof.* For  $0 \leq i \leq d$ , let  $c_i$  be the characteristic vector for  $V(C_i)$  and let  $f_{C_i} \in \mathcal{C}$  be the associated circulation for  $C_i$ . Assume that  $\alpha_i \in \{0, 1\}$  are such that  $\sum_{i=1}^d \alpha_i c_i = 0$  over GF(2), and let  $f \in \mathcal{C}(G)$  be given by

$$f = \sum_{i=0}^{d} \alpha_i f_{C_i}.$$

Suppose  $\alpha_i \neq 0$  for some  $0 \leq i \leq d$ . Since  $f_{C_1}, \ldots, f_{C_d}$  are independent over GF(2),  $f \neq 0$  and there exists an edge h of G such that  $f(h) \neq 0$ . Without loss of generality, assume that  $h \in \delta(X)$  and consider the subgraph H of  $H_d$  where  $e \in E(H)$  if and only

if  $f(e) \neq 0$ . Thus  $h \in E(H)$ , and by the assumption that  $\sum_{i=1}^{d} \alpha_i c_i = 0$ , the sum of the weights of all edges out of each vertex is zero, as is the sum of the weights of all edges into each vertex. So  $H[\delta(X)]$  is a nonempty subgraph with each vertex of degree at least two, and therefore  $H[\delta(X)]$  contains a circuit. But H is a subgraph of  $H_d$  and by construction  $H_d[\delta(X)]$  is acyclic. Thus  $\alpha_i = 0$  for all i, and  $\{c_1, \ldots, c_d\}$  are independent over GF(2).

We can now prove the main result of this section.

**Theorem 4.7.** If G is a strongly connected bipartite directed graph with vertex bipartition  $\{X, Y\}$ , and  $c_X$  and  $c_Y$  are the number of connected components in  $G[\delta(X)]$  and  $G[\delta(Y)]$  respectively, then the dimension of  $C_V(G)$  over GF(2) is

$$|X \cup Y| - c_X - c_Y + 1.$$

*Proof.* Let  $C \subseteq E(G)$  be a directed circuit in G and let x be the characteristic vector over GF(2) for V(C). Since  $C \cap \delta(X)$  is a matching in  $G[\delta(X)]$ , V(C) intersects each component in  $G[\delta(X)]$  with even parity, and therefore x(W) = 0 whenever  $W \subseteq V(G)$  is a component of  $G[\delta(X)]$ . Similarly,  $C \cap \delta(Y)$  is a matching in  $G[\delta(Y)]$  and if  $W \subseteq V(G)$ is a component of  $G[\delta(Y)]$  then x(W) = 0.

Since each function in  $C_V(G)$  can be expressed as the sum of directed circuits, it follows that y(W) = 0 for all  $y \in C_V(G)$ , and by Lemma 4.4, the dimension of  $C_V(G)$  over GF(2)is at most  $|V| + 1 - c_X - c_Y$ . By Lemma 4.6, there exist  $|V| + 1 - c_X - c_Y$  directed circuits that are independent with respect to vertices, and the theorem follows.

## 4.3 The binary space for a pair of matroids

Let P = (M, N) be a connected matroid pair. Since  $\mathcal{B}(P) = \mathcal{B}(M) \cap \mathcal{B}(N)$ ,

$$\operatorname{Bin}(\mathcal{B}(P)) \subseteq \operatorname{Bin}(\mathcal{B}(M)) \cap \operatorname{Bin}(\mathcal{B}(N)).$$
(4.8)

We consider when this containment is strict and show that this is related to the vertex cycle space (Section 4.2) for the fundamental graph of P.

Let M and N be matroids on the ground set S and assume that M has  $c_M$  connected components and that N has  $c_N$  connected components. For  $1 \leq i \leq c_M$  and  $1 \leq j \leq c_N$ , let  $S_i^M \subseteq S$  and  $S_j^N \subseteq S$  be the ground sets corresponding to the  $i^{th}$  and  $j^{th}$  components of M and N respectively. Let  $Q^M$  and  $R^M$  be subspaces defined over GF(2) with

$$Q^{M} = \{x \in \{0, 1\}^{S} : x(S_{i}^{M}) = 0 \text{ for all } 1 \le i \le c_{M}\} \text{ and}$$
$$R^{M} = \{x \in \{0, 1\}^{S} : x(S_{i}^{M}) = \operatorname{rank}_{M}(S_{i}^{M}) \text{ for all } 1 \le i \le c_{M}\}.$$

If we define  $Q^N$  and  $R^N$  similarly, then by Theorem 4.1 and Equation (4.8),

$$\operatorname{Bin}(\mathcal{B}(M,N)) \subseteq (Q^M \cap Q^N) \cup (Q^M \cap R^N) \cup (R^M \cap Q^N) \cup (R^M \cap R^N).$$
(4.9)

This containment can be tightened can be P is connected, as the next theorem shows.

**Theorem 4.8.** If P is the connected matroid pair (M, N), then

$$\operatorname{Bin}(\mathcal{B}(P)) \subseteq (Q^M \cap Q^N) \cup (R^M \cap R^N).$$
(4.10)

*Proof.* If every component of M is even, then  $Q^M = R^M$  and therefore Equations (4.9) and (4.10) are equivalent. Similarly, if  $Q^N = R^N$  then (4.9) and (4.10) are equivalent. Thus we may assume that there exists a component  $X_M \subseteq S$  of M such that  $\operatorname{rank}_M(X_M)$  is odd and a component  $X_N \subseteq S$  of N such that  $\operatorname{rank}_M(X_N)$  is odd. If  $y \in \operatorname{Bin}(\mathcal{B}(P))$  then there exist  $b_1, \ldots, b_d \in \mathcal{B}(P)$  such that  $y = \sum_{i=1}^d b_i$  and therefore

$$y(X_M) = \sum_{i=1}^d b_i(X_M) = d = \sum_{i=1}^d b_i(X_N) = y(X_N).$$
(4.11)

If  $x \in \mathbb{R}^M$  then  $x(X_M) = 1$  and if  $x \in \mathbb{Q}^N$  then  $x(X_N) = 0$ , and therefore Equation 4.11 implies that  $(\mathbb{R}^M \cap \mathbb{Q}^N) \cap \operatorname{Bin}(\mathcal{B}(P)) = \emptyset$ . Similarly,  $(\mathbb{Q}^M \cap \mathbb{R}^N) \cap \operatorname{Bin}(\mathcal{B}(P)) = \emptyset$ , and (4.9) simplifies to

$$\operatorname{Bin}(\mathcal{B}(P)) \subseteq (Q^M \cap Q^N) \cup (R^M \cap R^N).$$

Theorem 4.8 suggests characterizing the matroid pairs P = (M, N) for which  $Bin(\mathcal{B}(P))$ is equal to  $(Q^M \cap Q^N) \cup (R^M \cap R^N)$ , rather than characterizing the matroid pairs for which  $Bin(\mathcal{B}(P)) = Bin(\mathcal{B}(M)) \cap Bin(\mathcal{B}(N))$ .

**Corollary 4.9.** If P = (M, N) is a connected matroid pair with even rank and M has a component of odd rank, then

$$\operatorname{Bin}(\mathcal{B}(P)) \neq \operatorname{Bin}(\mathcal{B}(M)) \cap \operatorname{Bin}(\mathcal{B}(N)).$$

*Proof.* If M has a component of odd rank then the proof of Theorem 4.8 shows that  $R^M \cap Q^N$  and  $\operatorname{Bin}(\mathcal{B}(P))$  have no common elements. Since  $R^M \cap Q^N$  is contained in  $\operatorname{Bin}(\mathcal{B}(M)) \cap \operatorname{Bin}(\mathcal{B}(N))$ , the corollary follows if  $R^M \cap Q^N$  is nonempty.

If  $R^M \cap Q^N$  is empty, then there does not exists a vector x over GF(2) which satisfies both  $x(S_i^M) = \operatorname{rank}_M(S_i^M)$  for all  $1 \leq i \leq c_M$  and  $x(S_i^N) = 0$  for all  $1 \leq i \leq c_N$ . By Observation 3.9, there exists  $I_M \subseteq \{1, \ldots, c_M\}$  and  $I_N \subseteq \{1, \ldots, c_N\}$  such that

$$\bigcup_{i \in I_M} S_i^M = \bigcup_{i \in I_N} S_i^N \tag{4.12}$$

and

$$\sum_{i \in I_M} \operatorname{rank}_M(S_i^M) = 1 \pmod{2}.$$
(4.13)

Since P is connected, (4.12) is satisfied only if

$$\bigcup_{i \in I_M} S_i = S$$

This implies that  $\sum_{i \in I_M} \operatorname{rank}_M(S_i^M) = x(S)$  which contradicts (4.13) since P has even rank by assumption. Thus  $R^M \cap Q^N$  is nonempty and the corollary follows.

Before characterizing those matroid pairs P = (M, N) for which  $Bin(\mathcal{B}(P))$  is equal to  $(Q^M \cap Q^N) \cup (R^M \cap R^N)$ , we first explain why such pairs, if Pfaffian, have essentially unique Pfaffian signings. Recall from the introduction to this chapter that a constantparity intersecting set of the matroid pair P is a subset S of the ground set for P such that either  $|B \cap X|$  is even for all bases B of P, or  $|B \cap X|$  is odd for all bases B of P.

**Theorem 4.10.** If P = (M, N) is a strongly connected matroid pair on the ground set S and

$$\operatorname{Bin}(\mathcal{B}(P)) = (Q^M \cap Q^N) \cup (R^M \cap R^N),$$

then the set  $X \subseteq S$  is a constant-parity intersecting set for P if and only if there exists a separator  $X_M$  of M and a separator  $X_N$  of N such that  $X = X_M \Delta X_N$ .

*Proof.* Let P = (M, N) be a strongly connected matroid pair on the ground set S. If  $X_M$  and  $X_N$  are separators of M and N respectively and B is a basis of P, then over GF(2),

 $|B \cap (X_M \Delta X_N)| = |B \cap X_M| + |B \cap X_N| = \operatorname{rank}_M(X_M) + \operatorname{rank}_N(X_N).$ 

Since  $\operatorname{rank}_M(X_M)$ + $\operatorname{rank}_N(X_N)$  has fixed parity,  $X_M \Delta X_N$  is a constant-parity intersecting set for P.

Suppose instead that  $X \subseteq S$  is a constant-parity intersecting set for P and let A be the matrix over GF(2) whose rows are the characteristic vectors for the components of M and N. Note that the vector space  $Q^M \cap Q^N$  is then equal to the nullspace of A.

First consider the case that  $|B \cap X|$  is even for all bases B of P. Then x(X) = 0 for all x in the nullspace of A, and thus X is the symmetric difference of components of Mand N. It follows that  $X = X_M \Delta X_N$  for some separator  $X_M$  of M and separator  $X_N$  of N.

Next consider the case that  $|B \cap X|$  is odd for all bases B of P. If some component of M or N has odd rank, then  $Q^M \cap Q^N$  is the set of all binary combinations of an even number of bases from  $\mathcal{B}(P)$ , and  $R^M \cap R^N$  is the set of all binary combinations of an odd number of bases from  $\mathcal{B}(P)$ . Since x(X) is odd if  $x \in R^M \cap R^N$ , if  $x \in Q^M \cap Q^N$  then x(X) is even and thus x(X) = 0 for all x in the nullspace of A. It again follows that  $X = X_M \Delta X_N$  for some separator  $X_M$  of M and separator  $X_N$  of N.

It remains to consider the case that  $|B \cap X|$  is odd for all bases B of P and every component of M and N has even rank. In this case,  $R^M \cap R^N = Q^M \cap Q^N$  and the binary space generated by  $\mathcal{B}(P)$  is equal to the nullspace of A. Since some vectors in the nullspace of A correspond to the sum of an even number of bases and some correspond to an odd number of bases, it is no longer the case that x(X) = 0 for all x in the nullspace of A. We show in fact that such a case can not occur.

Since every component of M has even rank, P has even rank and thus  $X \neq S$ . Let  $u \in S$  be such that  $u \notin X$  and let M' be obtained from M by replacing the element u by the two elements  $u_1$  and  $u_2$  such that

- (i)  $u_1$  and  $u_2$  are in series in M',
- (ii)  $u_2$  is a loop of  $M'/u_1$ , and

(iii) 
$$M/u_2 = M$$
.

Let  $S' = S\Delta\{u, u_1, u_2\}$  and r' = r + 1, where r is the rank of M. Then the rank of M' is r', and if  $T \subseteq S'$  then

$$\operatorname{rank}_{M'}(T) = \begin{cases} \operatorname{rank}_{M}(T), & \text{if } |T \cap \{u_{1}, u_{2}\}| = 0; \\ \operatorname{rank}_{M}(T \setminus \{u_{1}, u_{2}\}) + 1, & \text{if } |T \cap \{u_{1}, u_{2}\}| = 1; \\ \operatorname{rank}_{M}(T \Delta \{u, u_{1}, u_{2}\}) + 1, & \text{if } |T \cap \{u_{1}, u_{2}\}| = 2. \end{cases}$$

$$(4.14)$$

If we define N' similarly and let P' be the matroid pair (M', N'), then since P is a connected matroid pair it follows from Equation (4.14) that P' is a connected matroid pair. We show that  $|B' \cap X|$  is odd for all bases B' of P', and that this leads to a contradiction.

If B' is a basis of P' then since  $u_1$  and  $u_2$  are in series,  $|B' \cap \{u_1, u_2\}| \ge 1$ . Suppose first that  $u_2 \in B'$  and  $u_1 \in B'$ . Then  $B = B' \Delta \{u_1, u_2, u\}$  is a basis of P and since  $|B \cap X|$ is odd and  $\{u_1, u_2, u\} \cap X$  is empty,  $|B' \cap X|$  is odd. Next, suppose that  $u_2 \in B'$  and  $u_1 \notin B'$ . Then  $B = B' \Delta \{u_2\}$  is a basis of P which implies that  $|B \cap X|$  is odd and again  $|B' \cap X|$  is odd. Finally, suppose that  $u_2 \notin B'$ . Then  $u_1 \in B'$  and  $B = B' \Delta \{u, u_1\}$  is a basis of P. Since  $|B \cap X|$  is odd and  $\{u, u_1\} \cap X$  is empty, it follows that  $|B' \cap X|$  is odd.

Thus  $|B' \cap X|$  is odd for all bases B' of P, and since P' is connected it follows that  $X = X_{M'}\Delta X_{N'}$  for some separator  $X_{M'}$  of M' and  $X_{N'}$  of N'. Exactly one of  $X_{M'}$  and  $X_{N'}$  must have odd rank for  $|B' \cap X|$  to be odd, and since only the component containing  $u_1$  and  $u_2$  has odd rank in either M' or N', this implies that  $\{u_1 \cup u_2\}$  is in exactly one of  $X_{M'}$  and  $X_{N'}$ . This in turn implies that  $\{u_1, u_2\} \subset X$  and since  $\{u_1, u_2\} \cap S = \emptyset$ , this contradicts the assumption that  $X \subseteq S$ .

We now give a characterization in terms of fundamental graphs for when the binary space generated by  $(\mathcal{B}(M, N))$  is equal to  $(Q^M \cap Q^N) \cup (R^M \cap R^N)$ . We require one lemma concerning the dimension of  $(Q^M \cap Q^N) \cup (R^M \cap R^N)$ , the proof of which follows from the results of Section 4.2.

**Lemma 4.11.** Let P = (M, N) be a connected matroid pair on the ground set S, and assume M and N have  $c_M$  and  $c_N$  components respectively. If all components of M and N have even rank then  $(Q^M \cap Q^N) \cup (R^M \cap R^N)$  has dimension  $|S| - (c_M + c_N) + 1$ , and if a component of M or N has odd rank then  $(Q^M \cap Q^N) \cup (R^M \cap R^N)$  has dimension  $|S| - (c_M + c_N) + 2$ .

*Proof.* Let A be the matrix over GF(2) whose rows are the characteristic vectors for the components of M and N. The sum of the rows of A is zero, and since P is connected, no proper subset of the rows of A sums to zero. Thus A has rank  $c_M + c_N - 1$  and therefore the nullspace of A has dimension  $|S| - c_M - c_N + 1$ . Since  $Q^M \cap Q^N$  is equal to the nullspace of A, the dimension of  $Q^M \cap Q^N$  is  $|S| - c_M - c_N + 1$ . If every component of M and N has even rank, then  $R^M = Q^M$  and  $R^N = Q^N$  and the lemma follows.

Suppose a component of M or N has odd rank, and let y be the characteristic vector for a common basis of M and N. Then  $Ay \neq 0$  and thus the dimension of  $(Q^M \cap Q^N) \cup$  $(R^M \cap R^N)$  is at least  $|S| - c_M - c_N + 2$ . Let r be such that Ay = r and let  $x \in R^M \cap R^N$ . Then Ax = r = Ay, and thus A(x + y) = 0. Hence x + y is in the nullspace of A for all  $x \in R^M \cap R^N$ , and it follows that  $(Q^M \cap Q^N) \cup (R^M \cap R^N)$  has dimension  $|S| - c_M - c_N + 2$ whenever M or N has a component of odd rank.

We can now prove the main theorem of this section.

**Theorem 4.12.** If P = (M, N) is a connected matroid pair, then

$$\operatorname{Bin}(\mathcal{B}(P)) = (Q^M \cap Q^N) \cup (R^M \cap R^N)$$
(4.15)

if and only if  $V(C) \in Bin(\mathcal{B}(P))$  for every directed cycle C in G(P).

Proof. Let C be a directed circuit in G(P). Since E(C) is a matching in G(M), V(C) intersects each component in G(M) with even parity and thus  $V(C) \in Q^M$ . Similarly,  $V(C) \in Q^N$  and therefore  $V(C) \in Q^M \cap Q^N$ . Thus if Equation (4.15) holds, then  $V(C) \in Bin(\mathcal{B}(P))$  whenever C is a directed circuit in G(P).

Conversely, suppose  $V(C) \in Bin(\mathcal{B}(P))$  for all directed cycles C in G(P), and let G = G(P). Then  $G[\delta(B)] = G(M)$  and  $G[\delta(S \setminus B)] = G(N)$ . By Theorem 2.8, G is strongly connected. Furthermore,  $G[\delta(B)]$  has  $c_M$  components and  $G[\delta(S \setminus B)]$  has  $c_N$  components, and by Theorem 4.7, the dimension of  $\mathcal{C}_V(G)$  is  $|S| - c_M - c_N + 1$ . Since  $\mathcal{C}_V(G) \subseteq Bin(\mathcal{B}(P))$  by assumption, the dimension of  $Bin(\mathcal{B}(P))$  is at least  $|S| - c_M - c_N + 1$ . If B is a common basis for M and N and M or N has a component of odd rank, then  $B \notin \mathcal{C}_V(G)$ , and thus  $Bin(\mathcal{B}(P))$  has dimension at least  $|S| - c_M - c_N + 2$  whenever M or N has a component of

odd rank. By Lemma 4.11, the dimension of  $Bin(\mathcal{B}(P))$  is therefore at least the dimension of  $(Q^M \cap Q^N) \cup (R^M \cap R^N)$ . Since  $Bin(\mathcal{B}(P)) \subseteq (Q^M \cap Q^N) \cup (R^M \cap R^N)$ , and both  $Bin(\mathcal{B}(P))$  and  $(Q^M \cap Q^N) \cup (R^M \cap R^N)$  are binary spaces, equality holds.  $\Box$ 

In Chapter 5 we discuss the set of all *integer* linear combinations of the characteristic vectors for the bases in  $\mathcal{B}(P)$ , which we call the *lattice* generated by  $\mathcal{B}(P)$ . We denote this set by Lat $(\mathcal{B}(P))$ . Most of the steps in the proof of Theorem 4.12 apply to the lattice as well, the key difference being that the binary space is a vector space. Hence if Bin(A) and Bin(A') have the same dimension and Bin $(A) \subseteq Bin(A')$ , then Bin(A) = Bin(A'). The corresponding statement is not true for the integer lattice. For example, if

$$A = \{(2,0), (0,2)\}$$
 and  $B = \{(1,0), (0,1)\},\$ 

then  $Lat(A) \subseteq Lat(B)$  and both Lat(A) and Lat(B) have dimension 2, but clearly  $Lat(A) \neq Lat(B)$ . Thus our results for the binary space do not immediately apply to the lattice.

## 4.4 An excluded minor characterization

A natural problem suggested by Theorem 4.12 is to characterize the connected matroid pairs P for which there exists a directed circuit C in G(P) with  $V(C) \notin Bin(\mathcal{B}(P))$ . We give such a characterization in this section, and show that there is essentially only one matroid pair P = (M, N) for which  $Bin(\mathcal{B}(P))$  is not completely described by  $Bin(\mathcal{B}(M))$ and  $Bin(\mathcal{B}(N))$ .

Let M be the cycle matroid for three circuits of length 2 and let N be the cycle matroid for  $K_4$ . Let  $A_M$  and  $A_N$  be the representations for M and N given by

$$A_M = \begin{bmatrix} 1 & 0 & 0 & 1 & 0 & 0 \\ 0 & 1 & 0 & 0 & 1 & 0 \\ 0 & 0 & 1 & 0 & 0 & 1 \end{bmatrix}, \qquad A_N = \begin{bmatrix} 1 & 0 & 0 & 0 & 1 & 1 \\ 0 & 1 & 0 & -1 & 0 & 1 \\ 0 & 0 & 1 & -1 & -1 & 0 \end{bmatrix}.$$

If the elements of the ground set S for M and N are labeled with respect to the columns of  $A_M$  and  $A_N$ , then the bases of the matroid pair P = (M, N) are

$$B_1 = \{1, 2, 3\}, \quad B_2 = \{1, 5, 6\}, \\ B_3 = \{2, 4, 6\}, \quad B_4 = \{3, 4, 5\}.$$

Since  $|B \cap \{4, 5, 6\}|$  is even for all bases B of P and  $|S \cap \{4, 5, 6\}|$  is odd,  $S \notin Bin(\mathcal{B}(M, N))$ . We note that P is the twined  $K_4$  pair from Example 2 in Section 3.5 and that the same pair is used in Chapter 5 to show that the lattice for  $\mathcal{B}(M, N)$  need not be equal to  $Lat(\mathcal{B}(M)) \cap Lat(\mathcal{B}(N))$ . If we consider the fundamental graph  $G(P, B_1)$  in Figure 4.2, we see that  $G(P, B_1)$  has two Hamiltonian directed circuits. Thus there exists a circuit



Figure 4.2:  $G(P, B_1)$ 

C in  $G(P, B_1)$  with V(C) = S and hence there exists a circuit C in  $G(P, B_1)$  with  $V(C) \notin Bin(\mathcal{B}(P))$ .

Let  $C_1$  and  $C_2$  be the two Hamiltonian circuits in  $G(P, B_1)$  and let G be a graph obtained by subdividing some edges in  $E(C_1) \cap E(C_2)$  into directed paths of odd length. If the matroid pair P has a basis B such that G(P, B) = G, then we call P a *twined*  $K_4$  pair. In this section we show that by excluding the twined  $K_4$  pair as a minor, the binary space for the matroid pair (M, N) is completely characterized by  $Bin(\mathcal{B}(M))$  and  $Bin(\mathcal{B}(N))$ .

Given two directed circuits C and C' in a directed graph G, the components of  $C \cap C'$ are directed paths which we call the *intersection paths* between C and C'. If the cyclic order of these paths on C is the reverse of the cyclic order of these paths on C', then we say that C and C' are *twined* circuits. In particular, the two Hamiltonian circuits of  $G(P, B_1)$  in Figure 4.2 are twined. We require one observation about twined circuits.

**Lemma 4.13.** If  $C_1$  and  $C_2$  are directed circuits with at least two intersection paths, then there exists a directed circuit  $C_3 \subset C_1 \cup C_2$  such that  $C_1$  and  $C_3$  are twined.

*Proof.* Let  $\{P_1, P_2, \ldots, P_k\}$  be the intersection paths of  $C_1$  and  $C_2$ . We may assume the paths are indexed such that their cyclic order on  $C_1$  is  $P_k, P_{k-1}, \ldots, P_1$ . Let the cyclic order of the paths on  $C_2$  be  $P_{\alpha(1)}, P_{\alpha(2)}, \ldots, P_{\alpha(k)}$  with  $\alpha(1) = 1$ . If  $\alpha(i) = i$  for  $2 \le i \le k$ , then  $C_1$  and  $C_2$  are twined. Suppose instead that  $\alpha(i) \ne i$  for some  $2 \le i \le k$ . Let m be minimum such that

$$\alpha(1) < \alpha(2) < \dots < \alpha(m-1) > \alpha(m),$$

and let  $1 \leq n < m-1$  be maximum such that  $\alpha(n) < \alpha(m)$ . Define  $P_{m,n}$  to be the directed path in  $C_1$  from the head of  $P_{\alpha(m)}$  to the tail of  $P_{\alpha(n)}$ , and let  $P_{n,m}$  be the directed path in  $C_2$  from the tail of  $P_{\alpha(n)}$  to the head of  $P_{\alpha(m)}$ . Then  $C_3 = P_{m,n}P_{n,m}$  is a directed circuit in  $C_1 \cup C_2$ , and  $C_1$  and  $C_3$  are twined.

Using Theorem 4.12, we can now show that when we exclude the twined  $K_4$  pair as a minor of the matroid pair P, we can completely characterize the binary space generated by the bases of P.

**Theorem 4.14.** If M and N are binary matroids and the connected matroid pair P = (M, N) does not have a twined  $K_4$  pair as a minor, then

$$\operatorname{Bin}(\mathcal{B}(P)) = (Q^M \cap Q^N) \cup (R^M \cap R^N).$$

Proof. Assume  $\operatorname{Bin}(\mathcal{B}(P)) \neq (Q^M \cap Q^N) \cup (R^M \cap R^N)$ , and let B be a basis of P. By Theorem 4.12, there is a directed circuit C in G(P, B) with  $V(C) \notin \operatorname{Bin}(\mathcal{B}(P))$ . Over all bases B and all circuits C in G(P, B) with  $C \notin \operatorname{Bin}(\mathcal{B}(P))$ , choose B and C to minimize |C|. Let G be the subgraph of G(P, B) induced by V(C) and let  $G_M$  and  $G_N$  be the subgraphs of G induced by E(G(M, B)) and E(G(N, B)) respectively. If e is a chord of C then we let  $C_e$  denote the unique directed circuit in  $C \cup \{e\}$  with  $e \in C_e$ .

Claim 1: If C' is a directed circuit in G and C and C' are twined, then V(C') = V(C) and there are an odd number of edges in  $C' \setminus C$ .

*Proof.* Let C' be a directed circuit in G such that C and C' are twined, and let  $E(C')\setminus E(C) = \{1,\ldots,k\}$  with the edge labels respecting the cyclic order of the edges on C'. Then  $|C_i| < |C|$  for  $1 \le i \le k$ , and thus  $V(C_i) \in Bin(\mathcal{B}(P))$  for  $1 \le i \le k$ . Since C and C' are twined,

$$V(C) = V(C')\Delta V(C_1)\Delta V(C_2)\Delta \cdots \Delta V(C_k),$$

and therefore  $V(C') \notin Bin(\mathcal{B}(M, N))$ . Since  $V(C') \subseteq V(C)$ , the minimality of |C| implies that V(C') = V(C). If k is even, as in Figure 4.3, then V(C') = V(C) implies

$$V(C) = V(C_2)\Delta\cdots\Delta V(C_k).$$

This contradicts that  $V(C) \notin Bin(\mathcal{B}(M, N))$ , and thus k must be odd. The claim follows.



Figure 4.3:  $V(C) = V(C_2)\Delta V(C_4)\Delta V(C_6)$ 

Let  $T_M$  be the perfect matching in  $G_M$  with  $T_M = E(G_M) \cap C$ , and similarly let  $T_N$ be the perfect matching in  $G_N$  with  $T_N = E(G_N) \cap C$ . Since  $V(C) \notin Bin(\mathcal{B}(P))$  and  $B \in \mathcal{B}(P), V(C)\Delta B$  is not a common basis of M and N. Without loss of generality, we may assume that  $V(C)\Delta B$  is not a basis of N. By Observation 2.2,  $G_N$  has a perfect matching  $T_N^*$  with  $T_N^* \neq T_N$ . Since  $T_M$  and  $T_N^*$  are both perfect matchings of G,  $T_M \cup T_N^*$ covers V(G) with disjoint directed circuits. Let  $T_M \cup T_N^* = C_1 \cup \ldots \cup C_m$ , where  $C_i$ is a directed circuit for  $1 \leq i \leq m$ . If  $m \geq 1$ , then  $|C_i| < |C|$  for all  $1 \leq i \leq m$ , and by the minimality of C,  $V(C_i) \in \text{Bin}(\mathcal{B}(M,N))$ . This is a contradiction, since  $V(C) = V(C_1)\Delta\cdots\Delta V(C_m)$ . Thus  $T_M \cup T_N^*$  is a directed circuit  $C^*$  with  $V(C^*) = V(C)$ . Furthermore,  $C^*$  and C have at least two intersection paths, and therefore by Lemma 4.13, G has a directed circuit that is twined with C. Over all circuits in G that are twined with C, choose the circuit C' to minimize  $|C' \setminus C|$ .

Let  $k = |C' \setminus C|$  and let  $P_1, P_2, \ldots, P_k$  be the intersection paths between C and C', where we may assume that the indices of the intersection paths respect their cyclic ordering on C'. Let  $x_i$  and  $y_i$  be the head and tail respectively of  $P_i$ . Then

$$C \setminus C' = \{y_i x_{i-1} : 1 \le i \le k\}$$
 and  $C' \setminus C = \{y_i x_{i+1} : 1 \le i \le k\},\$ 

where the subscripts are taken modulo k.

Label the edges of  $C' \setminus C$  with 1 through k such that for all i the edge  $y_i x_{i+1}$  is labeled i. Without loss of generality we may assume that  $x_1 \in B$ . If S is the ground set for P then V(G(P)) has bipartition  $\{B, S \setminus B\}$ , and since  $y_2 x_1$  is an edge of G(P), the assumption that  $x_1 \in B$  implies that  $y_2 \notin B$ . Similarly,  $y_2 \notin B$  implies that  $x_3 \in B$ , since  $y_2 x_3 \in E(C')$ . Since k is odd, we conclude that  $x_i \in B$  for all  $1 \leq i \leq k$  and  $y_i \notin B$ for all  $1 \leq i \leq k$ . See Figure 4.4, where C is the outer circuit. Note that all edges in



Figure 4.4:  $C \cup C'$ 

 $C\Delta C'$  are directed from  $S \setminus B$  to B, and thus  $E(C\Delta C') \subseteq E(G_N)$ .

Each directed path  $P_i$  has odd length, and to prove that the graph G corresponds to a twined  $K_4$  pair, it remains to show that  $G = C \cup C'$  and k = 3. We first consider edges in  $G \setminus (C \cup C')$ .

Claim 2: If e is an edge in G and  $e \notin C \cup C'$ , then e is a backwards arc of an intersection path of C and C'.

Proof. Let  $e = xy \in E(G) \setminus (C \cup C')$ . Since V(C) = V(C'), there exists  $1 \leq i, j \leq k$  such that  $x \in V(P_i)$  and  $y \in V(P_j)$ . Suppose  $i \neq j$  and let Q' be the directed path in C' from y to x. (See Figure 4.5.) Without loss of generality we may assume that j < i and thus if  $C^*$  is the directed circuit in G formed by the union of Q' with e, then  $C^*$  and C are twined with i - j + 1 intersection paths. Furthermore,  $V(C^*) \neq V(C)$ : if  $j \not\equiv i + 1$  (mod k) then  $v \notin V(C^*)$  for all  $v \in V(P_{i+1})$ , and if  $j \equiv i + 1 \pmod{k}$  then either the head of  $P_i$  or the tail of  $P_j$  is not in  $V(C^*)$ . This contradicts Claim 1, and therefore we may assume that x and y are in the same intersection path  $P_i$  of C and C'.



Figure 4.5: A forbidden chord of G

Suppose that e is a shortcut arc of  $P_i$ , as in Figure 4.6. Let Q' be the directed path



Figure 4.6: A second type of forbidden chord of G

from y to  $x_{i+1}$  in C' and let Q be the directed path in C from  $x_{i+1}$  to x. If  $C^*$  is the directed circuit in G formed by the union of Q' with Q and e, then  $C^*$  and C are twined with two intersection paths. Furthermore,  $v \notin V(C^*)$  for all  $v \in V(C) \setminus (V(P_i) \cup V(P_{i+1}))$ 

and thus  $V(C^*) \neq V(C)$ . This contradicts Claim 1 and therefore e is a backwards arc of an intersection path of C and C'.

If  $G \neq C \cup C'$ , then there exists  $xy \in E(G) \setminus (C \cup C')$  such that  $C_{xy}$  is induced. Suppose first that  $x = y_i$  and  $y = x_i$  for some  $1 \leq i \leq k$ . Without loss of generality we may assume i = 1, as in Figure 4.7, and since k is odd,

$$V(C) = V(C_{y_1x_1})\Delta V(C_2)\Delta V(C_4)\Delta \cdots \Delta V(C_{k-1}).$$
(4.16)

Since  $V(C_{y_1x_1}) \in Bin(\mathcal{B}(P))$  and  $V(C_i) \in Bin(\mathcal{B}(P))$  for all  $1 \leq i \leq k$ , Equation 4.16



Figure 4.7:  $xy = y_1 x_1$ 

contradicts the assumption that  $V(C) \notin Bin(\mathcal{B}(P))$ . Therefore  $xy \neq y_i x_i$  for any  $1 \leq i \leq k$ .

Since  $C_{xy}$  is an induced circuit of G,  $C_{xy}$  is a contributing subgraph of G(P) and  $B' = B\Delta C_{xy}$  is a basis of P. Let G' be the subgraph of G(P, B') induced by V(C), and let  $x', y' \in V(G)$  be such that  $x'y, xy' \in E(C)$ . (See Figure 4.8.) Let Q be the directed



Figure 4.8: The pivot circuit  $C_{xy}$ 

path in C from y' to x'. The next claim implies that if  $e_1 \in E(C_{xy})$  and  $e_2 \in E(Q)$ , then  $e_1$  and  $e_2$  are not in a decomposing circuit together, and thus  $E(Q) \subseteq E(G')$ .

Claim 3: Let  $xy \in E(G) \setminus (C \cup C')$  with  $xy \neq y_i x_i$  for any  $1 \leq i \leq k$ . If  $e_1$  is an edge in  $C_{xy}$ , and  $e_2$  is an internal edge in the directed path of C from x to y, then  $e_1$  and  $e_2$  are not in the same decomposing circuit.

*Proof.* Suppose by contradiction that  $e_1$  and  $e_2$  are edges in a decomposing circuit K of G and let Q be the directed path in C from x to y. Since  $e_1$  is an edge in  $C_{xy}$  and

 $e_2$  is an internal edge in Q,  $e_1$  and  $e_2$  are not incident. Let  $V(K) = \{u_1, v_1, u_2, v_2\}$  and assume that the edges and vertices of K are labeled as in Figure 4.9. Since  $xy \notin C \cup C'$ ,



Figure 4.9: The decomposing circuit  $C^*$  of Claim 3

Claim 2 implies that there exists an intersection path  $P_i$  of C and C' such that xy is a backwards arc of  $P_i$ . Since  $xy \neq y_i x_i$  and  $e_1 \in E(C_{xy})$ , either  $f_1 \notin C \cup C'$  or  $f_2 \notin C \cup C'$ . If  $f_1 \notin C \cup C'$  then by Claim 2,  $v_2 \in V(P_i)$  and  $v_2$  precedes  $u_1$  along  $P_i$ . Since  $e_2 = u_2 v_2$  is an edge of Q, the edge  $f_2 = u_2 v_1$  contradicts Claim 2. Similarly, if  $f_2 \notin C \cup C'$  then by Claim 2,  $u_2 \in V(P_i)$  and  $v_1$  precedes  $u_2$  along  $P_i$ . Since  $e_2 = u_2 v_2$  is an edge of Q, the edge  $f_1 = u_1 v_2$  contradicts Claim 2. Therefore  $e_1$  and  $e_2$  are not in a decomposing circuit of G and the claim follows.

Claim 2 implies that  $x'y' \notin E(G)$ , and therefore  $x'y' \in E(G')$ . It follows that  $C^* = Q, x'y'$  is a directed circuit in G', and

$$V(C) = V(C^*)\Delta V(C_{xy}). \tag{4.17}$$

Since  $V(C^*) \in Bin(\mathcal{B}(P))$  and  $V(C_{xy}) \in Bin(\mathcal{B}(P))$ , Equation 4.17 contradicts the assumption that  $V(C) \notin Bin(\mathcal{B}(P))$ , and therefore  $G = C \cup C'$ .

To prove that G corresponds to a twined  $K_4$  pair, it remains to show that  $|C \setminus C'| = k = 3$ . Since  $G = C \cup C'$ , the directed circuit  $C_1$  is induced in G and is therefore contributing in G(P), and thus  $B' = B\Delta V(C_1)$  is a basis of P. Let G' be the subgraph of G(P, B') induced by V(C), and let Q be the directed path in C from  $x_k$  to  $y_3$ . If  $k \ge 5$  then there are no decomposing circuits in G, and thus Q is a directed path in G'. Since  $y_1x_2, y_1x_k$  and  $y_3x_2$  are all edges in  $G_N$ , pivoting on  $y_1x_2$  creates the edge  $x_3y_k$  in G'. Thus the union of Q and  $x_3y_k$  is a directed circuit  $C^*$  in G', and

$$V(C) = V(C_1)\Delta V(C^*).$$
 (4.18)

Since  $V(C^*) \in Bin(\mathcal{B}(P))$  and  $V(C_1) \in Bin(\mathcal{B}(P))$ , Equation 4.18 contradicts the assumption that  $V(C) \notin Bin(\mathcal{B}(M, N))$ . Therefore k = 3, and the minor of P corresponding to  $G = C \cup C'$  is a twined  $K_4$  pair.

We note that the converse of Theorem 4.14 is not true. For example, let P be the matroid pair on the ground set  $S = \{1, 2, ..., 8\}$  with basis  $B = \{1, 2, 3, 4\}$  for which



Figure 4.10: G(P, B) has a twined  $K_4$  pair as a minor

G(P, B) is shown in Figure 4.10. Note that G(P, B) has six directed circuits. The vertex set for each of the six circuits is contained in  $Bin(\mathcal{B}(P))$ , and thus by Theorem 4.12,  $Bin(\mathcal{B}(P)) = (Q^M \cap Q^N) \cup (R^M \cap R^N)$ . However, the matroid pair  $P/\{4\}\setminus\{8\}$  is a twined  $K_4$  pair.

A corollary of Theorem 4.14 is that Pfaffian matroid pairs with no twined  $K_4$  minor have essentially unique Pfaffian signings.

**Corollary 4.15.** If P = (M, N) is a connected and Pfaffian matroid pair on the ground set S and P does not have a twined  $K_4$  pair as a minor, then  $X \subseteq S$  is a constant-parity intersecting set of P if and only if  $X = X_M \cap X_N$  for some separator  $X_M$  of M and  $X_N$ of N.

*Proof.* This follows from Theorems 4.10 and 4.14.

# Chapter 5 Lattices

A description of the convex hull generated by the characteristic vectors for a combinatorial structure can lead to efficient algorithms for problems defined on the underlying combinatorial structure. For example, the convex hull generated by the characteristic vectors for independent sets in a matroid pair has a beautiful description due to Edmonds [10], and this description leads to an efficient method of finding an independent set of maximum weight. In this chapter we focus on a discrete analogue of convex hulls called *lattices*. Lattices associated with combinatorial structures were first studied by Lovász [26], [25], in the context of perfect matchings.

Given a set of vectors A, the lattice of A contains all integer linear combinations of vectors in A. Formally, if  $A = \{a_1, \ldots, a_m\}$  with  $a_i \in \mathbb{R}^n$  for  $1 \leq i \leq m$ , then the lattice generated by A, denoted Lat(A), is defined as

$$\operatorname{Lat}(A) = \left\{ \sum_{i=1}^{m} \lambda_i a_i : \lambda_i \in \mathbb{Z} \right\}.$$

When A is a matrix, Lat(A) denotes the lattice generated by the columns of A. A basis of a lattice is a minimum set of vectors that generates the lattice, and if the vectors in the set A are rational, then a basis of Lat(A) can be found in polynomial time (Lovász [26]).

Lovász [25] showed that when G is bipartite and  $\mathcal{M}(G)$  is the set of characteristic vectors for the perfect matchings of G then the lattice generated by  $\mathcal{M}(G)$  is the set of integer points in the space generated by all linear combinations of vectors in  $\mathcal{M}(G)$ . A description of the lattice generated by the characteristic vectors of a combinatorial structure is an extremely powerful tool for studying combinatorial problems; here we consider the lattice generated by the characteristic vectors for the bases of a matroid pair.

If P is a matroid pair then we denote the set of characteristic vectors for the bases of P by  $\mathcal{B}(P)$ . Our interest in Lat $(\mathcal{B}(P))$  is motivated by Pfaffian signings: the *constant*parity intersecting sets of P are those subsets X of the ground set of P for which either  $|B \cap X|$  is even for all bases B of P, or  $|B \cap X|$  is odd for all bases B of P. Two Pfaffian representations of P are equivalent up to resigning across a constant-parity intersecting set, and a characterization of  $\text{Lat}(\mathcal{B}(P))$  implies a characterization of the constant-parity intersecting sets. Characterizing these sets is therefore a vital step towards constructing an algorithm to find a Pfaffian representation of P. The constant-size intersecting sets are closely related to the constant-parity intersecting sets: a subset X of the ground set of P is a constant-parity intersecting set of P if and only if there exists an integer k such that  $|X \cap B| = k$  for every basis B of P. In this chapter we characterize the constant-size intersecting sets for any connected matroid pair (M, N) in terms of the separators for Mand N.

## 5.1 Matroid polytopes

When M is a matroid on the ground set S, the convex hull of  $\mathcal{B}(M)$ , denoted conv $(\mathcal{B}(M))$ , has a simple description. If  $x \in \text{conv}(\mathcal{B}(M))$  and  $T \subseteq S$ , then let x(T) denote the sum of  $x_s$  over all  $s \in T$ . Note that for all  $x \in \text{conv}(\mathcal{B}(M))$ , x is nonnegative, x(S) = r, and  $x(T) \leq \text{rank}_M(T)$  for all  $T \subseteq S$ . The Matroid Polytope Theorem of Edmonds [10] states that these necessary conditions are sufficient, and therefore

$$\operatorname{conv}(\mathcal{B}(M)) = \{ x \in \mathbb{R}^S : x(T) \le \operatorname{rank}_M(T) \ \forall \ T \subseteq S, \ x(S) = r, \ x \ge 0 \}.$$

If P is the matroid pair (M, N), then  $\mathcal{B}(P) = \mathcal{B}(M) \cap \mathcal{B}(N)$  and therefore

$$\operatorname{conv}(\mathcal{B}(P)) \subseteq \operatorname{conv}(\mathcal{B}(M)) \cap \operatorname{conv}(\mathcal{B}(N)).$$
(5.1)

The Matroid Intersection Polytope Theorem of Edmonds [10] states that the containment in 5.1 is met with equality.

**Theorem 5.1 (Matroid Intersection Polytope Theorem).** If M and N are rank r matroids on the same ground set S and P is the matroid pair (M, N), then

$$\operatorname{conv}(\mathcal{B}(P)) = \{ x \in \mathbb{R}^S : x(T) \leq \operatorname{rank}_M(T) \quad \forall T \subseteq S, \\ x(T) \leq \operatorname{rank}_N(T) \quad \forall T \subseteq S, \\ x(S) = r, \\ x \geq 0 \}$$

and the linear system which defines  $\operatorname{conv}(\mathcal{B}(P))$  is totally dual integral.

A rational linear system  $Ax \leq b$  is totally dual integral if the linear program

$$\min\{y^{\top}b: y^{\top}A = w, \ y \ge 0\}$$

has an integral optimal solution y whenever w is integral and an optimal solution exists. Suppose that  $Ax \leq b$  is a totally dual integral system. If the vector b is integral and the polytope given by  $\{x : Ax \leq b\}$  is a rational polytope, then  $\max\{w^{\top}x : Ax \leq b\}$  has an integral optimal solution for all integral vectors w. Totally dual integral systems are therefore extremely valuable in linear programming.

When P is the matroid pair (M, N),

$$\operatorname{Lat}(\mathcal{B}(P)) \subseteq \operatorname{Lat}(\mathcal{B}(M)) \cap \operatorname{Lat}(\mathcal{B}(N)),$$

and therefore it is natural to consider if the Matroid Intersection Polytope Theorem has an analogous theorem in lattices. However, there are matroid pairs P for which  $Lat(\mathcal{B}(P))$  is strictly contained in  $Lat(\mathcal{B}(M)) \cap Lat(\mathcal{B}(N))$ . For example, let M and Nbe the matroids with representations  $A_M$  and  $A_N$  over the reals, as given below:

$$A_M = \begin{bmatrix} 1 & 0 & 0 & 1 & 0 & 0 \\ 0 & 1 & 0 & 0 & 1 & 0 \\ 0 & 0 & 1 & 0 & 0 & 1 \end{bmatrix}, \qquad A_N = \begin{bmatrix} 1 & 0 & 0 & 0 & 1 & 1 \\ 0 & 1 & 0 & -1 & 0 & 1 \\ 0 & 0 & 1 & -1 & -1 & 0 \end{bmatrix}.$$

Since  $(1, 1, 1, 0, 0, 0) \in \mathcal{B}(M)$  and  $(0, 0, 0, 1, 1, 1) \in \mathcal{B}(M)$ , the vector of all ones is contained in Lat $(\mathcal{B}(M))$ . Similarly, (1, 1, 0, 1, 0, 0) and (0, 0, 1, 0, 1, 1) are both elements in  $\mathcal{B}(N)$  and thus the vector of all ones is contained in Lat $(\mathcal{B}(N))$ . It follows that  $(1, 1, 1, 1, 1, 1) \in \text{Lat}(\mathcal{B}(M)) \cap \text{Lat}(\mathcal{B}(N))$ . However, if P is the matroid pair (M, N), then

$$\mathcal{B}(P) = \{ (1, 1, 1, 0, 0, 0), (1, 0, 0, 0, 1, 1), (0, 1, 0, 1, 0, 1), (0, 0, 1, 1, 1, 0) \}$$

and if the  $i^{th}$  columns of  $A_M$  and  $A_N$  correspond to element i in the ground set for Mand N, then we see that  $x(\{4, 5, 6\})$  is even for each  $x \in \mathcal{B}(P)$ . Thus  $x(\{4, 5, 6\})$  is even for all  $x \in \text{Lat}(\mathcal{B}(P))$  and this implies that  $(1, 1, 1, 1, 1, 1) \notin \text{Lat}(\mathcal{B}(P))$ . It follows that  $\text{Lat}(\mathcal{B}(P)) \neq \text{Lat}(\mathcal{B}(M)) \cap \text{Lat}(\mathcal{B}(N))$  in this case. We note that P is a twined- $K_4$  pair, as given in Example 2 of Section 3.5.

As a partial result towards a description of  $Lat(\mathcal{B}(P))$ , we show that if S' is a constantsize intersecting set of P, then S' can be characterized by the separators of P. We note that lattice generated by  $\mathcal{B}(M)$  has been well characterized by Rieder [33] and depends only on the components of M and the rank of these components.

**Theorem 5.2 (Rieder).** Let M be a rank r matroid on the ground set S and let  $\{S_1, S_2, \ldots, S_k\}$  be the partition of S corresponding to the components of M. Then  $Lat(\mathcal{B}(M))$  is the set of  $x \in \mathbb{Z}^S$  satisfying all of the following:

(i) 
$$x(S) \equiv 0 \pmod{r}$$
,

- (*ii*)  $x(S_i) \equiv 0 \pmod{\operatorname{rank}_M(S_i)} \forall 1 \le i \le k$ ,
- (*iii*)  $\operatorname{rank}_M(S_j)x(S_i) = \operatorname{rank}_M(S_i)x(S_j) \ \forall \ 1 \le i, j \le k.$

#### 5.2 Separators and constant-size intersecting sets

Let M and N be rank r matroids on the ground set S and let P be the matroid pair (M, N). Recall from Section 2.6 that a *separator* of M is a subset T of S such that

$$\operatorname{rank}_M(T) + \operatorname{rank}_M(S \setminus T) = r$$

and a *non-trivial* separator is a non-empty separator that is not equal to the ground set. A *component* of M is a nontrivial inclusion minimal separator. Similarly, a separator of P is a subset T of S such that

$$\operatorname{rank}_M(T) + \operatorname{rank}_N(S \setminus T) = r.$$

We wish to characterize the constant-size intersecting sets of P, that is, we wish to characterize those subsets T of S for which x(T) is constant for all  $x \in \mathcal{B}(P)$ . When Pis the pair (M, M) the characterization of the constant-size intersecting sets is simple: there exists a basis of M that intersects  $T \subseteq S$  in  $\operatorname{rank}_M(T)$  elements and thus if  $|T \cap B|$ is constant for all bases B of M then  $|T \cap B| = \operatorname{rank}_M(T)$  for all B. Furthermore, when  $|T \cap B|$  is constant for all bases B then  $|(S \setminus T) \cap B|$  is constant and therefore all bases intersect  $S \setminus T$  with cardinality equal to the rank of  $S \setminus T$ . Thus

$$|T \cap B| + |(S \setminus T) \cap B| = \operatorname{rank}_M(T) + \operatorname{rank}_M(S \setminus T).$$

Since  $|T \cap B| + |(S \setminus T) \cap B| = |B| = r$ , it follows that the constant-size intersecting sets of the matroid pair (M, M) are exactly the separators of M. When  $M \neq N$  and  $T \subseteq S$ has the property that  $|T \cap B| = k$  for all bases B of P, then certainly  $k \leq \operatorname{rank}_M(T)$ and  $k \leq \operatorname{rank}_N(T)$ , but it need not be that  $k = \operatorname{rank}_M(T)$  or  $k = \operatorname{rank}_N(T)$ . Using the Matroid Intersection Polytope Theorem (Theorem 5.1) and linear programming, we characterize the constant-size intersecting sets for all connected matroid pairs. Unlike the matroid pairs discussed in most of this thesis, in this section we neither assume the matroids to be regular or representable.

The total dual integrality of the linear system for the convex hull of  $\mathcal{B}(P)$  given by Edmonds' Matroid Intersection Polytope Theorem implies that the linear program

Maximize 
$$c^T x$$
  
subject to (5.2)  
 $x \in \operatorname{conv}(\mathcal{B}(P))$ 

has an optimal solution that is the characteristic vector for a basis of P. From the characterization of  $conv(\mathcal{B}(P))$  given by the Matroid Intersection Polytope Theorem, the

dual of the maximization problem (5.2) is

Minimize 
$$\sum_{T \subseteq S} (y_T^M \operatorname{rank}_M(T) + y_T^N \operatorname{rank}_N(T)) + zr$$
  
subject to  
$$\sum_{T \subseteq S: e \in T} (y_T^M + y_T^N) + z \ge c_e \quad \forall e \in S,$$
  
$$y^M, y^N \ge 0.$$
 (5.3)

If the basis B of P gives an optimal solution to (5.2) and  $y^M, y^N, z$  is an optimal solution to (5.3), then the complementary slackness conditions between (5.2) and (5.3) imply that

$$\sum_{T \subseteq S: e \in T} (y_T^M + y_T^N) + z = c_e \ \forall \ e \in B,$$
$$|T \cap B| = \operatorname{rank}_M(T) \text{ if } y_T^M > 0,$$
$$|T \cap B| = \operatorname{rank}_N(T) \text{ if } y_T^N > 0.$$

If  $T \subseteq S$  is a separator of M then  $|B \cap T| = \operatorname{rank}_M(T)$  for all bases B of M and therefore  $|B \cap T| = \operatorname{rank}_M(T)$  for all bases B of P. The next lemma shows that the converse holds when P is connected.

**Lemma 5.3.** If P = (M, N) is a connected pair of matroids on the ground set S and  $T \subseteq S$  is such that  $|T \cap B| = \operatorname{rank}_M(T)$  for all bases B of P, then T is a separator of M.

*Proof.* Let  $\mathcal{S}_M$  contain the subsets T of S which satisfy  $|T \cap B| = \operatorname{rank}_M(T)$  for all bases B of P and let  $\mathcal{S}_N$  contain all  $U \subseteq S$  such that  $|U \cap B| = \operatorname{rank}_N(U)$  for all bases B of P.

Claim 1:  $S_M$  and  $S_N$  are closed under union and intersection. Proof. If  $T, T' \in S_M$  and B is a basis of P, then

$$\operatorname{rank}_{M}(T \cup T') + \operatorname{rank}_{M}(T \cap T') \ge |(T \cup T') \cap B| + |(T \cap T') \cap B|$$
$$= |T \cap B| + |T' \cap B|$$
$$= \operatorname{rank}_{M}(T) + \operatorname{rank}_{M}(T').$$

The submodularity of the rank function implies that

$$\operatorname{rank}_M(T) + \operatorname{rank}_M(T') \ge \operatorname{rank}_M(T \cup T') + \operatorname{rank}_M(T \cap T')$$

and therefore  $|(T \cup T') \cap B| = \operatorname{rank}_M(T \cup T')$  and  $|(T \cap T') \cap B| = \operatorname{rank}_M(T \cap T')$ . Since B was arbitrary,  $T \cup T' \in S_M$  and  $T \cap T' \in S_M$  and thus  $S_M$  is closed under intersection and union. The symmetric argument for N shows that  $S_N$  is similarly closed under intersection and union.

Claim 2: If  $T \in S_M$  is nontrivial, then  $S \setminus T \notin S_N$ . *Proof.* Assume that P has rank r and let  $T \subseteq S$  be such that  $T \in S_M$  and  $S \setminus T \in S_N$ . For all bases B of P

$$r = |T \cap B| + |(S \setminus T) \cap B| = \operatorname{rank}_M(T) + \operatorname{rank}_N(S \setminus T).$$

The assumption that P is connected therefore implies that either T = S or  $T = \emptyset$ , and  $S \setminus T \notin S_N$  whenever  $T \in S_M$  is nontrivial.

Claim 3: If  $T \in S_M$  and  $S \setminus T \in S_M$  then T is a separator of M. Proof. Assume that T and  $S \setminus T$  are both elements in  $S_M$ . If B be is a basis of P, then

$$r = |T \cap B| + |(S \setminus T) \cap B| = \operatorname{rank}_M(T) + \operatorname{rank}_M(S \setminus T)$$

and thus T is a separator of M whenever  $T \in \mathcal{S}_M$  and  $S \setminus T \in \mathcal{S}_M$ .

Let  $R \in \mathcal{S}_M$  and consider the linear program

$$\begin{array}{ll}
\text{Maximize} & x(S \setminus R) \\
& \text{subject to} \\
& x \in \operatorname{conv}(\mathcal{B}(P))
\end{array} \tag{5.4}$$

and its dual

Minimize 
$$\sum_{T \subseteq S} (y_T^M \operatorname{rank}_M(T) + y_T^N \operatorname{rank}_N(T)) + zr$$
  
subject to (5.5)  
$$\sum_{T \subseteq S: e \in T} (y_T^M + y_T^N) \ge \begin{cases} 1 - z, & \text{if } e \notin R; \\ -z, & \text{if } e \in R; \end{cases}$$
$$y^M, y^N \ge 0.$$

We show that  $S \setminus R \in S_M$  and thus by Claim 3, R is a separator of M. If  $x \in \mathcal{B}(P)$ , then

$$x(S \setminus R) = x(S) - x(R) = r - \operatorname{rank}_M(R)$$

and therefore  $x(S \setminus R) = r - \operatorname{rank}_M(R)$  for all  $x \in \operatorname{conv}(\mathcal{B}(P))$ . Every feasible solution for (5.4) is therefore optimal and in particular every basis of P is optimal for (5.4). For all  $e \in S$  the connectivity of P therefore implies an optimal solution  $x \in \mathcal{B}(P)$  with  $x_e = 1$ .

Let  $y^M, y^N, z$  be an optimal solution to (5.5). By the total dual integrality of the linear system for  $\operatorname{conv}(\mathcal{B}(P))$  we may assume that  $y^M, y^N$  and z are integral and thus we can let  $\mathcal{R}$  be the multiset defined on the subsets of S such that  $T \subseteq S$  has multiplicity  $y_T^M + y_T^N$  in  $\mathcal{R}$ . The complementary slackness conditions between (5.4) and (5.5) imply the following:

$$\sum_{T \subseteq S: e \in T} (y_T^M + y_T^N) = \begin{cases} 1 - z, & \text{if } e \in S \setminus R; \\ -z, & \text{else;} \end{cases}$$
(5.6)

If 
$$y_T^N > 0$$
, for some  $T \subseteq S$ , then  $|B \cap T| = \operatorname{rank}_M(T)$  for all  $B \in \mathcal{B}(P)$ , (5.7)

If 
$$y_T^N > 0$$
, for some  $T \subseteq S$ , then  $|B \cap T| = \operatorname{rank}_N(T)$  for all  $B \in \mathcal{B}(P)$ . (5.8)

Conditions (5.7) and (5.8) imply that the sets in  $\mathcal{R}$  are from  $\mathcal{S}_M \cup \mathcal{S}_N$ . Condition (5.6) implies that each element of  $S \setminus R$  is covered by exactly -z+1 sets in  $\mathcal{R}$  and each element in R is covered exactly -z times by  $\mathcal{R}$ . (Note that the nonnegativity requirement of  $y^M$  and  $y^N$  implies that  $-z \ge 0$ .) Over all multisets on  $\mathcal{S}_M \cup \mathcal{S}_N$  that cover the elements of  $S \setminus R$  exactly k+1 times and cover the elements of R exactly k times for some integer k, choose the multiset  $\mathcal{R}'$  to minimize k. Let  $\mathcal{R}_M = \mathcal{R}' \cap \mathcal{S}_M$  and let  $\mathcal{R}_N = \mathcal{R}' \setminus \mathcal{R}_M$ .

Claim 4: We may assume that  $\mathcal{R}_M$  is such that either  $T \subseteq T'$  or  $T' \subseteq T$  for all  $T, T' \in \mathcal{R}_M$ .

*Proof.* If  $T, T' \in \mathcal{R}_M$  are such that  $T \setminus T'$  and  $T' \setminus T$  are both nonempty, then let

$$\mathcal{R}'_M = \mathcal{R}_M \cup \{T \cup T', T \cap T'\} \setminus \{T, T'\}.$$

Each  $e \in S$  is covered by the same number of sets of  $\mathcal{R}'_M$  as by  $\mathcal{R}_M$  and since  $\mathcal{S}_M$  is closed under union and intersection, the elements of  $\mathcal{R}'_M$  are all contained in  $\mathcal{S}_M$ .

We can similarly assume that  $T \subseteq T'$  or  $T' \subseteq T$  for all  $T, T' \in \mathcal{R}_N$ . The minimality of k implies that  $S \notin \mathcal{R}_M$  and  $S \notin \mathcal{R}_N$  and thus neither  $\mathcal{R}_M$  nor  $\mathcal{R}_N$  is a cover of S.

Claim 5:  $\mathcal{R}_N$  is empty.

Proof. If  $\mathcal{R}_N$  is not empty then let  $\mathcal{R}'_M = \mathcal{R}_M \cup R$  and consider the multiset  $\mathcal{R}'_M \cup \mathcal{R}_N$ which covers each element of S exactly k + 1 times. If there exists  $T, T' \in \mathcal{R}'_M$  with  $T \setminus T' \neq \emptyset$  and  $T' \setminus T \neq \emptyset$  then  $\mathcal{R}' \setminus \{T, T'\} \cup \{T \cup T', T \cap T'\}$  is also a k + 1 cover of Swith all elements from  $\mathcal{R}_M \cup \mathcal{R}_N$ . Thus we may again assume that  $T \subseteq T'$  or  $T' \subseteq T$ for all  $T, T' \in \mathcal{R}'_M$ , although we can no longer use minimality to conclude that S is not contained in  $\mathcal{R}'_M$ .

Let T be a minimal element of  $\mathcal{R}'_M$  and let U be a maximal element of  $\mathcal{R}_N$ . If  $|\mathcal{R}'_M| = m$  and  $e \in S \setminus (T \cup U)$ , then e is covered by at most m-1 elements of  $\mathcal{R}'_M$  and

*e* is not covered by  $\mathcal{R}_N$ . This contradicts the assumption that  $\mathcal{R}'$  is a k + 1 cover, since each element of T is covered m times by  $\mathcal{R}'_M$ . It follows that  $T \cup U = S$  and since the minimality assumption implies that  $U \neq S$ , there exists  $e \in T \setminus U$ . Any element in  $T \setminus U$  is covered exactly m times in  $\mathcal{R}'$  and thus  $\mathcal{R}'$  is an m cover of all elements in S. If  $e \in T \cap U$ then e is covered m times by  $\mathcal{R}'_M$  and at least once by  $\mathcal{R}_N$  and thus  $T \cap U$  is empty and  $\{T, U\}$  is a partition of S. Since T and U are nontrivial, this partition contradicts the assumption that P is connected, and thus  $\mathcal{R}_N$  is empty.

Since  $\mathcal{R}_N$  is empty,  $\mathcal{R}_M$  covers all elements in  $S \setminus R$  exactly k+1 times and all elements in R exactly k times. Since  $\mathcal{R}_M$  is not a cover of S, k = 0 and therefore m = 1 and  $\mathcal{R}_M = S \setminus R$ . It follows that  $S \setminus R \in \mathcal{S}_M$ . Since  $R \in \mathcal{S}_M$  by assumption, Claim 3 implies that R is a separator of M.

By symmetry, Lemma 5.3 implies that if P = (M, N) is a connected matroid pair on the ground set S and  $U \subseteq S$  is such that  $|U \cap B| = \operatorname{rank}_N(U)$  for all bases B of P, then U is a separator of N.

Suppose that T and U are separators of M and N respectively with  $T \cup U$  spanning S. If  $R = T \cap U$  and M and N have rank r, then for all bases B of P,

$$|B \cap R| = |B \cap (T \cap U)|$$
  
=  $|B \cap T| + |B \cap U| - |B \cap (T \cup U)|$   
=  $\operatorname{rank}_M(T) + \operatorname{rank}_N(U) - r.$ 

Thus every basis of P intersects R with fixed cardinality. We claim that whenever P is connected and  $R \subseteq S$  is a constant-size intersecting set for P, then there exists a partition  $\{R_1, R_2, \ldots, R_k\}$  of R and separators  $T_i$  of M and  $U_i$  of N such that  $R_i = T_i \cap U_i$  and  $T_i \cup U_i = S$  for  $1 \leq i \leq k$ . A constant-size intersecting set R is *minimal* if there does not exist a non-trivial subset R' of R such that R' is also a constant-size intersecting set and it therefore suffices to consider minimal constant-size intersecting sets.

**Theorem 5.4.** Let P = (M, N) be a connected matroid pair on the ground set S and suppose that  $R \subseteq S$  is a constant-size intersecting set of P. If R is minimal, then there exist separators T and U of M and N respectively such that  $T \cap U = R$  and  $T \cup U = S$ .

*Proof.* Consider the linear program

Maximize 
$$x(R)$$
  
subject to (5.9)  
 $x \in \operatorname{conv}(\mathcal{B}(P)).$ 

If  $|B \cap R|$  is constant for all bases B of P then x(R) is constant for all  $x \in \text{conv}(\mathcal{B}(P))$ , and therefore every feasible solution to (5.9) is optimal. In particular, every basis of P is an optimal solution. If P has rank r, then the dual of (5.9) is

$$\begin{aligned} \text{Minimize} \quad & \sum_{T \subseteq S} (y_T^M \operatorname{rank}_M(T) + y_T^N \operatorname{rank}_N(T)) + zr \\ & \text{subject to} \end{aligned} \tag{5.10} \\ & \sum_{T \subseteq S: e \in T} (y_T^M + y_T^N) + z \geq \begin{cases} 1, & \text{if } e \in R, \\ 0, & \text{else}, \end{cases} \\ & y^M, y^N \geq 0. \end{aligned}$$

Let  $y^M, y^N, z$  be an optimal solution to (5.10). By the total dual integrality of the linear system for conv( $\mathcal{B}(P)$ ) we may assume that  $y^M, y^N$  and z are integral. If  $e \in S$  then the connectivity of P implies a basis B of P with  $e \in B$  and thus for all  $e \in S$  there exists an optimal solution x to (5.9) with  $x_e > 0$ . The complementary slackness conditions between (5.9) and (5.10) therefore imply that

$$\sum_{T \subseteq S: e \in T} (y_T^M + y_T^N) = \begin{cases} 1 - z, & \text{if } e \in R; \\ -z, & \text{else,} \end{cases}$$
(5.11)

$$y_T^M > 0$$
 only if  $|B \cap T| = \operatorname{rank}_M(T)$  for all bases  $B$  of  $P$ , (5.12)

$$y_U^N > 0$$
 only if  $|B \cap U| = \operatorname{rank}_N(U)$  for all bases B of P. (5.13)

Since P is connected, Lemma 5.3 simplifies conditions (5.12) and (5.13) to

$$y_T^M > 0$$
 only if  $T$  is a separator of  $M$   
 $y_U^N > 0$  only if  $U$  is a separator of  $N$   
(5.14)

Let  $\mathcal{R}_M$  be the multiset on the separators of M such that the separator  $T \subseteq S$  of M has multiplicity  $y_T^M$  in  $\mathcal{R}_M$ . Similarly, let  $\mathcal{R}_N$  be the multiset on the separators of N such that the separator  $U \subseteq S$  of N has multiplicity  $y_U^N$  in  $\mathcal{R}_N$ . Condition (5.6) implies that the multiset  $\mathcal{R}_M \cup \mathcal{R}_N$  covers all  $e \in R$  exactly -z + 1 times and e is covered exactly -z times by  $\mathcal{R}_M \cup \mathcal{R}_N$  when  $e \in S \setminus R$ . Thus there exists an integer k and a multiset  $\mathcal{R}$  such that the elements of  $\mathcal{R}$  are separators of either M or N and  $e \in S$  is covered exactly k+1 times by  $\mathcal{R}$  when  $e \in R$  and e is covered exactly k times by  $\mathcal{R}$  when  $e \in S \setminus R$ . Over all such multisets, choose  $\mathcal{R}$  to minimize k and let  $\mathcal{R}_M = \mathcal{R} \cap \mathcal{S}_M$  and let  $\mathcal{R}_N = \mathcal{R} \setminus \mathcal{R}_M$ .

If there exist  $T, T' \in \mathcal{R}_M$  with both  $T \setminus T'$  and  $T' \setminus T$  nonempty then let the multiset  $\mathcal{R}'_M$  be defined by

$$\mathcal{R}'_M = \mathcal{R}_M \cup \{T \cup T', T \cap T'\} \setminus \{T, T'\}.$$

Each element of S is covered by the same number of sets in  $\mathcal{R}'_M$  as it is covered by in  $\mathcal{R}_M$ . Furthermore, if T and T' are separators of M, then  $T \cup T'$  and  $T \cap T'$  are separators of
M and thus the elements of  $\mathcal{R}'_M$  are all separators of M. We may therefore assume that either  $T \subseteq T'$  or  $T' \subseteq T$  for all  $T, T' \in \mathcal{R}_M$  and if  $|\mathcal{R}_M| = m$  then  $\mathcal{R}_M = \{T_1, T_2, \ldots, T_m\}$ with  $T_i \subseteq T_{i+1}$  for  $1 \leq i \leq m-1$ . If  $|\mathcal{R}_N| = n$  then we can similarly assume that  $\mathcal{R}_N = \{U_1, U_2, \ldots, U_n\}$  with  $U_i \subseteq U_{i+1}$  for  $1 \leq i \leq n-1$ . The minimality of k implies that  $T_m \neq S$  and  $U_n \neq S$ .

If  $e \in T_1$  then e is covered m times by  $\mathcal{R}_M$  and is therefore covered at least m times by  $\mathcal{R}$ , and hence  $m \leq k + 1$ . Suppose m = k + 1. Since all elements of  $T_1$  are therefore covered k + 1 times in  $\mathcal{R}_M$  and only elements of R are covered k + 1 times by  $\mathcal{R}$ , every element in  $T_1$  must be contained in R. Since  $T_1$  is a separator of M, all bases of Pintersect  $T_1$  with fixed cardinality, and since R is minimal with this property,  $T_1 \subseteq R$ implies  $T_1 = R$  and thus R is a separator of M when m = k + 1. The elements in  $S \setminus U_n$ are covered only by  $\mathcal{R}_M$  and are therefore covered at most m times by  $\mathcal{R}$  implying that  $m \geq k$ , and thus either m = k or R is a separator of M. If R is a separator of M then the lemma follows by taking the trivial separator U = S of N. The symmetric argument with  $\mathcal{R}_N$  shows that either n = k or R is a separator of N.

Hence we may assume that n = m = k. Since all elements of S are covered at least k times by  $\mathcal{R}$  and all elements in  $S \setminus (T_1 \cup U_k)$  are covered at most k - 1 times by  $\mathcal{R}$ , the set  $T_1 \cup U_k$  spans S. Since  $T_1 \neq S$  and  $U_k \neq S$ , both  $T_1$  and  $U_k$  are nontrivial. If  $T_1 \cap U_k$  is empty then  $\{T_1, U_k\}$  is a partition of S and for all bases B of P,

$$r = |B \cap T_1| + |B \cap U_k| = \operatorname{rank}_M(T_1) + \operatorname{rank}_N(U_k).$$
(5.15)

Since  $T_1$  and  $U_k$  are nontrivial, (5.15) contradicts the assumption that P is connected and thus  $T_1 \cap U_k$  is nonempty. All elements in  $T_1 \cap U_k$  are covered by at least k + 1 elements of  $\mathcal{R}$  and are thus covered by exactly k + 1 elements of  $\mathcal{R}$  and therefore  $T_1 \cap U_k \subseteq R$ . Since  $T_1$  and  $U_k$  are separators of M and N respectively and  $T_1 \cup U_k = S$ , all bases of P intersect  $T_1 \cap U_k$  with fixed cardinality and the minimality of R with respect to this property implies that  $R = T_1 \cap U_k$ . Thus either R is a separator of M or N or there exist separators T and U of M and N respectively such that  $T \cup U$  spans S and  $R = T \cap U$ .  $\Box$ 

# Chapter 6 Series-Parallel Matroids

Many of the difficulties associated with finding a Pfaffian representation of a regular matroid pair are simplified in the case of graphic matroids pairs P = P(G) where Gis a bipartite graph and the matroids in the pair P(G) are the partition matroids for a bipartite graph, as defined in Section 3.3. For example, finding a totally unimodular edge signing of the fundamental graph for P is easy in this case since the even edge signing (all edges are signed 0, Section 2.5) of G(P) is totally unimodular when P is a graphic matroid pair. Recall from Section 2.3 that if S is the ground set for P and Bis a basis for P, then the *contributing* sets of G(P, B) correspond to the subsets  $H \subseteq S$ for which  $B\Delta H$  is a basis of P. When P = (M, N) is a graphic matroid pair, then the fundamental graphs G(M) and G(N) are both acyclic, and thus by the observation that  $H \subseteq S$  is contributing if and only if  $G(M)[H\Delta B]$  and  $G(N)[H\Delta B]$  each have an odd number of perfect matchings (Observation 2.2), identifying the contributing sets for G(P, B) is simple:  $X\Delta B$  is a basis of P if and only if there is a unique set D of disjoint directed circuits in G(P, B) such that V(D) = X.

In this section we consider a minor and duality closed class of regular matroid pairs for which both of these properties hold; namely the even edge signing is a totally unimodular signing and the contributing sets are easily identified in the fundamental graph. The matroids in this class are called *series-parallel* matroid pairs. By exploiting these properties of series-parallel matroid pairs we get some characterizations of G(P) when P is non-Pfaffian, and we show that when P is Pfaffian then a Pfaffian signing of G(P)has a simple description in terms of signings of directed circuits. We first require some definitions and characterizations of series-parallel matroids.

Let G be a graph. Recall that for  $W \subseteq V(G)$ , the edges in E(G) with exactly one end in W are denoted  $\delta(W)$ , and  $\delta(W)$  is called an *edge cut* of G. The edge cut  $\delta(W)$  is *minimal* if there is no nonempty edge cut  $\delta(W')$  with  $\delta(W') \subset \delta(W)$ . Two edges e and f in G are said to be in *series* if  $\{e, f\}$  is a minimal edge cut of G, and if e and f are in series in G and the graph H is such that H = G/e, then G is a *series extension* of H. Two edges in

G are parallel if they are incident to exactly the same two vertices and parallel extensions of graphs have an analogous definition to series extensions: if e and f are parallel edges in G and  $G \setminus e = H$ , then G is a parallel extension of H. A series-parallel graph is a graph which can be obtained from a single edge by a sequence of parallel extensions and series extensions. Electrical networks constructed by adding resistors in series and in parallel correspond to series-parallel graphs, and thus series-parallel graphs are studied for their applications in electrical engineering. The recursive structure of series-parallel graphs is useful in implementing algorithms, and is another motivation for the study of series-parallel graphs.

If H is a 2-connected graph then all series and parallel extensions of H are 2-connected. Since the graph consisting of a single edge is 2-connected, all series-parallel graphs are 2-connected. (Note that the path of length 2 is not a series extension of the single edge, since the two edge cut in the path is not a minimal cut.) Let  $K_4$  be the complete graph on four vertices. Since  $K_4$  does not contain edges in series and  $K_4$  does not have any parallel edges,  $K_4$  is not series-parallel. Dirac [7], Adam [1] and Duffin [8] independently showed that a graph is series-parallel if and only if it is 2-connected and does not have a  $K_4$  minor.

A matroid M is series-parallel if M = M(G) and each connected component of the graph G is series-parallel. Since minors of M(G) correspond to minors of G, it follows from the definition that series-parallel matroids do not have a  $M(K_4)$  minor. A stronger characterization holds.

**Theorem 6.1.** A matroid is series-parallel if and only if it does not have a minor isomorphic to  $M(K_4)$  or  $U_{2,4}$ .

The matroid  $U_{2,4}$  is the rank two matroid on four elements such that any two elements of the ground set form a basis. Since series-parallel matroids are graphic and graphic matroids are binary, one direction of Theorem 6.1 follows from Tutte's [43] characterization of binary matroids together with the excluded minor characterization for series-parallel graphs. The converse direction uses a connectivity result and Tutte's Wheels and Whirls theorem [42]. For a proof of Theorem 6.1, including a proof of the excluded minor characterization of series-parallel graphs, see Oxley [31], page 363.

The matroid pair P = (M, N) is series-parallel if M and N are series-parallel matroids. When P is series-parallel and minimally non-Pfaffian we show that either G(P) has a Hamiltonian directed circuit (Section 6.6) or G(P) has a "weak odd double cycle" (defined in Section 6.7) as a spanning subgraph (Section 6.7). In the case that G(P)has a Hamiltonian directed circuit there is a non-Pfaffian certificate of constant size that proves that P is Pfaffian and when G(P) has a weak odd double cycle then the number of circuits composing the double cycle is small.

Let G be a bipartite graph with vertex bipartition  $\{U, W\}$  and let  $M_U$  and  $N_W$  be the partition matroids for U and W respectively, such that  $(M_U, M_W) = P(G)$  (see Section 3.3). If T is a perfect matching of G and  $R_U$  is the partial representation of  $M_U$  with respect to T, then each column in  $R_U$  is parallel to a column of the identity matrix and therefore each element in  $S \setminus T$  is parallel with an element of T. It follows that  $R_U$ is the partial representation for the cycle matroid of a graph formed by adding parallel edges to a spanning tree. Since such a graph clearly has no  $K_4$  minor,  $R_U$  and  $R_W$  are partial representations of series-parallel matroids, and thus the class of series-parallel matroid pairs contains the class of bipartite graphs. In particular, Pfaffian series-parallel matroid pairs are an extension of the class of Pfaffian bipartite graphs.

#### 6.1 Fundamental graphs of series-parallel matroids

Using Theorem 6.1 we deduce some properties of the fundamental graph G(P) when P is a series-parallel matroid pair. These properties allow a simple characterization of Pfaffian signings of P with respect to totally unimodular signings of G(P).

A domino graph is a bipartite graph formed by adding one chord to a cycle of length 6, as in Figure 6.1. Graphs that do not have a domino graph as an induced minor are called domino free. If G is a directed bipartite graph with bipartition  $\{U, W\}$  then G is domino free if the underlying undirected graph for  $G[\delta(U)]$  is domino free and the underlying undirected graph for  $G[\delta(W)]$  is domino free.



Figure 6.1: domino graph

If B is a basis for  $M(K_4)$  and B corresponds to a path of length 3 in  $K_4$ , then up to permuting rows and columns, the partial representation  $R_{K_4}$  of  $M(K_4)$  with respect to B is given by

$$R_{K_4} = \begin{bmatrix} 0 & 1 & 1 \\ 1 & 0 & 1 \\ 1 & 1 & 1 \end{bmatrix}.$$

Thus  $G(M(K_4), B)$  is isomorphic to a domino graph. It follows that if the matroid M does not have an  $M(K_4)$  minor then G(M) is domino free. By Theorem 6.1, G(M) is domino free whenever M is series-parallel and thus if P is a series-parallel matroid pair, then G(P) is domino free.

Let  $W_k$  denote the wheel with k spokes (Figure 6.2). If B is the basis of  $M(W_k)$  corresponding to the spanning tree of  $W_k$  containing all the spoke edges, then up to permuting rows and columns, the partial representation  $R_{W_k}$  of  $M(W_k)$  with respect to



Figure 6.2: The wheel with 5 spokes,  $W_5$ 

B is given by

$$R_{W_k} = \begin{bmatrix} 1 & 1 & 0 & 0 & \cdots & 0 & 0 & 0 \\ 0 & 1 & 1 & 0 & \cdots & 0 & 0 & 0 \\ 0 & 0 & 1 & 1 & \cdots & 0 & 0 & 0 \\ & & & \vdots & 1 & 1 & 0 \\ 0 & 0 & 0 & 0 & \cdots & 0 & 1 & 1 \\ 1 & 0 & 0 & 0 & \cdots & 0 & 0 & 1 \end{bmatrix}$$

Thus  $G(M(W_k), B)$  is isomorphic to a circuit of length 2k. It follows that if the matroid M has an induced circuit of length 2k then M has a minor isomorphic to the cycle matroid for  $W_k$ . Since  $W_3$  is isomorphic to  $K_4$  and  $W_3$  is a minor of  $W_k$  whenever  $3 \le k$ , if M does not have a  $M(K_4)$  minor then G(M) does not have an induced circuit of length greater than 4. Thus we have the following observation:

**Observation 6.2.** If M is a series-parallel matroid then all induced circuits in G(M) have length 4 and G(M) is domino free.

We apply this observation to Camion's signing algorithm. Let M be a series-parallel matroid and let F be a maximal forest of G(M). Let  $E(G(M)) \setminus E(F) = \{e_1, \ldots, e_k\}$ where for  $i = 1, \ldots, k$  there exists an induced circuit  $C_i$  of G(M) with  $e_i \in E(C_i)$  and  $E(C_i) \subseteq E(F) \cup \{e_1, \ldots, e_i\}$ . Since M is graphic it is regular and therefore G(M) has a totally unimodular signing. All induced cycles in G(M) have length 4, and thus Camion's algorithm implies that an edge signing of G(M) is a totally unimodular signing if and only if the signing restricted to every circuit of length 4 in G(M) is totally unimodular. Since  $\begin{pmatrix} 1 & 1 \\ 1 & 1 \end{pmatrix}$  is the adjacency matrix for a circuit of length 4, a totally unimodular edge signing s of a circuit C of length 4 satisfies s(C) = 0 and thus any edge signing of G(M). Recalling that the even edge signing of a graph assigns 0 to each edge, we have the following observation:

**Observation 6.3.** If M is a series-parallel matroid and R is the partial representation of M with respect to the basis B, then [I|R] defined over the reals is totally unimodular and the even edge signing of G(M) is totally unimodular.

The edges e and f are in parallel in the graph G if and only if every spanning tree of G contains at most one edge from  $\{e, f\}$ , and thus e and f are in parallel in the graphic

matroid M(G) if and only if  $|B \cap \{e, f\}| \leq 1$  for all bases B of M(G). The edges e and f are in series in G if and only if every spanning tree of G contains at least one edge from  $\{e, f\}$ , and thus e and f are in series in M(G) if and only if  $|B \cap \{e, f\}| \geq 1$  for all bases B of M(G). More generally, given a matroid M and two elements x and y in the ground set for M, the elements x and y are said to be parallel if  $|B \cap \{x, y\}| \leq 1$  for all bases B and x and y are in series if  $|B \cap \{x, y\}| \geq 1$  for all bases B. Thus x and y are parallel in M if and only if x and y are in series in the dual  $M^*$ . If x and y are parallel in M and the basis B of M contains x, then  $B\Delta\{x, y\}$  is a basis of M. By duality, if x and y are in series in M and B is a basis of M with  $x \in B$ ,  $y \notin B$ , then  $B\Delta\{x, y\}$  is a basis of M.

Let M be a series-parallel matroid on the ground set S and let R be the partial representation of M with respect to the basis B. Since M is series-parallel, there exist  $e, f \in S$  such that either e and f are parallel or e and f are in series.

Suppose that e and f are parallel, and therefore the columns for e and f in the representation [I|R] are equal. Since two parallel elements can not be in the same basis, we may assume that  $f \notin B$ . If  $e \in B$  then e is a column in the identity matrix and therefore the column for f in R has exactly one non-zero entry. Since R is the adjacency matrix for G(M, B), it follows that the degree of f in G(M, B) is one. If e and f are parallel in M and neither e nor f is contained in B, then e and f index equal columns in R and thus e and f have the same neighbours in G(M, B). A pair of vertices with the same set of neighbours in a graph are *twin vertices*.

Suppose instead that  $e, f \in S$  are in series in M. By duality, e and f are parallel in  $M^*$  and thus one of the following holds:

- (i) e and f are twin vertices in  $G(M^*, S \setminus B)$ , or
- (ii) e or f has degree one in  $G(M^*, S \setminus B)$ .

Since  $G(M^*, S \setminus B)$  is isomorphic to G(M, B), one of the following hold:

- (i) e and f are twin vertices in G(M, B), or
- (ii) e or f has degree one in G(M, B).

We summarize these observations:

**Observation 6.4.** If M is a series-parallel matroid, then either G(M) has a pair of twin vertices or G(M) has a vertex of degree one. If B is a basis of M and u and v are twin vertices in G(M, B) then u and v are parallel if and only if  $u, v \notin B$ , and u and v are in series if and only if  $u, v \in B$ . If u is a degree one vertex in G(M, B) and  $uv \in E(G(M, B))$  then u and v are parallel if  $u \notin B$ , and u and v are in series if  $u \in B$ .

Observation 6.4 is used in the proof of the next lemma, which in turn will be used to identify contributing sets in G(P) when P is a series-parallel matroid pair.

**Lemma 6.5.** If M is a series-parallel matroid on the ground set S then  $X \subseteq S$  is contributing if and only if G(M)[X] has a unique perfect matching.

*Proof.* Let B be a basis of M. If the subgraph of G(M, B) induced by X has a unique perfect matching then G(M, B)[X] has an odd number of perfect matchings. Since M is binary,  $X\Delta B$  is a basis of M. Conversely, suppose  $X\Delta B$  is a basis of M and let G be the subgraph of G(M, B) induced by X. Since  $X\Delta B$  is a basis of M, G has at least one perfect matching and thus |X| is even. We will use induction on |X| to show that G has a unique perfect matching.

If |X| = 2 then since G has a perfect matching and G is simple, G has a unique perfect matching. Suppose |X| = 2k with k > 1 and assume that the theorem holds for smaller k. Let  $\overline{B} = S \setminus B$  and  $M' = M/(B \setminus X) \setminus (\overline{B} \setminus X)$ , and let  $B' = B \cap X$  and  $X' = X \setminus B$ . Since  $B' \cup (B \setminus X) = B$  and  $X' \cup (B \setminus X) = B \Delta X$ , both B' and X' are bases of M'. Furthermore, G = G(M', B'). Since M' is a minor of a series-parallel matroid, M' is series-parallel and by Observation 6.4 either G has a vertex of degree one or G has a pair of twin vertices.

Suppose that G has twin vertices u and v. Since V(G) has bipartition  $\{B', X'\}$ , either  $u, v \in B'$  or  $u, v \in X'$ . If  $u, v \in B'$  then u and v are in series in M' and since  $|X' \cap \{u, v\}| = 0$ , this contradicts that X' is a basis of M'. If  $u, v \in X'$  then u and v are parallel in M' and since  $|X' \cap \{u, v\}| = 2$ , this again contradicts that X' is a basis of M'. Therefore there are no twin vertices in G and G must have a vertex of degree one.

Let u be a vertex of G with degree one and let  $v \in V(G)$  be such that  $uv \in E(G)$ . Every perfect matching of G must use uv and thus G has a unique perfect matching if and only if  $G \setminus \{u, v\}$  has a unique perfect matching. Since  $G \setminus \{u, v\}$  is the subgraph of G(M, B) induced by  $X \setminus \{u, v\}$ , by the induction hypothesis it is sufficient to prove that  $(X \setminus \{u, v\}) \Delta B$  is a basis of M.

Observation 6.4 implies that either u and v are parallel in M' or u and v are in series in M'. Since X' is a basis of M' with  $|X' \cap \{u, v\}| = 1$ , it follows that  $X'\Delta\{u, v\}$  is a basis of M', and therefore  $(X'\Delta\{u, v\}) \cup (B\setminus X)$  is a basis of M. Since  $(X'\Delta\{u, v\}) \cup (B\setminus X) = X\setminus\{u, v\}\Delta B$ , the graph  $G\setminus\{u, v\}$  has a unique perfect matching by the induction hypothesis and therefore G has a unique perfect matching.  $\Box$ 

Let P = (M, N) be a series-parallel matroid pair. If D is a set of vertex disjoint directed circuits in G(P) then D induces a perfect matching  $T_M(D)$  in G(M)[V(D)] and D induces a perfect matching  $T_N(D)$  in G(N)[V(D)]. Furthermore, if D' is a set of vertex disjoint circuits in G(P) with V(D) = V(D'), then  $D \neq D'$  if and only if either  $T_M(D) \neq T_M(D')$  or  $T_N(D) \neq T_N(D')$ . It follows that if H is a subgraph of G(P) then there is a unique set of disjoint directed circuits D in G(P) with V(D) = V(H) if and only if G(M)[V(H)] and G(N)[V(H)] have unique perfect matchings. By Lemma 6.5, His contributing if and only if there is a unique set D of vertex disjoint directed circuits in G(P) with V(D) = V(H). In particular, we have the following observation. **Observation 6.6.** If P is a series-parallel matroid pair then all contributing circuits in G(P) are simple contributing circuits.

# 6.2 Contributing circuits and decomposing circuits

Using the structure of G(P) when P is a series-parallel matroid pair, we show that a totally unimodular signing of G(P) is a Pfaffian signing if and only all simple contributing circuits are correctly signed. Since the even edge signing is a totally unimodular signing of G(P) whenever P is a series-parallel matroid pair, this signing result for contributing circuits gives an easy way to recognize if a signing of G(P) is non-Pfaffian.

A decomposing circuit K in a directed graph is a circuit of length 4 such that the orientation of the edges in K alternates, as in Figure 6.3. Let C be a directed circuit



Figure 6.3: A decomposing circuit

in G and let K be a decomposing circuit in G with  $|E(K) \cap E(C)| = 2$ . The two arcs in  $E(K) \cap E(C)$  are called the C-edge arcs of K and the two arcs in  $E(K) \setminus E(C)$  are called the C-chord arcs of K. If e is a directed chord in the directed circuit C then we let  $C_e$  denote the directed circuit in  $C \cup \{e\}$  through the arc e. If e and k are the C-chord arcs of K then e and k are chords of C and the circuits  $C_e$  and  $C_k$  are vertex disjoint with  $V(C) = V(C_e) \cup V(C_k)$ . We say that K decomposes C into  $C_e$  and  $C_k$ . Decomposing circuits are used extensively in our analysis of series-parallel matroids, and the next lemma is applied frequently.

**Lemma 6.7.** If P is a series-parallel matroid pair and D is a set of vertex disjoint directed circuits in G(P), then D is contributing if and only if G(P) does not have a decomposing circuit K with  $|E(K) \cap E(D)| = 2$ .

Proof. Let P be a series-parallel matroid pair and let D be a set of vertex disjoint directed circuits in G(P). The matching  $T_M(D)$  is the unique matching of G(M)[V(D)] if and only if G(M) does not contain a  $T_M(D)$  alternating circuit, and if K is a decomposing circuit in G(P) with  $|E(K) \cap E(D)| = 2$ , then K is either a  $T_M(D)$  alternating circuit in G(M) or a  $T_N(D)$  alternating circuit in G(N). Thus if D is contributing then there is no decomposing circuit K with  $|E(K) \cap E(D)| = 2$ . Conversely, suppose D is not contributing in G(P). Without loss of generality we may assume that G(M) has a  $T_M(D)$  alternating circuit C. If  $|C| \neq 4$  then since M is series-parallel, C has a chord e, and  $C \cup \{e\}$  has a  $T_M(D)$  alternating circuit C' through e with |C'| < |C|. Therefore there is a  $T_M(D)$  alternating circuit K in G(M) with length 4 and thus K is a decomposing circuit in G(P) with  $|E(K) \cap E(D)| = 2$ .  $\Box$ 

Let K and K' be decomposing circuits in the directed bipartite graph G. If K and K' share exactly one edge, then  $K \cup K'$  is a domino, and thus if G is domino free then  $K \cup K'$  is not an induced subgraph of G. There are two edges e and e' in the subgraph induced by  $K \cup K'$  such that  $K \cup K' \cup \{e\}$  is not a domino and  $K \cup K' \cup \{e'\}$  is not a domino, and these edges are called the *joining edges* between K and K' shown (see Figure 6.4). Note that one joining edge is directed from K to K' and the other joining edge is directed from K' to K.



Figure 6.4: Two possible joining edges in a domino

Given a matroid pair P and a signing of G(P), we would like a simple way to recognize if the signing is Pfaffian. The main theorem of this section states that if P is seriesparallel and a signing of G(P) is totally unimodular, then to determine if the signing is a Pfaffian signing it is sufficient to check that the contributing circuits in G(P) are correctly signed. Recall that all contributing circuits in G(P) are simple when P is series-parallel (Lemma 6.5), and that a simple contributing circuit C in G(P) is correctly signed by the edge signing s if s(C) has parity  $\frac{|C|}{2} + 1$  (Lemma 3.3).

**Theorem 6.8.** If P is a series-parallel matroid pair and s is a totally unimodular signing of G(P), then s is a Pfaffian signing of P if and only if all contributing circuits in G(P) are correctly signed.

*Proof.* Let P = (M, N) be a series-parallel matroid pair on the ground set S and let B be a basis of P. Let  $(R_M, R_N)$  be the partial representation of P with respect to B. If s is a Pfaffian signing of G(P, B) then all contributing circuits are correctly signed, so it suffices to show that if H is a contributing subgraph of G(P, B) and all contributing circuits of G(P, B) are correctly signed, then H is correctly signed.

Let s be a totally unimodular signing of G(P, B) and let  $(R'_M, R'_N)$  be the corresponding signing of  $(R_M, R_N)$ . Let  $U \subseteq B$  and  $W \subseteq S \setminus B$  be such that  $G(P, B)[U \cup W]$  is contributing and let  $D = \{C_1, C_2, \ldots, C_n\}$  be the unique set of disjoint directed circuits in G(P, B) with  $V(D) = U \cup W$ . Let  $A_M$  and  $A_N$  be the adjacency matrices for  $T_M(D)$ and  $T_N(D)$  respectively. Since  $T_M(D)$  is the unique perfect matching in  $G(M, B)[U \cup W]$  and there is a one-to-one correspondence between the perfect matchings and terms in the determinant of  $R_M[U;W]$ , the determinant of  $R_M[U;W]$  is equal to the determinant of  $A_M$ . Thus if  $A'_M$  is the signing of  $A_M$  corresponding to s then det  $R'_M[U;W] = \det A'_M$ . Similarly, if  $A'_N$  is the signing of  $A_N$  corresponding to s then det  $R'_N[U;W] = \det A'_N$ .

Let  $U_i = U \cap V(C_i)$  and  $W_i = W \cap V(C_i)$ . By simultaneously permuting rows and columns we may assume that  $A'_M$  and  $A'_N$  are block diagonal with  $i^{th}$  block indexed by rows  $U_i$  and columns  $W_i$ . Thus

$$\det R'_{M}[U;W] = \det A'_{M} = \sum_{i=1}^{k} \det A'_{M}[U_{i};W_{i}] = \sum_{i=1}^{k} \det R'_{M}[U_{i};W_{i}]$$

and similarly,

$$\det R'_N[U;W] = \sum_{i=1}^k \det R'_N[U_i;W_i].$$

Since  $C_i$  is a contributing circuit with  $V(C_i) = U_i \cup W_i$ , if the contributing circuits in G(P, B) are correctly signed then det  $R'_M[U_i; W_i] = \det R'_N[U_i; W_i]$  for  $1 \le i \le k$ . Thus det  $R'_M[U; W] = \det R'_N[U; W]$  if all contributing circuits are correctly signed and therefore all contributing subgraphs are correctly signed whenever all contributing circuits are correctly signed.

To determine if P is a Pfaffian pair, it therefore suffices to determine if G(P) has a totally unimodular signing s such that  $s(C) = \frac{|C|}{2} + 1$  for all contributing circuits C in G(P). Recall that when P is a Pfaffian matroid pair, a Pfaffian signing of G(P) can be obtained by resigning any totally unimodular signing of G(P) across a cut in G(N). Since the even edge signing of G(P) is a totally unimodular signing, if G(P) has a Pfaffian signing then it can be obtained by resigning the even edge signing across a cut in G(N).

# 6.3 Vertex signings of the fundamental graph

We use vertex signings to define a fundamental class of non-Pfaffian series-parallel matroid pairs, and in Section 6.6 we show that the matroids in this class have a non-Pfaffian certificate with at most 32 bases. A vertex signing of a graph is a  $\{0, 1\}$  function on the vertices of the graph. If  $\alpha_C \in \{0, 1\}$  is defined for all directed circuits in the directed graph G, then a vertex signing t is a proper vertex signing with respect to  $\alpha$  if  $t(V(C)) = \alpha_C$ over GF(2) for each directed circuits C in G. Similarly, a proper edge signing of G with respect to  $\alpha$  is an edge signing s such that  $s(C) = \alpha_C$  for all directed circuits C. When P is series-parallel, we show that the problem of finding a proper edge signing of G(P) is equivalent to finding a proper vertex signing of G(P). This equivalence leads naturally to a class of non-Pfaffian series-parallel matroid pairs, and simplifies the analysis of these non-Pfaffian matroid pairs in Section 6.6. Let P = (M, N) be a series-parallel matroid pair on the ground set S, and let t be a vertex signing of G(P). Let  $X \subseteq S$  be such that  $t_v = 1$  if and only if  $v \in X$ , and let s be the totally unimodular edge signing of G(P) obtained from the even edge signing by resigning across the cut  $\delta(X)$  in G(N). Assume that C is a directed circuit in G(P). If T is the matching in G(N) induced by C and uv is an edge in T, then  $s_{uv} = t_u + t_v$ . Since V(T) = V(C) and  $s_e = 0$  unless e is an edge in G(N),

$$s(C) = s(T) = t(V(T)) = t(V(C)).$$

Thus for all vertex signings t of G(P) there is a totally unimodular edge signing s of G(P)with s(C) = t(C) for all directed circuits C in G(P). If P is a Pfaffian series-parallel matroid then G(P) has a Pfaffian signing obtained by resigning the even signing across the edges in a cut  $\delta(X)$  of G(N). Thus Theorem 6.8 implies the following observation:

**Observation 6.9.** The series-parallel matroid pair P is Pfaffian if and only if G(P) has a vertex signing t satisfying

$$t(V(C)) = \frac{|C|}{2} + 1$$

for each contributing circuit C of G(P).

Let t be a vertex signing of the directed graph G satisfying  $t(V(C)) = \frac{|C|}{2} + 1$  for all contributing circuits C in G. If  $\{C_1, C_2, \ldots, C_k\}$  is a decomposition of the directed circuit C into contributing circuits, then over GF(2),

$$t(V(C)) = \sum_{i=1}^{k} t(V(C_i))$$
  
=  $\sum_{i=1}^{k} \frac{|C_i|}{2} + 1$   
=  $\frac{|C|}{2} + k$  (6.1)

It follows that if C has a decomposition into k contributing circuits and C also has a decomposition into k' contributing circuits, then k and k' must have the same parity. If C is a directed circuit in a graph and all decompositions of C have the same parity k, then we define the *decomposing parity* of C to be k modulo 2. The decomposing parity of C is denoted dp(C) and in particular, dp(C) = 1 when C is contributing. If P is a series-parallel matroid pair and the directed circuit C in G(P) has a decomposition into an even number of contributing circuits and a decomposition into an odd number of circuits, then it follows from Equation (6.1) and Observation 6.9 that P is non-Pfaffian. A circuit with a decomposition into both an even number of contributing circuits and an odd number of contributing circuits is called a *neutral* circuit.

If C is a directed circuit in the graph G and C is not neutral, then we say that an edge signing s of G respects the decomposing parity of C if  $s(C) = \frac{|C|}{2} + dp(C)$ . Similarly, a vertex signing t of G respects the decomposing parity of C if  $t(V(C)) = \frac{|C|}{2} + dp(C)$ . If a signing respects the decomposing parity of all directed circuits in G, then the signing is said to respect the decomposing parity of G. We therefore have an alternative form of Theorem 6.8:

**Theorem 6.10.** If P is a series-parallel matroid pair and G(P) does not have any neutral circuits, then P is Pfaffian if and only if G(P) has a totally unimodular edge signing that respects the decomposing parity of all directed circuits in G(P).

In the next section we prove that the totally unimodular requirement can be omitted from Theorem 6.10.

# 6.4 Proper edge signings of the fundamental graph

The obstructions to a graph having a proper edge signings have been well characterized by Gerards [15], as we will see in Section 6.7. When P is a series-parallel matroid we would like to use this characterization together with Theorem 6.10 to get some structural results about G(P) whenever G(P) does not have an edge signing that respects the decomposing parity of G(P). However, if G(P) has an edge signing that respects the decomposing parity of G(P), such an edge signing need not be a Pfaffian signing: if  $s(C) = \frac{|C|}{2} + dp(C)$  for all directed circuits C but s is not a totally unimodular signing of G(P, B), then from the results so far we can not conclude that P is Pfaffian. In this section we show that if P is connected and G(P) does not have any neutral circuits, then determining if P is Pfaffian is equivalent to determining if G(P) has an edge signing which respects the decomposing is parity of the directed circuits in G(P). The main theorem of this section is the following:

**Theorem 6.11.** If P is a connected series-parallel matroid and G(P) has no neutral circuits, then P is Pfaffian if and only if there is an edge signing of G(P) that respects the decomposing parity of G(P).

Let s be an edge signing of G and let  $E' \subseteq E(G)$  be a cut of G. The edge signing s' defined by

$$s'_e = \begin{cases} s_e + 1, & \text{if } e \in E'; \\ s_e, & \text{otherwise} \end{cases}$$

is said to be obtained by resigning s across E'. Since all circuits in G intersect E' an even number of times, s'(C) = s(C) for all directed circuits C in G. Thus if  $\alpha$  is a circuit signing for G, then s' is a proper edge signing of G with respect to  $\alpha$  if and only if s is a proper edge signing with respect to  $\alpha$ . If the circuit signing  $\alpha$  of G satisfies  $\alpha_C = \alpha_{C_1} + \alpha_{C_2}$  whenever a directed circuit C has a decomposition into  $C_1$  and  $C_2$ , then we say  $\alpha$  is a good circuit signing. Clearly  $\alpha$  must be a good circuit signing if G has a proper vertex signing with respect to  $\alpha$ . The proof of Theorem 6.11 requires the following two lemmas about decomposing circuits in a proper edge signing of a directed graph.

**Lemma 6.12.** Let K be a decomposing circuit in the directed graph G and assume that  $ab, cd \in E(K)$  are such that G has a directed path from b to a that is disjoint from a directed path from d to c. If  $\alpha$  is a good circuit signing of G and s is a proper edge signing of G with respect to  $\alpha$ , then s(K) = 0.

*Proof.* Assume that  $\alpha$  is a good circuit signing of G and let s be a proper edge signing of G with respect to  $\alpha$ . By resigning across cuts in G we may assume that s(ab) = s(ad) = s(cb) = 0. Let  $P_{ba}$  be a directed path in G from b to a and let  $P_{dc}$  be a directed path in G from d to c and assume  $P_{ba}$  and  $P_{dc}$  are vertex disjoint, as shown in Figure 6.5. If C



Figure 6.5: The decomposing circuit K for Lemma 6.12

is the directed circuit in  $P_{ba} \cup P_{dc} \cup \{ad, cb\}$ , then

$$s(C) = s(P_{ba}) + s(P_{dc}) + s(ad) + s(cb) = s(P_{ba}) + s(P_{dc}).$$
(6.2)

Furthermore, K decomposes C into  $C_{ab}$  and  $C_{cd}$ , and since  $\alpha$  is a good circuit signing of G and s is a proper edge signing with respect to  $\alpha$ ,

$$s(C) = s(C_{ab}) + s(C_{cd})$$
  
= (s(ab) + s(P\_{ba})) + (s(cd) + s(P\_{dc}))  
= s(P\_{ab}) + s(P\_{dc}) + s(cd). (6.3)

Equations (6.2) and (6.3) imply that s(cd) = 0 and thus s(K) = 0.

**Lemma 6.13.** Let G be a bipartite, domino free, strongly connected bipartite graph, and let  $\alpha$  be a good circuit signing of G. If s is a proper edge signing of G with respect to  $\alpha$ and  $\alpha_C = \frac{|C|}{2} + 1$  whenever C is a contributing circuit, then s(K) = 0 for all decomposing circuits K in G. Proof. Let s be a proper edge signing of G with respect to  $\alpha$ . We will say that  $E' \subseteq E(G)$  is even if s(E') = 0 and odd if s(E') = 1. Assume that G has an odd decomposing circuit K with  $V(K) = \{a, b, c, d\}$  and  $ab, cd \in E(K)$ . Since G is strongly connected, there is a directed path  $P_{ba}$  from b to a and a directed path  $P_{dc}$  from d to c. in G. Over all odd decomposing circuits in G, choose K,  $P_{ba}$ , and  $P_{dc}$  to minimize  $|P_{ba}| + |P_{dc}|$ . By resigning s across cuts in G, we may assume that s(ab) = s(ad) = s(cb) = 0. Since K is odd, s(cd) = 1.

Lemma 6.12 implies that  $P_{ba}$  and  $P_{dc}$  are not disjoint. Let  $x, y \in V(P_{ba}) \cap V(P_{dc})$  be such that the distance from b to x along  $P_{ba}$  is minimum and the distance from y to aalong  $P_{ba}$  is minimum. Suppose  $x \neq y$  and y precedes x along  $P_{dc}$ . The choice of x and y implies that the subpaths of  $P_{ba}$  from b to x and from y to a are edge disjoint from the subpaths of  $P_{dc}$  from d to y and from x to c. (See Figure 6.6, where the bold lines indicate edges, the thin lines indicate paths, and only the relevant subpaths of  $P_{ba}$  and  $P_{dc}$  are shown.) Thus if  $P_{da}$  is the directed path formed by the union of the subpath of



Figure 6.6: y precedes x along  $P_{dc}$ 

 $P_{dc}$  from d to y with the subpath of  $P_{ba}$  from y to a, and  $P_{bc}$  is the directed path formed from the union of the subpath of  $P_{ba}$  from b to x with the subpath of  $P_{dc}$  from x to c, then  $P_{da}$  and  $P_{bc}$  are vertex disjoint. This contradicts Lemma 6.12, and therefore either y = x or x precedes y along  $P_{dc}$ . By the minimality of  $|P_{ba}| + |P_{dc}|$ , we may assume that the subpath of  $P_{ba}$  from x to y is equal to the subpath of  $P_{dc}$  from x to y, as shown in Figure 6.7. The minimality of  $|P_{ba}| + |P_{dc}|$  implies that cd is not an edge in  $P_{ba}$ , but x need not be distinct from b or d, and y need not be distinct from a or c.

For each edge e in K, let C(e) be the unique directed circuit in  $K \cup P_{ba} \cup P_{dc}$  with  $e \in C(e)$ . Each edge in  $P_{ba}\Delta P_{dc}$  is in exactly two such circuits, each edge in  $P_{ba} \cap P_{dc}$  is in all four such circuits, and each edge in K is in exactly one circuit. It follows that over GF(2),

$$\sum_{e \in K} s(C(e)) = s(K) = 1,$$
(6.4)

and

$$\sum_{e \in K} |C(e)| = 2(|P_{ba}| + |P_{dc}|) + |K|.$$
(6.5)



Figure 6.7: x precedes y along  $P_{dc}$ 

Since a and b are in opposite sides of the vertex bipartition of G, both  $|P_{ba}|$  and  $|P_{dc}|$  are odd. Equation (6.5) therefore implies that

$$\sum_{e \in K} |C(e)| \equiv 0 \pmod{4}.$$
(6.6)

If C(e) is contributing for all  $e \in K$  then

$$\sum_{e \in K} s(C(e)) = \sum_{e \in K} (\frac{|C(e)|}{2} + 1)$$

and thus by (6.6),  $\sum_{e \in K} s(C(e))$  is even. This contradicts (6.4), and thus without loss of generality we may assume that C(ab) has a decomposing circuit  $K_1$ . Let  $V(K_1) =$  $\{a_1, b_1, c_1, d_1\}$  and label the vertices of  $K_1$  such that  $a_1b_1$  and  $c_1d_1$  are the C(ab)-edge arcs of  $K_1$  and  $b_1$  precedes  $d_1$  along  $P_{ba}$ . If  $a_1b_1 \neq ab$ , then  $a_1d_1$  is a shortcut arc of  $P_{ba}$ , which contradicts the minimality of  $|P_{ba}|$ . Thus  $a_1b_1 = ab$ . Note that since cd is not an edge in  $P_{ba}$ , either  $d_1 \neq d$  or  $c_1 \neq c$ .

If  $d_1 = d$  then  $s(ad_1) = s(ad) = 0$ , and if  $d_1 \neq d$  then  $d_1 \notin V(K)$  and resigning s across  $\delta(d_1)$  does not affect the sign of any edges in K. Thus we may assume  $s(ad_1) = 0$ . Similarly, either  $c_1 = c$  and  $s(c_1b) = 0$ , or we may resign s across  $\delta(c_1)$  without affecting s on either the edges of K or on  $s(ad_1)$  and thus we may assume  $s(c_1b) = 0$ . Let  $P_{bc_1}$  be the subpath of  $P_{ba}$  from b to  $c_1$  and let  $P_{d_1a}$  be the subpath of  $P_{ba}$  from  $d_1$  to a. Then  $P_{bc_1}$  is a directed  $b_1, c_1$  path and  $P_{d_1a}$  is a directed  $d_1, a_1$  path, and since  $P_{b_1c_1}$  and  $P_{d_1a_1}$  are disjoint, Lemma 6.12 implies that  $s(K_1) = 0$ . Thus  $s(c_1d_1) = 0$ , as shown in Figure 6.8(a).

If  $d_1 = d$  then there is an odd decomposing cycle  $K_2$  with  $cd, c_1b \in E(K_2)$ . Since  $P_{dc}$ is disjoint from  $P_{bc_1}$ , the cycle  $K_2$  contradicts Lemma 6.12. Similarly, if  $c_1 = c$  then there is an odd decomposing cycle  $K_2$  with  $cd, ad_1 \in E(K_2)$ . The path  $P_{dc}$  is disjoint from  $P_{d_1a}$ , and thus  $K_2$  again contradicts Lemma 6.12. Hence  $d_1 \neq d$  and  $c_1 \neq c$ , and therefore K and  $K_1$  share exactly one edge, ab. Since G is domino free, either  $cd_1 \in E(G)$ , or  $c_1d \in E(G)$ . By the symmetry from reversing the orientation of G we may assume that  $cd_1 \in E(G)$ . By considering where  $d_1$  lies on  $P_{ba}$  relative to y, we show that  $s(cd_1) = 0$ .



Figure 6.8: The decomposing circuit  $K_1$  of C(ab)

If  $d_1 = y$  or  $d_1$  precedes y along  $P_{ba}$ , then let  $K_2$  be the decomposing circuit with edges  $cd_1$  and  $c_1b$ , as shown in bold in Figure 6.9(a). Let  $P_{d_1c}$  be the subpath of  $P_{dc}$  from  $d_1$  to c. Since  $P_{bc_1}$  and  $P_{d_1c}$  are disjoint, Lemma 6.12 implies that  $K_2$  is even and therefore  $s(cd_1) = 0$ . If y precedes  $d_1$  along  $P_{ba}$ , then let  $K_2$  be the decomposing circuit with edges  $cd_1$  and ab, as shown in bold in Figure 6.9(b). Let  $P_{bc}$  be the directed path formed from



Figure 6.9: y relative to  $d_1$  on  $P_{ba}$ 

the union of the subpath of  $P_{ba}$  from b to y with the subpath of  $P_{dc}$  from y to c. Since  $P_{bc}$ and  $P_{d_1a}$  are disjoint, Lemma 6.12 implies that  $K_2$  is even and therefore  $s(cd_1)$  is again 0. Thus if  $cd_1 \in E(G)$ , then  $s(cd_1) = 0$ . However there is then an odd decomposing circuit in G with edges  $ad_1$  and cd, and  $P_{d_1a}$  is a strict subpath of  $P_{ba}$ , contradicting the minimality of  $|P_{ba}| + |P_{dc}|$  in our choice of K. Thus  $cd_1 \notin E(G)$  and since this edge is necessary for G to be domino free, it follows that s(K) = 0 for all decomposing circuits in G.

We can now prove the main theorem of this section.

Proof of Theorem 6.11. Let P be a connected series-parallel matroid pair and suppose

that G(P) does not have any neutral circuits. If the edge signing s is a Pfaffian signing of G(P), then by Theorem 6.10, s respects the decomposing parity of all directed circuits in G(P). Conversely, let  $\alpha$  be the circuit signing of G(P) defined by  $\alpha_C = \frac{|C|}{2} + \operatorname{dp}(C)$  for all directed circuits C in G(P), and suppose s is a proper signing of G(P) with respect to  $\alpha$ . If C decomposes into  $C_1$  and  $C_2$ , then  $|C| = |C_1| + |C_2|$  and since G(P) does not have any neutral circuits,  $\operatorname{dp}(C) = \operatorname{dp}(C_1) + \operatorname{dp}(C_2)$ . It follows that  $\alpha_C = \alpha_{C_1} + \alpha_{C_2}$ and therefore  $\alpha$  is a good circuit signing of G(P). By Lemma 6.13, s(K) = 0 for all decomposing circuits K. Thus s is a totally unimodular edge signing of G(P) which respects the decomposing parity of G(P), and Theorem 6.8 implies that s is a Pfaffian signing.

If G(P) does not have any neutral circuits, then by Theorem 6.11 determining if P is Pfaffian is equivalent to determining if G(P) has an edge signing s satisfying  $s(C) = \frac{|C|}{2} + \operatorname{dp}(C)$  for all directed circuits C in G(P). We use this result in Section 6.7 to obtain a partial description of G(P) when P is non-Pfaffian and G(P) has no neutral circuits. In Section 6.6 we show that if G(P) has a neutral circuit, then there exists a set of at most 28 bases which certifies that P is non-Pfaffian.

## 6.5 Accordions

In this section we describe a decomposition of a directed graph into contributing circuits. The decomposition, which we call an *accordion*, is used in Sections 6.6 and 6.7 as a method for combining disjoint contributing circuits into a larger contributing subgraph.

Let G be a directed graph and let R and Q be vertex disjoint directed paths in G. A bridge circuit between R and Q is a decomposing circuit in G that contains one edge of R and one edge of Q. Suppose  $K_1$  and  $K_2$  are bridge circuits between R and Q and  $E(K_1) \cap E(R)$  precedes  $E(K_2) \cap E(R)$  along R. If  $E(K_2) \cap E(Q)$  precedes  $E(K_1) \cap E(Q)$ along Q then  $K_1$  and  $K_2$  are parallel (Figure 6.10(a)), and  $K_1$  and  $K_2$  are strongly crossing if  $E(K_1) \cap E(Q)$  precedes  $E(K_2) \cap E(Q)$  along Q (Figure 6.10(b)). When  $K_1$  and  $K_2$  are



Figure 6.10: Parallel and strongly crossing bridge circuits

parallel then the directed circuit in the subgraph  $R \cup Q \cup K_1 \cup K_2$  is called an *accordion* circuit with ends  $K_1$  and  $K_2$  (Figure 6.11(a)).

Let e be an arc with both endpoints in a directed path R with e not in R. If the tail of e precedes its head, then e is a *shortcut* arc in R, and e is a *backwards* arc otherwise. When  $K_1$  and  $K_2$  share an edge of R and there is a joining edge g between  $K_1$  and  $K_2$ that is a backwards arc of Q, then the directed circuit in  $Q \cup g$  is also called an accordion circuit with ends  $K_1$  and  $K_2$  (Figure 6.11(b)).



Figure 6.11: Accordion circuits

Let  $K_1, K_2, \ldots, K_m$  be distinct bridge circuits between R and Q and for  $1 \leq i \leq m$ let  $e_i = E(K_i) \cap E(R)$  and  $f_i = E(K_i) \cap E(Q)$ . Suppose that either  $e_i$  precedes  $e_{i+1}$ along R or  $e_i = e_{i+1}$ , and that either  $f_{i+1}$  precedes  $f_i$  along Q or  $f_i = f_{i+1}$ . If there is an accordion circuit  $C_i$  with ends  $K_i$  and  $K_{i+1}$  for all  $1 \leq i \leq m-1$ , then the union of R,  $Q, K_1, K_2, \ldots, K_m$  and  $C_1, C_2, \ldots, C_{m-1}$  is called an *accordion* between R and Q. The circuits  $K_1, K_2, \ldots, K_m$  are the bridge circuits of the accordion and the bridge circuits  $K_1$  and  $K_m$  are the ends of the accordion. Figure 6.12 is an example of an accordion with 8 bridge circuits and seven accordion circuits. If there is at most one bridge circuit between R and Q then the paths R and Q together with possibly one bridge circuit are called a *trivial accordion*. If  $\mathcal{H}$  is an accordion and all the accordion circuits in  $\mathcal{H}$  are



Figure 6.12: An accordion

contributing, then  $\mathcal{H}$  is a *contributing accordion*.

The two lemmas in this section describe when accordion circuits in a contributing accordion  $\mathcal{H}$  can be combined in a trivial way into contributing subgraphs of  $\mathcal{H}$ . These lemmas will be used in Section 6.6 to find a small set of bases that certify that a matroid pair is non-Pfaffian. If  $\mathcal{H}$  is a contributing accordion between the directed paths R and

Q then there are three types of contributing circuits in  $\mathcal{H}$ , as seen in Figure 6.12: those that intersect only V(R), those that intersect only V(Q), and those that intersect both V(R) and V(Q). We first consider the contributing circuits that intersect only V(R). A shortcut arc e in R is an *internal* shortcut arc if both ends of e are internal vertices in R.

**Lemma 6.14.** If  $\mathcal{H}$  is a contributing accordion between the directed paths R and Q in the directed graph G and R does not have any internal shortcut arcs, then the set of accordion circuits C in  $\mathcal{H}$  with  $V(C) \subseteq V(R)$  can be partitioned into two contributing subgraphs.

*Proof.* Let  $\mathcal{C}$  be the set of accordion circuits C in  $\mathcal{H}$  with  $V(C) \subseteq V(R)$  and let  $\{\mathcal{C}_A, \mathcal{C}_B\}$  be a partition of  $\mathcal{C}$  such that if  $C_A, C_B \in \mathcal{C}$  and an edge in R is incident to a vertex in  $C_A$  and a vertex in  $C_B$  then  $C_A \in \mathcal{C}_A$  if and only if  $C_B \in \mathcal{C}_B$ .

Let  $C_1$  and  $C_2$  be circuits in  $C_A$  and suppose  $C_1 \cup C_2$  is not a contributing cycle of G(P, B). Then there exists a decomposing circuit K in G with E(K) equal to  $\{u_1v_1, u_1v_2, u_2v_1, u_2v_2\}$  where  $u_1v_1 \in E(C_1)$  and  $u_2v_2 \in E(C_2)$ , and we may assume that  $u_1$  and  $v_1$  precede  $u_2$  and  $v_2$  along R. Since  $C_1$  and  $C_2$  are circuits in C, the vertices  $u_1, v_1, u_2$ , and  $v_2$  are all in R. Furthermore, since  $C_1, C_2 \in C_A$ , neither  $u_1v_2$  nor  $u_2v_1$  is an edge of R. Since  $u_1v_2$  and  $u_2v_1$  have opposite orientation with respect to R, one of these chords is an internal shortcut arc of R, contradicting the assumption that no such chord exists. Thus  $C_1 \cup C_2$  is contributing for any  $C_1, C_2 \in C_A$ . The union of all circuits in  $C_A$  is therefore contributing, as is the union of all circuits in  $C_B$ , and thus C is the disjoint union of two contributing subgraphs.

A bridge circuit between the directed paths R and Q that strongly crosses two parallel bridge circuits is a *twisted* bridge circuit. A twisted bridge circuit that strongly crosses the parallel bridge circuits  $K_1$  and  $K_2$  is *internally twisted* if  $K_1$  and  $K_2$  are edge disjoint from the ends of  $\mathcal{H}$ . The last lemma of this section shows how to combine contributing circuits that intersect both V(R) and V(Q) into a contributing set.

**Lemma 6.15.** Let  $\mathcal{H}$  be a contributing accordion between R and Q in the directed graph G where G is domino free and bipartite. If R and Q do not have any shortcut arcs and there are no internally twisted bridge circuits between R and Q, then the set of accordion circuits in  $\mathcal{H}$  which intersect both V(R) and V(Q) can be partitioned into two contributing subgraphs.

*Proof.* Let  $\mathcal{C}$  be the set of accordion circuits C in  $\mathcal{H}$  that intersect both V(R) and V(Q)and let  $\{\mathcal{C}_A, \mathcal{C}_B\}$  be a partition of  $\mathcal{C}$  such that if  $C_A, C_B \in \mathcal{C}$  and an edge in R or Qintersects both  $V(C_A)$  and  $V(C_B)$  then  $C_A \in \mathcal{C}_A$  if and only if  $C_B \in \mathcal{C}_B$ . We claim that  $\mathcal{C}_A$  and  $\mathcal{C}_B$  are contributing.

Let  $C_1 \in \mathcal{C}_A$  have ends  $L_1$  and  $R_1$  and let  $C_2 \in \mathcal{C}_A$  have ends  $L_2$  and  $R_2$ . Then  $L_1, R_1, L_2, R_2$  are parallel and distinct and we may assume that  $E(L_i) \cap E(R)$  precedes

 $E(R_i) \cap E(R)$  along R for i = 1, 2 and that  $E(R_1) \cap E(R)$  precedes  $E(L_2) \cap E(R)$  along R. Let  $l_1$  be the edge in  $L_1$  directed from Q to R and for i = 1, 2, let  $r_i$  be the edge



Figure 6.13: Contributing accordion circuits  $C_1$  and  $C_2$  from  $\mathcal{C}_A$ 

in  $R_i$  directed from R to Q, as shown in Figure 6.13. If  $C_1 \cup C_2$  is not contributing in G then there exists a decomposing circuit K in G with  $V(K) = \{u_1, v_1, u_2, v_2\}$  and  $u_i v_i = E(K) \cap E(C_i)$  for i = 1, 2.

First consider the case that  $u_1v_1$  is an edge in R. Since R does not have any shortcut arcs and  $u_1v_2 \in E(G)$ ,  $v_2$  must be in V(Q) and thus either  $u_2v_2 = r_2$  or  $u_2v_2 \in E(Q)$ . If  $u_2v_2 \in E(Q)$  then K crosses both  $R_1$  and  $L_2$  and is therefore a twisted bridge circuit, contradicting our assumption that no such circuits exists in  $\mathcal{H}$ . If  $u_1v_1 \in E(R)$  and  $u_2v_2 = r_2$  then K and  $R_2$  share exactly one edge, namely  $r_2$ . Both possible joining edges between K and  $R_2$  lead to a contradiction: if a joining edge between K and  $R_2$  is incident to  $u_1$  then R has a shortcut arc, and if a joining edge g between K and  $R_2$  is incident to  $v_1$  then the bridge circuit through g in  $K \cup R_2 \cup \{g\}$  crosses both  $R_1$  and  $L_2$ . It follows that  $u_1v_1$  is not an edge of R.

Suppose instead that  $u_1v_1 = r_1$ . If  $v_2 \in V(R)$  then  $u_1v_2$  is a shortcut arc of R and if  $u_2 \in V(Q)$  then  $u_2v_1$  is shortcut arc of Q. Thus  $u_1v_1 = r_1$  implies  $u_2v_2 = r_2$  and Kand  $R_2$  share the edge  $r_2$ . The joining edge between K and  $R_2$  is either a shortcut arc of R or a shortcut arc of Q, and thus  $u_1v_1 \neq r_1$ . We may therefore assume that either  $u_1v_1 \in E(Q)$  or  $u_1v_1 = l_1$ . By symmetry, we may assume that either  $u_2v_2 \in E(R)$  or  $u_2v_2 = r_2$ .

If  $u_1v_1 \in E(Q)$  and  $u_2v_2 \in E(R)$  then K strongly crosses  $R_2$  and  $L_2$ , and thus either  $u_1v_1 = l_1$  or  $u_2v_2 = r_2$ . If  $u_1v_1 = l_1$  and  $u_2v_2 \in E(R)$  then K and  $L_1$  share the edge  $l_1$ . The joining edge between K and  $L_1$  that is directed from  $L_1$  to K is a shortcut arc of R. If the joining edge g between K and  $L_1$  is directed from K to  $L_1$  then the bridge circuit containing g in  $L_1 \cup K \cup \{g\}$  strongly crosses  $R_1$  and  $L_2$  and thus contradicts the assumption that there are no twisted bridges. Thus if  $u_1v_1 = l_1$  then  $u_2v_2 \notin E(R)$  and by symmetry  $u_1v_1 \notin E(Q)$  when  $u_2v_2 = r_2$ . More generally, there is no decomposing circuit through  $l_1$  and an edge e of R where e is preceded along R by the edge of  $L_2$  in R. The only case that remains is  $u_1v_1 = l_1$  and  $u_2v_2 = r_2$ .

If  $u_1v_1 = l_1$  and  $u_2v_2 = r_2$  then K and  $R_2$  share the edge  $r_2$ . If the joining edge g between K and  $R_2$  is directed from K to  $R_2$ , then there is a decomposing circuit in  $K \cup R_2 \cup \{g\}$  that contains  $l_1$  and an edge e of R such that  $L_2$  precedes e along R. Thus

the joining edge g between K and  $R_2$  must be directed from  $R_2$  to K. By symmetry the joining edge g' between K and  $L_1$  is directed from  $L_1$  to K. This is a contradiction since there is then a bridge circuit through g and g' that strongly crosses  $R_1$  and  $L_2$ .

Thus  $C_1 \cup C_2$  is contributing for all circuits  $C_1, C_2 \in \mathcal{C}_A$  and  $\mathcal{C}_A$  is therefore a contributing set, as is the set of circuits in  $\mathcal{C}_B$ . Since  $\{\mathcal{C}_A, \mathcal{C}_B\}$  is a partition of  $\mathcal{C}$ , the lemma follows.

## 6.6 Neutral circuits and a certificate of constant size

Neutral circuits are directed circuits which have a decomposition into an even number of contributing circuits and a decomposition into an odd number of circuits. Section 6.3 showed that if P is a series-parallel matroid pair and G(P) has a neutral circuit, then P is non-Pfaffian. In this section we prove that if a series-parallel matroid pair has a neutral circuit, then it has an non-Pfaffian bases certificate of at most 32 bases which certifies that the matroid pair is non-Pfaffian. This supports Conjecture 3.12 that there exists a constant c such that every non-Pfaffian matroid pair has a non-Pfaffian bases certificate of size at most c.

Let P be a minimally non-Pfaffian series-parallel matroid pair on the ground set S and let C be a neutral circuit in G(P). The minor of P corresponding to the minor G(P)[V(C)] of G(P) is non-Pfaffian, and the assumption that P is minimally non-Pfaffian therefore implies that V(C) = S. Hence all neutral circuits of G(P) are Hamiltonian. If C' is a non-Hamiltonian directed circuit in G(P) then it follows by minimality that the decomposing parity of C' is well defined. For a non-Hamiltonian circuit C' in G(P), define sign(C') by

$$\operatorname{sign}(C') \equiv \frac{|C|}{2} + \operatorname{dp}(C') \pmod{2}.$$

It follows from Theorem 6.10 that if s is a Pfaffian signing of the minor of P corresponding to G(P)[V(C')] then  $s(C') = \operatorname{sign}(C')$ . If the directed circuit C is neutral in G and the decomposing circuit K decomposes C into  $C_f$  and  $C_k$ , then let  $\operatorname{sign}(C, K) = \operatorname{sign}(C_f) + \operatorname{sign}(C_k)$ . Note that  $C_f$  and  $C_k$  are not Hamiltonian and thus  $\operatorname{sign}(C, K)$  is well defined. For all neutral circuits C in G(P) there exist decomposing circuits K and K' of C such that  $\operatorname{sign}(C, K) \neq \operatorname{sign}(C, K')$ . In particular, if  $\{k, h\}$  are the C-chord edges of K and  $\{k', h'\}$  are the C-chord edges of K', then

$$\operatorname{sign}(C_k) + \operatorname{sign}(C_h) + \operatorname{sign}(C_{k'}) + \operatorname{sign}(C_{h'}) = 1.$$

Furthermore,  $\{C_k, C_h, C_{k'}, C_{h'}\}$  is an even cover of S, and thus a decomposition of  $\{C_k, C_h, C_{k'}, C_{h'}\}$  into contributing circuits leads to a non-Pfaffian basis certificate for P. Our approach to finding a certificate of bounded size for P is therefore to show that

 $C_k$ ,  $C_h$ ,  $C_{k'}$ , and  $C_{h'}$  each partition into a bounded number of contributing sets. We first require some definitions.

Let f and h be vertex disjoint chords of the directed circuit C. If exactly one end of f is contained in  $C_h$ , then f and h cross. If K is a decomposing circuit of C and h is a chord of C, then K and h cross if h crosses a C-chord edge of K. If h crosses both C-chord edges in K then h strongly crosses K. If K and K' are distinct decomposing circuits of C and a C-chord edge of K crosses K', then we say that K and K' cross. Two decomposing circuits of C that share an edge are weakly crossing, and crossing decomposing circuits which are not weakly crossing are *strongly crossing*. A pair of decomposing circuits of C are *parallel* if they do not cross. (See Figure 6.14.)



Figure 6.14: Pairs of decomposing circuits

If D is a set of directed circuits in the directed graph G and each vertex in G is in an even number of circuits of D, then D is an even cover of G. An even cover D of G is spanning if V(D) = V(G) and D and is non-spanning otherwise.

Let D be an even cover of G(P). If the sum of  $\operatorname{sign}(C')$  over all directed circuits C' in D is odd, then G(P)[V(D)] does not have a signing which respects the decomposing parity of G(P)[V(D)] and thus the minor of P corresponding to G(P)[V(D)] is non-Pfaffian. The assumption that P is minimally non-Pfaffian therefore implies that D is spanning. Our proof that a neutral circuit has a non-Pfaffian bases certificate containing a bounded number of bases which certifies that the matroid pair is non-Pfaffian uses this observation together with three lemmas relating  $\operatorname{sign}(C, K)$  and  $\operatorname{sign}(C, K')$  for decomposing circuits K and K' in the neutral circuit C in G(P). The first of the three lemmas considers  $\operatorname{sign}(C, K)$  and  $\operatorname{sign}(C, K')$  when K and K' are parallel.

**Lemma 6.16.** Let P be a minimally non-Pfaffian matroid pair and let C be a neutral circuit in G(P). If K and K' are parallel decomposing circuits of C, then  $\operatorname{sign}(C, K) = \operatorname{sign}(C, K')$ .

*Proof.* Let the C-chord edges of K and K' be  $\{f, h\}$  and  $\{f', h'\}$  respectively, with  $C_f$  and  $C_{f'}$  vertex disjoint as shown in Figure 6.6(a). If  $C' = (C_h)'_h$  as shown in Figure 6.6(b), then K' decomposes  $C_h$  into C' and  $C_{f'}$ , and K decomposes  $C_{h'}$  into C' and  $C_f$ . Since



Figure 6.15: Parallel decomposing circuits of C

*P* is minimally non-Pfaffian,  $\operatorname{sign}(C_{h'}) = \operatorname{sign}(C') + \operatorname{sign}(C_f)$  and  $\operatorname{sign}(C_h) = \operatorname{sign}(C') + \operatorname{sign}(C_{f'})$ , and thus

$$sign(C, K) = sign(C_f) + sign(C_h)$$
  
= sign(C\_f) + (sign(C\_{f'}) + sign(C'))  
= sign(C\_f) + sign(C\_{f'}) + (sign(C\_{h'}) + sign(C\_f))  
= sign(C\_{f'}) + sign(C\_{h'})  
= sign(C, K').

The second of the three lemmas considers  $\operatorname{sign}(C, K)$  and  $\operatorname{sign}(C, K')$  when the decomposing circuits K and K' of C are weakly crossing.

**Lemma 6.17.** Let P be a minimally non-Pfaffian matroid pair and let C be a neutral circuit in G(P). If K and K' are weakly crossing decomposing circuits of C with  $e = E(K) \cap E(K')$ , then  $\operatorname{sign}(C, K) = \operatorname{sign}(C, K')$  if either of the following hold:

- (i) A joining edge g between K and K' is a backwards arc of  $C \setminus \{e\}$ .
- (ii) The C-chord edge f of K is such that  $C_f$  is disjoint from K' and the path  $C_f \setminus \{f\}$  has a shortcut arc.

*Proof.* Let  $E(K) = \{e, k, f, h\}$  and let  $E(K') = \{e, k', f', h'\}$  where the edges of K and K' are labeled as in Figure 6.16(a).

(i) Assume a joining edge g between K and K' is a backwards arc of  $C \setminus \{e\}$ , as in Figure 6.16(b). Then  $C_h$  decomposes into  $C_{f'}$  and  $C_g$ , and  $C_{h'}$  decomposes into  $C_f$  and  $C_q$ . Thus

$$\operatorname{sign}(C_h) = \operatorname{sign}(C_{f'}) + \operatorname{sign}(C_g), \quad \text{and} \\ \operatorname{sign}(C_{h'}) = \operatorname{sign}(C_f) + \operatorname{sign}(C_g),$$



Figure 6.16: Weakly crossing decomposing circuits K and K' of C

and therefore

$$\operatorname{sign}(C, K) = \operatorname{sign}(C_f) + \operatorname{sign}(C_h)$$
$$= \operatorname{sign}(C_{f'}) + \operatorname{sign}(C_{h'})$$
$$= \operatorname{sign}(C, K').$$

(ii) Let w be a shortcut arc of  $C_f \setminus \{f\}$ . If  $D_1$  and  $D_2$  are given by

 $D_1 = \{(C_f)_w, C_f, (C_{h'})_w, C_{h'}\} \text{ and } D_2 = \{(C_f)_w, (C_{h'})_w, C_h, C_{h'}\},$ 

then  $D_1$  and  $D_2$  are both non-spanning even cycle covers of G(P), as shown in Figure 6.6(a) and (b) respectively. Since P is minimally non-Pfaffian, sign $(D_1) =$ 



Figure 6.17: Non-spanning even cycle covers of G(P)

 $sign(D_2) = 0$  and thus

$$\operatorname{sign}(C, K) + \operatorname{sign}(C, K') = (\operatorname{sign}(C_f) + \operatorname{sign}(C_h)) + (\operatorname{sign}(C_{f'}) + \operatorname{sign}(C_{h'}))$$
$$= \operatorname{sign}(D_1) + \operatorname{sign}(D_2)$$
$$= 0.$$

By reversing the orientation of the edges of G, it follows that if  $C_{f'} \setminus \{f'\}$  has a shortcut arc then  $\operatorname{sign}(C, K) = \operatorname{sign}(C, K')$ .

The third and final lemma relating  $\operatorname{sign}(C, K)$  and  $\operatorname{sign}(C, K')$  considers the case that K and K' are strongly crossing.

**Lemma 6.18.** Let P be a minimally non-Pfaffian matroid pair and let C be a neutral circuit in G(P). If K and K' are strongly crossing decomposing circuits of C, then  $\operatorname{sign}(C, K) = \operatorname{sign}(C, K')$  if either of the following hold:

- (i) A directed path in C that is edge disjoint from K and K' has a shortcut arc.
- (ii) A chord of C strongly crosses K and K'.

*Proof.* Assume  $\{f, h\}$  are the *C*-chord edges of *K* and  $\{f', h'\}$  are the *C*-chord edges of *K'*, where the edges of *K* and *K'* are labeled as shown in Figure 6.18.



Figure 6.18: Strongly crossing decomposing circuits K and K'

(i) Let Q be a directed path in C that is edge disjoint from E(K) and E(K') and assume that Q has a shortcut arc w. Without loss of generality we may assume that Q is directed from the head of h' to the tail of f. Then  $D_1 = \{(C_f)_w, C_h, (C_{h'})_w, C_{f'}\}$  and  $D_2 = \{(C_f)_w, C_f, (C_{h'})_w, C_{h'}\}$  (Figures 6.19(a) and (b) respectively) are non-spanning even cycle covers of G(P). Since G(P) is minimally non-Pfaffian, sign $(D_1) = (D_1 + D_2)$ 



Figure 6.19: Cycle even covers through the shortcut arc w

 $\operatorname{sign}(D_2) = 0$  and since

$$\operatorname{sign}(D_1) + \operatorname{sign}(D_2) = \operatorname{sign}(C_h) + \operatorname{sign}(C_f) + \operatorname{sign}(C_{h'}) + \operatorname{sign}(C_{f'})$$
$$= \operatorname{sign}(C, K) + \operatorname{sign}(C, K'),$$

it follows that  $\operatorname{sign}(C, K) + \operatorname{sign}(C, K') = 0$ .

(ii) If w is a chord of C that strongly crosses K and K' then without loss of generality we may assume that the tail of w is contained in  $V(C_h) \cap V(C_{f'})$  and that the head of w is contained in  $V(C_f) \cap V(C_{h'})$ , as in Figure 6.20(a). Let C' be the circuit in  $C \cup \{f, f', w\}$  with  $f, f', w \in C'$  as shown in Figure 6.20(b). Then



Figure 6.20: A chord of C that strongly crosses K and K'

 $D_1 = \{C', C_w, C_f, C_{h'}\}$  and  $D_2 = \{C', C_w, C_{f'}, C_h\}$  are both non-spanning cycle even covers of G(P) and thus  $\operatorname{sign}(D_1) = 0 = \operatorname{sign}(D_2)$ . Then

$$sign(D_1) + sign(D_2) = sign(C_f) + sign(C_{h'}) + sign(C_{f'}) + sign(C_h)$$
$$= sign(C, K) + sign(C, K')$$

and thus  $\operatorname{sign}(C, K) = \operatorname{sign}(C, K')$ .

Using Lemmas 6.16, 6.17, and 6.18, we can now show that P has an non-Pfaffian bases certificate containing at most 32 bases. Since C is a neutral circuit in G(P), there exist decomposing circuits F and H of C such that  $\operatorname{sign}(C, F) \neq \operatorname{sign}(C, H)$  and by Lemma 6.16 the circuits F and H cross in C. There are two cases to consider, depending on whether F and H strongly cross or weakly cross. We first consider the case that Fand H strongly cross, and construct a contributing accordion in C.

**Lemma 6.19.** Let C be a neutral directed circuit in G(P) and let F and H be strongly crossing decomposing circuits of C with  $\operatorname{sign}(C, F) \neq \operatorname{sign}(C, H)$ . If f is a C-chord edge of F, then  $V(C_f)$  can be decomposed into at most 8 contributing sets.



Figure 6.21: strongly crossing decomposing circuits F and H

*Proof.* Let  $V(F) = \{f_1, f_2, f_3, f_4\}$  and  $V(H) = \{h_1, h_2, h_3, h_4\}$  where we may assume that the cyclic order of V(F) and V(H) along C is as shown in Figure 6.21(a). Let  $f = f_1 f_4$  and  $h = h_1 h_2$ . We show that  $C_f$  has a decomposition into at most 8 contributing subgraphs.

Let R be the path in C from  $f_3$  to  $h_1$  together with the edge  $h_1h_4$  and let Q be the edge  $h_3h_2$  together with the path in C from  $h_2$  to  $f_2$ . Then R and Q are vertex disjoint directed paths and F and H are parallel bridge circuits between R and Q. The circuit  $C_f$  is a circuit in  $R \cup Q \cup F \cup H$  with ends F and H (see Figure 6.21(b)) and by Lemma 6.18(i), R and Q do not have any internal shortcut arcs. We use induction on  $|C_f|$  to prove that there is a contributing accordion between R and Q with ends F and H. For the base case, if  $C_f$  has length two then it is contributing and thus there is a trivial accordion between R and Q with ends F and H.

If  $C_f$  is contributing then it has a trivial decomposition into one contributing cycle, and thus we may assume that  $C_f$  has a decomposing circuit K. Suppose E(K) intersects E(R) and E(K) contains f but not h. Let  $k = E(K) \cap E(R)$  with  $k = k_1k_2$ , and over all such decomposing circuits in  $C_f$ , choose K to minimize the distance along R from  $f_4$  to  $k_1$ . If  $C_{k_1f_4}$  is not contributing then let K' be a decomposing circuit of  $C_{k_1f_4}$ . If  $k_1f_4 \notin E(K')$  then K' is a decomposing circuit of C that is parallel to both F and H, which contradicts Lemma 6.16. If  $k_1f_4 \in E(K')$  then let  $k' = E(K') \cap E(C)$  with  $k' = k'_1k'_2$ . Since K' and K share exactly one edge, namely  $k_1f_4$ , either  $k'_1k_2$  or  $f_1k'_2$  is a joining edge between K and K'. Both joining edges lead to contradictions:  $k'_1k_2$  is a shortcut arc of R and if  $f_1k'_2$  is a chord in C then  $C_f$  has a decomposing circuit through f and k' that contradicts the choice of K. Thus  $C_{k_1f_4}$  is contributing.

The decomposing circuits F and K share the edge f and thus there is a joining edge between K and F. If  $f_3k_2$  is the joining edge then C has a decomposing circuit K'with  $f_3f_4, k \in E(K')$ . The circuits K' and F weakly cross in C with  $E(K') \cap E(F) =$  $f_3f_4$ , and the joining edge  $f_1k_2$  between K' and F is a backwards arc of  $C \setminus \{f_3f_4\}$ . Thus Lemma 6.17(i) implies that  $\operatorname{sign}(C, F) = \operatorname{sign}(C, K')$  which then implies that  $\operatorname{sign}(C, K') \neq \operatorname{sign}(C, H)$ . This is a contradiction of Lemma 6.16, since K' and H are parallel in C. Thus  $k_1 f_2$  is the unique joining edge between F and K, as shown in Figure 6.22. Let F' be the decomposing circuit of C with C-edge arcs  $\{f_1 f_2, k\}$ .



Figure 6.22: Building an accordion

Then F' and F weakly cross in C and F' and H strongly cross. The joining edge  $k_1f_4$ between F and F' implies that  $\operatorname{sign}(C, F) = \operatorname{sign}(C, F')$  by Lemma 6.17(i), and thus  $\operatorname{sign}(C, F') \neq \operatorname{sign}(C, H)$ . By the induction hypothesis, there is a contributing accordion between R and Q with ends F' and H. Adding E(F) and  $k_1f_4$  to this accordion gives a contributing accordion between R and Q with ends F and H. Thus there is an accordion between R and Q with ends F and H whenever a decomposing circuit K of  $C_f$  contains an edge of R and contains f but not h. By reversing the orientation of the edges of G, we conclude that there is a contributing accordion between R and Q with ends F and H whenever a decomposing circuit K of  $C_f$  contains an edge of Q and the edge f but not h. Let C' be the directed circuit with  $E(C') = E(C)\Delta E(F)\Delta E(H)$ . Then  $C_f = C'_h$ and thus if  $C_f$  has a decomposing circuit K with  $h \in E(K)$  and  $f \notin E(K)$  then  $C'_h$  has a decomposing circuit K with  $h \in E(K)$  and  $f \notin E(K)$ . Hence there is a contributing accordion between R and Q whenever a decomposing circuit K of  $C_f$  contains the edge f but not the edge h.

So we may assume that any decomposing circuit for  $C_f$  contains the edge f if and only if it also contains h. We show that if such a circuit exists then  $C_{h_1f_4}$  and  $C_{f_1h_2}$  are contributing.

Let K be a decomposing circuit of  $C_f$  with  $f, h \in E(K)$  and suppose  $C_{h_1f_4}$  is not contributing. If K' is a decomposing circuit of  $C_{h_1f_4}$  and K' is edge disjoint from K then K' is a decomposing circuit of C that is parallel to both F and H, contradicting Lemma 6.16. Thus we may assume that K' is a decomposing circuit of  $C_{h_1f_4}$  with  $h_1f_4 \in E(K')$ . Let  $k' = E(K') \cap E(R)$  with  $k' = k'_1k'_2$ , as shown in Figure 6.23. If  $f_1k'_2$  is a joining edge between K' and K then  $C_f$  has a decomposing circuit that intersects  $C_f$  in edges f and k', contradicting our assumption that any decomposing circuit of  $C_f$  through f also contains h. If  $k'_1h_2$  is a joining edge between K and K' then  $C_f$  has a decomposing circuit that intersects  $C_f$  in edges h and k', again contradicting our assumption that any decomposing circuit of  $C_f$  through h also contains f. It follows that  $C_{h_1f_4}$  is contributing, and by symmetry we conclude that  $C_{f_1h_2}$  is contributing.



Figure 6.23: A decomposing circuit through f and h

Consider F and K, which share the edge f. By reversing the orientation of G if necessary, we may assume that  $f_3h_2$  is a joining edge between F and K. This creates a decomposing circuit K' of C with  $h, f_3f_4 \in E(K')$ . Since K' and F are weakly crossing decomposing circuits of C, Lemma 6.17(i) and the joining edge  $f_1h_2$  imply that  $\operatorname{sign}(C, K') = \operatorname{sign}(C, F)$  and therefore  $\operatorname{sign}(C, K') \neq \operatorname{sign}(C, H)$ . Applying Lemma 6.17(i) again implies that  $h_3f_4$  is the unique joining edge between K' and H.

Finally, suppose every decomposing circuit of  $C_f$  contains neither f nor h, and let K be a decomposing circuit of  $C_f$ . Since Lemma 6.16 implies that K can not be parallel to both F and H, K is a bridge circuit between R and Q. Let  $k_1k_2 = E(K) \cap E(R)$ ,  $l_1l_2 = E(K) \cap E(Q)$  and over all such bridge circuits choose K to minimize  $|(C_f)_{k_1l_2}|$ . Then  $(C_f)_{k_1l_2}$  is contributing and K and F are parallel decomposing circuits in C. Lemma 6.16 implies that  $\operatorname{sign}(C, K) = \operatorname{sign}(C, F)$ . This implies that K and H give opposite signs to C, and by the induction hypothesis there is a contributing accordion between R and Q with ends K and H. Hence by induction, there is a contributing accordion between R and Q with ends F and H.

Let  $\mathcal{C}$  be the contributing circuits in  $\mathcal{H}$ . Let  $\mathcal{C}_R$  be the circuits in  $\mathcal{C}$  that only use vertices in R and let  $\mathcal{C}_Q$  be the circuits in  $\mathcal{C}$  that only use vertices in Q. If  $\mathcal{C}_{f,h}$  contains the circuits in  $\mathcal{C}$  that use f or h and  $\mathcal{C}' = \mathcal{C} \setminus \{\mathcal{C}_R, \mathcal{C}_Q, \mathcal{C}_{f,h}\}$  then  $\{\mathcal{C}_R, \mathcal{C}_Q, \mathcal{C}_{f,h}, \mathcal{C}'\}$  is a partition of  $\mathcal{C}$ . Since  $\mathcal{C}_{f,h}$  contains at most two circuits, to prove the lemma it remains to show that the circuits in  $\{\mathcal{C}_R, \mathcal{C}_q, \mathcal{C}'\}$  can be covered with at most six contributing subgraphs. Since R and Q have no internal shortcut arcs, Lemma 6.14 implies that  $\mathcal{C}_R$ and  $\mathcal{C}_Q$  can each be partitioned into at most two contributing cycles.

If  $R' = R \setminus \{f_3 f_4, h_1 h_4\}$  and  $Q' = Q \setminus \{h_3 h_2, f_1 f_2\}$  and  $\mathcal{H}'$  is the accordion between R'and Q' given by the bridge circuits in  $\mathcal{H}$  that are bridge circuits between R' and Q', then the circuits in  $\mathcal{C}'$  are all accordion circuits in H'. Since R and Q do not have internal shortcut arcs, R' and Q' do not have shortcut arcs. Suppose there exists an internally twisted bridge circuit K' in H where K' strongly crosses the parallel bridge circuits  $K_1$ and  $K_2$  in H'. Then K' is parallel to F and thus  $\operatorname{sign}(C, K') \neq \operatorname{sign}(C, H)$ . Furthermore, since K' is internally twisted, both  $K_1$  and  $K_2$  are parallel to F and cross H. If  $E(K_1)$  precedes  $E(K_2)$  along R then the edge of  $K_1$  directed from the tail of  $E(K_1) \cap E(Q)$ to the head of  $E(K_1) \cap E(R)$  strongly crosses K and H, as shown in Figure 6.24. This



Figure 6.24: An internally twisted bridge circuit

contradicts Lemma 6.18(ii), and thus  $\mathcal{H}'$  has no twisted bridge circuits. Lemma 6.15 therefore implies that the circuits in  $\mathcal{C}'$  can be partitioned into at most two contributing sets, and thus  $\mathcal{C}$  can be partitioned into at most eight contributing sets. It follows that if F and H are strongly crossing decomposing circuits of C with  $\operatorname{sign}(C, F) \neq \operatorname{sign}(C, H)$  and f is a C-chord edge of F, then  $C_f$  can be partitioned into at most 8 contributing sets.  $\Box$ 

We can now prove our main result for neutral circuits.

**Theorem 6.20.** If P is a minimally non-Pfaffian series-parallel matroid pair and G(P) has a neutral circuit, then P has a non-Pfaffian bases certificate containing at most 32 bases.

Proof. Let P be a minimally non-Pfaffian matroid pair and let C be a neutral circuit in G(P). Suppose there exist strongly crossing decomposing circuits F and H of C such that  $\operatorname{sign}(C, F) \neq \operatorname{sign}(C, H)$ . If  $\{f, f'\}$  and  $\{h, h'\}$  are the C-chord edges of F and H respectively, then by Lemma 6.19,  $C_f, C_{f'}, C_h$ , and  $C_{h'}$  each partition into at most eight contributing sets. Since  $\{C_f, C_{f'}, C_h, C_{h'}\}$  is a certificate that G(P) is not signable, there is a set of at most 32 contributing subgraphs that proves that G(P) is not signable. By the bijection between contributing subgraphs and bases of P, there is a non-Pfaffian bases certificate containing at most 32 bases of P.

We may assume that whenever C' is a directed circuit in G(P) and F and H are decomposing circuits of C' with  $\operatorname{sign}(C', F) \neq \operatorname{sign}(C', H)$ , then F and H are weakly crossing in C'. Let F and H be such a pair of decomposing circuits for C let  $\{e, f\}$  and  $\{e, h\}$  be the C-chord edges of F and H respectively, where the orientation of G is such that h precedes f on the directed path  $C \setminus \{e\}$ . For i = 1, 2, assume  $f_i, e_i, h_i \in V(C)$ are such that  $e = e_1e_2$ ,  $f = f_1f_2$ , and  $h = h_1h_2$ . By Lemma 6.17(i),  $h_1f_2$  is the unique joining edge between F and H (see Figure 6.25(a)). Let Q be the path in C from h to f



Figure 6.25: Weakly crossing decomposing circuits of C and C'

and over all neutral circuits C and all pairs of weakly crossing decomposing circuits F, H with  $sign(C, F) \neq sign(C, H)$ , choose C, F, and H to minimize |Q|.

Let  $f' = f_1 e_2$  and  $h' = e_1 h_2$  and let C' be the directed circuit in G with

$$E(C') = E(C)\Delta\{e, f, h\}\Delta\{e', f', h'\}$$

Then C' has weakly crossing decomposing circuits F' and H' where e' and f' are the C'-edge arcs of F' and E' and h' are the C'-edge arcs of H'. Furthermore,  $C_{f'} = C'_h$  and  $C_{h'} = C'_f$  (Figure 6.25(b)), and the assumption that  $\operatorname{sign}(C, F) \neq \operatorname{sign}(C, H)$  implies that  $\operatorname{sign}(C', F') \neq \operatorname{sign}(C', H')$ . Let R be the directed path in C from  $e_2$  to  $h_1$ , such that  $C_{f'} = R, h, Q, f'$ .

Claim 1: There is no decomposing circuit of  $C_{f'}$  that uses both edge h and edge f'. Proof of Claim 1: Since  $\operatorname{sign}(C, F) \neq \operatorname{sign}(C, H)$ , Lemma 6.17(i) implies that  $f_1h_2$  is not an edge in G(P). In particular there is no decomposing circuit of  $C_{f'}$  that uses  $f_1h_2$  and therefore there is no decomposing circuit of  $C_{f'}$  that uses both edge h and edge f'.

Claim 2: There is no decomposing circuit in  $C_{f'}$  that uses edge h. Proof of Claim 2: Let K be a decomposing circuit in  $C_{f'}$  with  $C_{f'}$ -edge arcs h and k. By Claim 1, either k is an edge of R or k is an edge of Q.

Let  $k = k_1k_2$  and suppose first that k is an edge of R. Then K and F are parallel and therefore sign(C, K) = sign(C, F) and thus sign $(C, K) \neq \text{sign}(C, H)$ . Since K and H are weakly crossing, Lemma 6.17(i) implies that  $e_1k_2$  is the unique joining edge between K and H and  $k_1e_2 \notin E(G(P))$ , as shown in Figure 6.26(a). The edge  $e_1k_2$  creates a decomposing circuit K' in C' with  $E(K') \cap E(C') = \{h', k\}$  (see Figure 6.26(b)). Since K' and F' are strongly crossing in C', sign(C', K') = sign(C', F') by assumption and thus sign $(C', K') \neq \text{sign}(C', H')$ . The decomposing circuits K' and H' are weakly crossing in C' and the joining edge  $h_1k_2$  between K' and H' therefore contradicts Lemma 6.17(i). Hence if h and k are the  $C_{f'}$ -edge arcs of K then k is not an edge of R.

Suppose instead that k is an edge of Q. Then K and F are again parallel and thus  $\operatorname{sign}(C, K) = \operatorname{sign}(C, F)$  and  $\operatorname{sign}(C, K) \neq \operatorname{sign}(C, H)$ . Since K and H are weakly



Figure 6.26: Weakly crossing decomposing circuits of C and C'

crossing, Lemma 6.17(i) implies that the edge  $k_1e_2$  is the unique joining edge of K and H and  $e_1k_2 \notin E(G)$  (Figure 6.27(a)).



Figure 6.27: K contains h and an edge in Q

The decomposing circuits H' and K share the edge h, and since  $e_1k_2$  is not an edge in G(P),  $k_1f_2$  is the unique joining edge between H' and K. Then C' has a decomposing circuit K' with C'-edge arcs  $h_1f_2$  and  $k_1k_2$ , and K' weakly crosses both F' and H'. (See Figure 6.27(b).) This contradicts the minimality of |Q| since the both the directed path in C' from  $h_2$  to  $k_1$  and the directed path in C' from  $k_2$  to  $f_1$  are strict subpaths of Qand either sign $(C', K') \neq \text{sign}(C', F')$  or sign $(C', K') \neq \text{sign}(C', H')$ .

It follows that there is no decomposing circuit of  $C_{f'}$  that contains the edge h and the proof of Claim 2 is complete.

Claim 3: There is no decomposing circuit in  $C_{f'}$  that contains the edge h. Proof of Claim 3: Reversing the orientation of the edges in G(P) and considering C', Claim 2 implies that a decomposing circuit of  $C'_h$  does not contain f'. Since  $C'_h = C_{f'}$ , there is no decomposing circuit of  $C'_{f'}$  that contains f'.

Claim 4: The directed circuits  $C_{f'}$  and  $C_{h'}$  are contributing in G(P). Proof of Claim 4: Suppose  $C_{f'}$  is not contributing and let K be a decomposing circuit of  $C_{f'}$ . By Claims 1 and 2, K is edge disjoint from f' and h. If both  $C_{f'}$ -edge arcs of K are in R then K is parallel to both F and H, which contradicts Lemma 6.16. Thus at most one  $C_{f'}$ -edge arc of K is an edge of R. Similarly, at most one  $C_{f'}$ -edge arc of K is an edge of Q. Thus K is a bridge circuit between R and Q, and therefore K crosses H and is parallel to F. By Lemma 6.16,  $\operatorname{sign}(C, K) = \operatorname{sign}(C, F)$  and thus  $\operatorname{sign}(C, K) \neq \operatorname{sign}(C, H)$ . This contradicts the assumption C does not contain strongly crossing decomposing circuits which give different signs to C and the circuit  $C_{f'}$  must be contributing.

By reversing the orientation of G(P) it follows that  $C_{h'}$  is contributing. Since

$$\{C_{e_1f_2}, C_{f'}, C_{h_1e_2}, C'_h\}$$

is a certificate that G(P) is not signable, it remains to show that  $C_{e_1f_2}$  and  $C_{h_1e_2}$  can be partitioned into a small number of contributing cycles. By symmetry it suffices to consider  $C_{h_1e_2}$ .

Let w be a backwards arc of R and let K be a decomposing circuit of  $C_w$ . If  $w \notin E(K)$ then K is also a decomposing circuit of  $C_{f'}$ . This contradicts Claim 3, and thus  $w \in E(K)$ . Let w' and w'' be the  $C_w$ -chord arcs of K. Then w' and w'' are both backwards arcs of R and  $V(C_w)$  is the disjoint union of  $V(C_{w'})$  and  $V(C_{w''})$ . Since both  $C_{w'}$  and  $C_{w''}$  are shorter than  $C_w$ , repeating this process leads to a decomposition of  $C_w$  into contributing circuits of the form  $C_{w'}$  for some backwards arc w' or R. Since  $h_1e_2$  is a backwards arc of R, we can therefore decompose  $C_{h_1e_2}$  into contributing circuits of the form  $C_w$  where w is a backwards arc of R. Let C be the set of contributing circuits in such a decomposition of  $C_{h_1e_2}$  and let  $\{C_A, C_B\}$  be a partition of C such that if  $D_A, D_B \in C$ and an edge of R is incident to both a vertex in  $D_A$  and a vertex in  $D_B$ , then  $D_A \in C_A$ if and only if  $D_B \in C_B$ .

Let  $D_1, D_2 \in \mathcal{C}_A$ . If  $D_1 \cup D_2$  is not contributing, then there exists a decomposing circuit K in G(P) with  $V(K) = \{u_1, u_2, v_1, v_2\}$  where  $u_1v_1 \in E(D_1) \cap E(K)$  and  $u_2v_2 \in E(D_2) \cap E(K)$ . By the construction of  $\mathcal{C}$  we may assume that both  $u_1$  and  $v_1$  precede  $u_2$  and  $v_2$  along R, and by the construction of  $\mathcal{C}_A$  neither  $u_1v_2$  nor  $u_2v_1$  is an edge of R. This is a contradiction, since  $u_1v_2$  is a shortcut arc of R. Therefore the union of all contributing circuits in  $\mathcal{C}_A$  is a contributing subgraph, as is the union of all contributing circuits in  $\mathcal{C}_B$ , and hence  $V(C_{h_1e_2})$  is the disjoint union of at most two contributing subgraphs. Reversing the orientation of the edges of G(P) implies that  $V(C_{e_1f_2})$  is also the disjoint union of at most two contributing cycles, and we therefore have a set of at most six contributing cycles which certify that V(C) is not signable.

The constant of 32 in Theorem 6.20 is not claimed to be optimal. We do not have an example of a neutral circuit that requires more than six bases to certify that it is non-Pfaffian, and in fact we do not have an example of a minimal non-Pfaffian matroid pair that requires more than six bases to certify that it is non-Pfaffian.

# 6.7 Odd double cycles

There remains one case left to consider for series-parallel matroid pairs, namely that P is a non-Pfaffian series-parallel matroid pair and there are no neutral circuits in G(P). Theorem 6.10 implies that G(P) does not have an edge signing that respects the decomposing parity of all directed circuits in G(P), and we can therefore use a characterization of Gerards [15] to get a structural result for G(P). Unfortunately the structure is not induced in G(P) and so this result on G(P) does not translate back to a good structural characterization of P. Some definitions are required first.

Given two directed circuits C and C', the connected components in  $C \cap C'$  are directed paths which we call the intersection paths of C and C'. If there are at least two intersection paths of C and C' and the cyclic order of the intersection paths on C is the inverse of the cyclic order of the paths on C', then we say C and C' are *twined* (Figure 6.28).



Figure 6.28: Twined circuits

Let  $C^F$  and  $C^B$  be twined directed circuits in the directed graph G. If  $C^F$  and  $C^B$  have k intersection paths and  $k \geq 2$ , then  $C^F \cup C^B$  contains k directed circuits  $\{C^1, C^2, \ldots, C^k\}$  each distinct from  $C^F$  and  $C^B$ . The union of these k circuits together with  $C^F$  and  $C^B$  is called a *weak* k-double cycle. When each intersection path of  $C^F$  and  $C^B$  is a single vertex, then  $C^F \cup C^B$  is a k-double cycle. Let D be a weak k-double cycle and let  $\alpha_C \in \{0, 1\}$  be defined for each of the directed circuits in D. If  $\alpha_C = 1$  for an odd number of directed circuits in D, then D is an odd double cycle with respect to  $\alpha$ . Note that D contains k + 2 directed circuits and each edge in D is in either two or four of these k + 2 circuits. If D is an odd double cycle in the directed graph G and s is an edge signing of G, then let s(D) be the sum of s(C) over all circuits C in D. Since each  $e \in E(D)$  is in an even number of circuits in D, s(D) = 0 for all edge signings s of D. It follows that if G has an odd double cycle as a subgraph with respect to  $\alpha$ . The structural theorem of Gerards [15] that is used in this section is that the converse also holds.

**Theorem 6.21 (Gerards).** If G is a directed graph and  $\alpha_C \in \{0, 1\}$  is defined for all directed circuits C in G, then either G has a proper edge signing with respect to  $\alpha$  or G

has a weak double cycle that is odd with respect to  $\alpha$  as a subgraph.

If P is a series-parallel non-Pfaffian matroid pair and G(P) does not have a neutral circuit, then Theorem 6.21 together with Theorem 6.11 implies that G(P) has a weak odd double cycle with respect to the circuit signing  $\alpha_C = \frac{|C|}{2} + dp(C)$ . The results in this section can be combined with the results Section 6.6 to get the following result for series-parallel matroid pairs:

**Theorem 6.22.** If P is a minimally non-Pfaffian series-parallel matroid pair, then either G(P) has a Hamiltonian neutral cycle or G(P) has a spanning weak odd k-double cycle with  $k \leq 4$ .

If G(P) has a weak odd k-double cycle D as an induced subgraph and each of the k+2 circuits in D is contributing then each circuit corresponds to a basis of P and thus there is a set of at most k+3 bases which certifies that P is non-Pfaffian. Unfortunately, the odd double cycle in G(P) that is the obstruction for P to be Pfaffian need not be an induced subgraph of G(P) and therefore the circuits in D need not be contributing. We have been able to show that in most cases, the weak odd double cycle in G(P) has a decomposition into at most 40 contributing circuits, and thus there is a set of at most 40 bases which certifies that P is non-Pfaffian. Based on this partial result and the results of Section 6.6, we make the following weakening of Conjecture 3.12:

**Conjecture 6.23.** There exists a constant c such that if P is a series-parallel matroid pair, then either P is Pfaffian or P has a non-Pfaffian bases certificate with at most c bases.

If this conjecture is true, then it contains some of the power of an excluded minor characterization: if there is a finite set of excluded minors for Pfaffian matroids, then either a matroid pair is Pfaffian or there is a certificate of constant size, namely one of the excluded minors, which proves that the pair is non-Pfaffian. Similarly, if Conjecture 6.23 holds, then either a series-parallel matroid pair is Pfaffian or there is a certificate of constant size, namely a non-Pfaffian bases certificate, which proves that the pair is non-Pfaffian.

Some notation and terminology for weak double cycles is required. Let  $C^F$  and  $C^B$  be twined directed circuits and let D be the weak double cycle formed by  $C^F \cup C^B$ . If  $C^F$  and  $C^B$  have k intersection paths, then let  $S = \{C^1, C^2, \ldots, C^k\}$  be the set of distinct directed circuits in D such that each  $C^i$  is distinct from  $C^F$  and  $C^B$  and

$$C^F \cup C^B = C^1 \cup C^2 \cup \dots \cup C^k.$$

When k > 2 then the circuits in S are called the *short circuits* of D and the circuits  $C^F$  and  $C^B$  are the *long circuits* of D. If D is a weak k-double cycle then the cycle size of D is k and the cycle length of D is the sum of the lengths of the k + 2 directed circuits in D.

When D is a weak k-double cycle with short circuits  $\{C^1, C^2, \ldots, C^k\}$ , we assume that for all  $i, C^i$  intersects  $C^{i+1}$  and  $C^{i-1}$ , where the index addition is modulo k. The intersection path of  $C^i \cap C^{i-1}$  is denoted  $P_i$  and if  $C^F$  and  $C^B$  are the long circuits in D then we let  $F_i$  be the subpath of  $C^F \cap C^i$  and  $B_i$  be the subpath of  $C^B \cap C^i$  such that  $F_i, P_i, B_i$ , and  $P_{i+1}$  are all edge disjoint and  $C^i = P_i \cup F_i \cup P_{i+1} \cup B_i$ . We assume that  $C^F$  and  $C^B$  are labeled such that the  $C^F$  traverses the paths  $P_i$  in forward order,  $P_1, P_2, \ldots, P_k$ , and  $C^B$  therefore traverses the paths in reverse order,  $P_k, P_{k-1}, \ldots, P_1$  (see Figure 6.29).



Figure 6.29: The subpath labels for a weak 6-double cycle

Let G be a directed graph and suppose that the weak double cycle D is a subgraph of G. If u and v are vertices in D and  $uv \in E(G) \setminus E(D)$ , then uv is a chord of D. We classify three types of chords of D: forbidden short circuit chords, forbidden long circuit chords, and forbidden between-circuit chords. These chords are collectively referred to as forbidden chords of D.

**Forbidden short circuit chords:** Let uv be a chord in a short circuit  $C^i$  of D and suppose that uv is also a chord in a long circuit  $C^F$  or  $C^B$  of D. If u precedes v along the intersection path between  $C^i$  and  $C^F$  or  $C^B$ , then uv is a forbidden short circuit chord of D. Two examples of forbidden short circuit chords are shown in Figure 6.30(a).

**Forbidden long circuit chords:** Let uv be a chord in a long circuit of D, where we may assume that uv is a chord of  $C^F$ . Let i and j be such that either  $u \in P_i$  or u is an internal vertex in  $F_i$ , and either  $v \in P_{j+1}$  or v is an internal vertex in  $F_j$ . If  $j \notin \{i, i - 1, i - 2\}$ , then uv is a forbidden long circuit chord of D. A forbidden long circuit chord is shown in Figure 6.30(b).

**Forbidden between-circuit chords:** Let uv be a chord in D such that after possibly relabeling the circuits in D, u is an internal vertex in  $F_i$ , v is an internal vertex in  $B_j$ , and  $j \neq i$ . Further, suppose that if j = i - 1 then v is not the second vertex in  $B_{i-1}$  (Figure 6.31(a)) and if j = i + 1 then u is not the second to last vertex in  $F_i$


Figure 6.30: Forbidden short (a) and long (b) circuit chords

(Figure 6.31(b)). The chord uv is then a forbidden between-circuit chord of D.



Figure 6.31: Forbidden between-circuit chords

Let G be a directed graph with circuit signing  $\alpha$  and suppose that G has an odd weak double cycle with respect to  $\alpha$ . Let n be the minimum number of vertices in an odd weak double cycle in G and let k be the minimum cycle size of an odd weak double cycle in G with n vertices. A minimal weak odd double cycle in G is a weak k-double cycle D with n vertices and minimum cycle length. We consider where chords of D may lie.

**Lemma 6.24.** Let G be a directed graph and let D be a minimal weak odd k-double cycle in G with respect to a circuit signing of G. If  $k \ge 5$ , then D does not have any forbidden chords.

*Proof.* Assume that D has short circuits  $\{C^1, C^2, \ldots, C^k\}$  and long circuits  $\{C^F, C^B\}$  and let uv be a forbidden short circuit chord of D. By relabeling the circuits in D we may assume that uv is a chord in  $C^1$  and a chord in  $C^F$ . The circuits  $C^1$  and  $C_{uv}^F$  are twined with two intersection paths and form the weak 2-double cycle  $D_1 = \{C^1, C_{uv}^1, C^F, C_{uv}^F\}$ . The circuits  $C^B$  and  $C_{uv}^F$  are also twined with k intersection paths and form the weak k-double cycle  $D_2$  with short circuits  $\{C_{uv}^1, C^2, \ldots, C^k\}$ . Since D is the symmetric difference of the circuits in  $D_1$  with the circuits in  $D_2$ , either  $D_1$  or  $D_2$  is an odd weak double-cycle.

If  $D_1$  is odd then G has an odd weak 2-double-cycle, contradicting the minimality of D. If  $D_2$  is odd then since  $|C_{uv}^1| < |C^1|$  and  $|C_{uv}^F| < |C^F|$ , either  $D_2$  has fewer vertices than D or the cycle length of  $D_2$  is less than the cycle length of D. This again contradicts the minimality of D, and thus D does not contain any forbidden short circuit chords.

Suppose that uv is a forbidden long circuit chord of D. By relabeling the circuits in D, we may assume that either  $u \in P_1$  or u is an internal vertex of  $F_1$  and that either  $v \in P_{j+1}$  or v is an internal vertex in  $F_j$ , with  $j \neq k - 1, k, 1$ . Let  $P'_1$  be the directed path in  $C^1$  from the tail of  $P_1$  to u, and let  $P'_{j+1}$  be the directed path in  $C^j$  from v to the head of  $P_{j+1}$ . Let  $P^*$  be the directed path in  $C^B$  from the head of  $P_{j+1}$  to the tail of  $P_1$  or  $v \in P'_1 \cup \{uv\} \cup P'_{j+1}$ .

The circuits  $C_{uv}^F$  and  $C^B$  are twined with intersection paths  $P'_{j+1}, P_{j+2}, \ldots, P_k, P'_1$  and form a weak (k - j + 1)-double cycle  $D_1$ . The circuits  $C^*$  and  $C^F$  are also twined, with intersection paths  $P'_1, P_2, \ldots, P_j, P'_{j+1}$  and thus form a weak (j + 1)-double cycle  $D_2$ . The short circuits in  $D_1$  are  $\{C^{j+1}, C^{j+2}, \ldots, C^k, C^*\}$  and the short circuits in  $D_2$  are  $\{C^1, C^2, \ldots, C^j, C_{uv}^F\}$ , and therefore D is the symmetric difference of the circuits in  $D_1$ with the circuits in  $D_2$ . This implies that either  $D_1$  or  $D_2$  is odd, and since  $j \neq k-1, k, 1$ by assumption, either  $D_1$  or  $D_2$  contradicts the minimality of D. Hence D does not contain any forbidden long circuit chords.

Suppose uv is a forbidden between-circuit chord of D. Up to relabeling we may assume that u is an internal vertex of  $F_1$  and v is an internal vertex in  $B_j$  for some  $j \neq 1$ . Let  $P'_1$  be the directed path in  $C^1$  from the tail of  $P_1$  to u, and let  $P'_j$  be the directed path in  $C^j$  from the tail v to the head of  $P_j$ . Let  $P'_B$  be the directed path in  $C^B$  from the head of  $P_j$  to the tail of  $P_1$  and similarly let  $P'_F$  be the directed path in  $C^F$  from the head of  $P_j$  to the tail of  $P_1$ . Let  $C^{B*}$  be the directed circuit in  $P'_B \cup P'_1 \cup P'_j \cup \{uv\}$  and let  $C^{F*}$  be the directed circuit in  $P'_F \cup P'_1 \cup P'_j \cup \{uv\}$ .

The circuits  $C^{B*}$  and  $C^F$  are twined with intersection paths  $P'_1, P_2, \ldots, P_j$  and thus form a weak *j*-double cycle  $D_1$ . Similarly, the circuits  $C^{F*}$  and  $C^B$  are twined with intersection paths  $P'_j, P_{j+1}, \ldots, P_k, P_1$  and thus form a weak (k - j + 2)-double cycle  $D_2$ . The short circuits in  $D_1$  are  $\{C^1, C^2, \ldots, C^{j+1}, C^{F*}\}$  and the short circuits in  $D_2$  are  $\{C^j, C^{j+1}, \ldots, C^k, C^{B*}\}$ . The symmetric difference of the circuits in  $D_1$  with the circuits in  $D_2$  is D, and therefore either  $D_1$  or  $D_2$  is odd.

If  $D_1$  is odd then since  $D_1$  has cycle size j, the minimality of D implies that j = k. The assumption that uv is a forbidden between-circuit chord then implies that v is not the second vertex in  $B_j$  and the internal vertices of  $B^j$  between the tail of  $B_j$  and vare therefore not contained in  $D_1$ . This implies that  $D_1$  has fewer vertices than D, contradicting the minimality of D.

If  $D_2$  is odd then since  $D_2$  has cycle size k - j + 2 and  $j \neq 1$ , the minimality of the cycle size of D implies that j = 2. The assumption that uv is a forbidden between-circuit chord then implies that u is not the second to last vertex of  $F_1$ , and thus the internal

vertices on  $F_1$  between v and the head of  $F_1$  are not contained in  $D_2$ . This implies that  $D_2$  has fewer vertices than D and contradicts the minimality of D. Thus D does not have any forbidden between-circuit chords.

Suppose D is a minimal weak odd k-double cycle in the directed graph G with  $k \geq 5$ . We would like a decomposition of each of the circuits in D into contributing circuits. If each circuit is induced then this is easy, and thus we may assume that a circuit in D has a decomposing circuit. Let K be a decomposing circuit in the short circuit  $C^i$  in D. If the  $C^i$ -edge arcs of K are contained in  $P_i$  or  $P_{i+1}$  then one of the  $C^i$ -chord edges of K is a forbidden short circuit chord. If both  $C^i$ -edge arcs of K are contained in  $F_i$  or both are contained in  $B_i$  then one of the  $C^i$ -chord edges of K is a forbidden short circuit chord. Thus all decomposing circuits in  $C^i$  are bridge circuits between  $F_i$  and  $B_i$ . Using this observation we can construct a contributing accordion between  $F_i$  and  $B_i$ .

Let K be a decomposing circuit of  $C^i$  with  $V(K) = \{u_1, v_1, u_2, v_2\}$  where  $u_1v_1$  is an edge in  $F_i$  and  $u_2v_2$  is an edge in  $B_i$ , and over all such decomposing circuits in  $C^i$ , choose K to minimize the sum of the distance from the tail of  $F_i$  to  $u_1$  with the distance from  $v_2$  to the head of  $B_i$ . A decomposing circuit in  $C^i_{u_1v_2}$  which is edge disjoint from K contradicts the minimality in the choice of K and thus if  $C^i_{u_1v_2}$  is not contributing it has a decomposing circuit K' that contains the arc  $u_1v_2$ . Each of the possible joining edges between K and K' either contradicts the minimality of the choice of K or is a forbidden short circuit chord. Thus the choice of K implies that  $C^i_{u_1v_2}$  is contributing.

Consider  $C_{u_2v_1}^i$ , and suppose it has a decomposing circuit K' such that K' contains the edge  $u_2v_1$  and contains an edge uv in  $F^i$ . Choose K' to minimize the distance along  $F^i$  from  $v_1$  to u. The edge  $u_1v$  is a forbidden short circuit chord, and thus  $uv_2$  is the joining edge between K and K'. There is therefore a bridge circuit K'' between  $F_i$  and  $B_i$  such that K and K'' are the ends of the accordion circuit  $C_{uv_1}^i$ . The minimality of the length of the path between  $v_1$  and u implies that  $C_{uv_1}^i$  is contributing. If  $C_{u_2v_1}^i$  has a decomposing circuit which contains  $u_2v_1$  and an edge in  $B^i$  then we can similarly choose the decomposing circuit such  $C^i$  has a bridge circuit K'' and a contributing accordion circuit between K and K''.

If  $C_{u_2v_1}^i$  is not contributing and does not have a decomposing circuit that contains the edge  $u_2v_1$  then there is a decomposing circuit K'' of  $C_{u_2v_1}^i$  that is a bridge circuit between  $F_i$  and  $B_i$ . Let  $u_3v_3$  and  $u_4v_4$  be the  $C_{u_2v_1}^i$  edges of K'' and assume with  $u_3v_3 \in F_i$  and  $u_4v_4 \in B_i$ . If K'' is chosen to minimize the sum of the distance along  $F_i$  from  $v_1$  to  $u_3$  plus the distance along  $B_i$  from  $v_4$  to  $u_2$  then the accordion circuit between K and K'' is contributing.

Hence there is either a contributing accordion or a trivial accordion between  $F_i$  and  $B_i$  with ends K and K''. Over all contributing accordions between  $F_i$  and  $B_i$ , choose the accordion H with ends K and K' such that the number of bridges in H is maximum. If k is a  $C^i$ -chord edge of K and k' is a  $C^i$ -chord edge of K' such that  $C_k^i$  and  $C_{k'}^i$  are disjoint, then the maximality of the number of bridges in H implies that  $C_k^i$  and  $C_{k'}^i$  are

contributing. The contributing accordion circuits of H together with the circuits  $C_k^i$  and  $C_{k'}^i$  are a decomposition of  $C^i$  which we call an *accordion decomposition* of  $C^i$ .

**Observation 6.25.** If D is a minimal weak odd k-double cycle in the directed graph G and  $k \ge 5$ , then every short circuit in D has an accordion decomposition.

We can now prove the main structural result of this section.

Proof of Theorem 6.22. Let P be a minimally non-Pfaffian series-parallel matroid pair. If G(P) has a neutral circuit C, then the matroid pair corresponding to the subgraph of G(P) induced by V(C) is non-Pfaffian, and thus by the minimality of P the neutral circuit is Hamiltonian. If G(P) does not have a neutral circuit, then Theorems 6.21 and 6.11 imply that G(P) has a weak odd double-cycle D. Let D be a weak odd double cycle in G(P) with cycle size k and suppose that  $k \geq 5$ . We show that D is not minimal.

Claim 1: The long circuits in D are contributing.

Proof. If  $C^F$  is not contributing, then let K be a decomposing circuit of  $C^F$  with  $C^F$ -edge arcs  $u_1v_1$  and  $u_2v_2$ . By relabeling the circuits in D, we may assume that  $u_1v_1 \in P_1 \cup F_1$ . Suppose  $u_1v_2$  is a chord of D. Since D has no forbidden long circuit chords and no forbidden short circuit chords,  $v_2$  is an internal vertex in the directed path in  $C^F$  from the head of  $P_{k-1}$  to the tail of  $P_1$  and since  $k \geq 5$ ,  $u_2v_1$  is not an edge in D. Since  $u_2v_1$ is directed from  $C^k$  or  $C^{k-1}$  to  $C^1$ , it is a forbidden long circuit chord in D. If  $u_1v_2$  is an edge of D, then it is the path  $B_k$  and thus  $u_2$  is again a vertex in  $C^{k-1}$  and  $u_2v_1$ is a forbidden long circuit chord. This contradicts the minimality of D and thus  $C^F$  is contributing. By symmetry  $C^B$  is also a contributing circuit and thus the long circuits in D are contributing.

For  $1 \leq i \leq k$ , let  $H_i$  be an accordion between  $F_i$  and  $B_i$  with a maximum number of bridges, and let  $C_i$  be the set of contributing circuits in the accordion decomposition of  $C^i$  from  $H_i$ . Note that the construction of the accordion is such that exactly one circuit in  $C_i$  intersects  $P_i$  and exactly one circuit in  $C_i$  intersects  $P_{i+1}$ . Let  $A_1 \in C_1$  and  $A_2 \in C_2$ be such that both  $A_1$  and  $A_2$  intersect  $P_2$ .

Claim 2: If  $A \in C_2$  is distinct from  $A_2$  then  $A \cup A_1$  is contributing.

Proof. If  $A \cup A_1$  is not contributing then there exists a decomposing circuit K in G with edges uv and  $u_1v_1$  such that  $uv \in A$  and  $u_1v_1 \in A_1$ . Suppose  $uv \in C^F$ . If  $u_1v_1 \in C^1$ then since  $C^F$  is contributing,  $u_1v_1 \in B_1$ . The chord  $u_1v$  is a forbidden between-circuit chord unless v is the second vertex on  $F_1$ . If v is the second vertex on  $F_1$  then since  $uv \in D$ , u must be the first vertex on  $F_1$ , but the first vertex on  $F_1$  is also the tail of  $P_2$ , contradicting the assumption that A is vertex disjoint from  $P_2$ . Hence if  $uv \in C^F$ then  $u_1v_1 \notin C^1$ . If  $u_1v_1 \notin C^1$  then let  $K_1$  be the bridge circuit between  $F_1$  and  $B_1$  such that  $u_1v_1$  is a  $C^1$ -chord edge of  $K_1$ . Since  $u_1v_1 \in A_1$  and  $A_1$  is a contributing circuit containing  $P_2$ , if  $u_1v_1 \in K_1$  then  $u_1 \in B_1$  and  $v_1 \in F_1$ . If  $uv \in F_2$  then the chord  $u_1v$  is a forbidden chord unless v is the second vertex of  $F_1$ , but this again leads to a contradiction of the assumption that A and  $P_2$  are vertex disjoint. If  $uv \in P_3$  then the two possible joining edges between K and  $K_1$  both contradict the assumption that the long circuits are contributing. Thus  $uv \notin C^F$  and by symmetry  $uv \notin C^B$ . There remain two cases two check: either uv is an edge in a bridge circuit of  $C^2$ , or  $C_{uv}^2$  is a contributing accordion circuit in  $\mathcal{C}_2$ .

Suppose uv is an edge in a bridge circuit  $K_2$  of  $C^2$  and assume  $u \in B_2$  and  $v \in F_2$ . Then  $u_1v$  is a forbidden chord unless  $u_1$  is an interior vertex in  $B_1$ , and  $uv_1$  is a forbidden chord unless  $v_1$  is an interior vertex in  $F_1$ . Thus  $u_1 \in B_1$  and  $v_1 \in F_1$  and there is a bridge circuit  $K_1$  in  $C^1$  such that  $u_1v_1$  is a  $C^1$ -chord edge of  $K_1$ . Let  $u_2v_2$  be the other  $C^1$ -chord edge in  $K_1$ . Since  $K_1$  and K share exactly one edge, either  $u_2v$  or  $uv_2$  is a joining edge between  $K_1$  and K. The former is a forbidden  $C^B$  chord and the latter a forbidden  $C^F$  chord, and thus if uv is an edge in the bridge circuit  $K_2$  then  $u \in F_2$  and  $v \in B_2$ . Since A is disjoint from  $P_2$ , u is an interior edge of  $F_2$  and v is an interior edge of  $B_2$ .

Suppose uv and  $u_3v_3$  are the  $C^2$ -chord edges in  $K_2$ . Since K and  $K_2$  share exactly one edge, either  $u_1v_3$  or  $u_3v_1$  is a joining edge between K and  $K_2$ . The chord  $u_1v_3$ is a forbidden short circuit chord if  $u_1 \in P_2$  and is a forbidden long circuit chord if  $u_1 \in P_1 \cup F_1$ . If  $u_1 \in B_1$  then since u is an interior vertex of  $F_1$  the vertex  $v_3$  is not the second vertex on  $F_2$  and thus  $u_1v_3$  is a forbidden between-circuit chord. Hence  $u_1v_3$  is not a joining edge of K and  $K_2$ . By symmetry the chord  $u_3v_1$  is forbidden and thus uvis not an edge in a decomposing circuit of  $C^2$ .

Suppose uv is such that  $C_{uv}^2$  is a contributing circuit in  $\mathcal{C}_2$  and assume that uv is a backwards arc of  $F^2$ . By the construction of the accordion decomposition of  $C^2$  there exists an edge  $u_4v_4 \in B_2$  and edges  $u_3v, uv_3$  in  $F_2$  such that  $C^2$  has bridge circuits  $K_2$  and  $K_3$  with  $u_4v_4, u_3v \in E(K_2)$  and  $u_4v_4, uv_3 \in E(K_3)$ . If  $u_1v$  is not a forbidden chord in D then  $u_1 \in B_2$  and  $u_3$  is the tail of  $P_2$ .

Suppose  $u_1v_1 \in C^1$  and first consider the case that  $u_3 = u_1$ . There is then a decomposing circuit K' through  $u_3, v_1, u$ , and  $v_4$  such that K' and  $K_3$  share the edge  $uv_4$ , and thus either  $u_3v_3$  or  $u_4v_1$  is a joining edge between K' and  $K_3$ . The former is a forbidden short circuit chord and the latter is a forbidden long circuit chord, and thus  $u_3$  and  $u_1$ are distinct, and  $u_1$  is an internal vertex in  $B_1$ . Let K' be the decomposing circuit with edges uv and  $u_4v_3$ . Then K' and K share the edge uv and the possible joining edges between K' and K are  $u_4v_1$  and  $u_1v_3$ . Both joining edges lead to contradictions: the chord  $u_4v_1$  is a forbidden long circuit chord and the chord  $u_1v_3$  is either a forbidden long circuit chord (when  $v_3$  is the tail of  $F_2$ ) or a forbidden between-circuit chord (when  $v_3$ is internal in  $F_2$ ). Thus  $u_1v_1$  is not an edge in  $C^1$  and it follows that  $u_1v_1$  is a  $C^1$ -chord edge of the bridge circuit  $K_1$  between  $B_1$  and  $F_1$ , where the  $C^1$ -chord edges of  $K_1$  are  $u_1v_1$  and  $u_2v_2$ .

The decomposing circuits  $K_1$  and K share the edge  $u_1v_1$ . and since  $u_2v$  is a forbidden

long circuit chord,  $uv_2$  is the unique joining edge between  $K_1$  and K. This is a contradiction, since this creates a decomposing circuit through uv and  $u_1v_4 \in C^1$  which we saw leads to a contradiction.

We conclude that uv is not a backwards arc of  $F^2$ , and by reversing the orientation of G we conclude that uv is not a backwards arc of  $B^2$ . Hence there is no decomposing circuit K between an edge of A and an edge of  $A_1$ . It follows that  $A_1 \cup A$  is a contributing circuit whenever  $A \in \mathcal{C}_2$  is distinct from  $A_2$ .

Claim 3: If  $A \in C_1$  is distinct from  $A_1$ , then  $A \cup A_2$  is contributing. *Proof.* This follows from Claim 2 by symmetry.

Claim 4: If  $A \in C_j$  and  $j \neq 1, 2$  then either  $A \cup A_1$  is contributing or  $A \cup A_2$  is contributing. Proof. Let  $A \in C_j$  for some  $j \neq 1, 2$ . If  $j \neq k$  and  $u_1, v_1 \in V(A_1)$  and  $u_j, v_j \in V(C^j)$  then either  $u_1v_j$  is a forbidden chord or  $u_jv_1$  is forbidden chord. Thus there is no decomposing circuit K with edges  $u_1v_1$  and  $u_jv_j$  and therefore there is no decomposing circuit with an edge in A and an edge in  $A_1$ . It follows that  $A \cup A_1$  is contributing. If j = k and  $u_2, v_2 \in A_2$  and  $u_j, v_j \in V(C^j)$  then either  $u_2v_j$  is a forbidden chord or  $u_jv_2$  is forbidden chord. Thus there is no decomposing circuit with an edge in  $A_2$  and edge in A and  $A \cup A_2$ is contributing.

Let  $D' = \{C^F, C^B, \mathcal{C}_1, \mathcal{C}_2, \dots, \mathcal{C}_k\}$  be a decomposition of D into contributing circuits. If  $C \in D'$  and  $P_2$  is not a subpath in C, then by Claims 2 and 3 either  $C \cup A_1$  is contributing or  $C \cup A_2$  is contributing, and thus there is a decomposition  $D^*$  of D into contributing sets  $\{C'_1, C'_2, \dots, C'_m\}$  such that  $V(P_2) \subseteq V(C'_i)$  for all i and

$$V(C'_1)\Delta V(C'_2)\Delta\cdots\Delta V(C'_M)=V(D)=\emptyset.$$

This contradicts the assumption that P is minimally non-Pfaffian. Hence if P is minimally non-Pfaffian and G(P) has no neutral cycles, then for some  $k \leq 4$ , G(P) has a spanning subgraph D that is an odd weak k-double cycle.

## Bibliography

- A. Ádám. Über zweipolige elektrische Netze. I. Magyar Tud. Akad. Mat. Kutató Int. Közl., 2:211–218, 1957.
- [2] J. A. Bondy and U. S. R. Murty. *Graph theory with applications*. American Elsevier Publishing Co., Inc., New York, 1976.
- [3] P. Camion. Caractérisation des matrices unimodulaires. Cahier du Centre d'Études de Recherche Opérationelle, 5:181–190, 1963.
- [4] W. J. Cook, W. H. Cunningham, W. R. Pulleyblank, and A. Schrijver. Combinatorial optimization. Wiley-Interscience Series in Discrete Mathematics and Optimization. John Wiley & Sons Inc., New York, 1998. A Wiley-Interscience Publication.
- [5] G. Cornuéjols. Combinatorial Optimization: Packing and Covering, volume 74 of CBMS-NSF Regional Conference Series in Applied Mathematics. Society for Industrial and Applied Mathematics (SIAM), Philadelphia, PA, 2001.
- [6] M. H. de Carvalho, C. L. Lucchesi, and U. S. R. Murty. The matching lattice. In *Recent advances in algorithms and combinatorics*, volume 11 of *CMS Books Math./Ouvrages Math. SMC*, pages 1–25. Springer, New York, 2003.
- G. A. Dirac. A property of 4-chromatic graphs and some remarks on critical graphs. J. London Math. Soc., 27:85–92, 1952.
- [8] R. J. Duffin. Topology of series-parallel networks. J. Math. Anal. Appl., 10:303–318, 1965.
- [9] J. Edmonds. Matroid partition. In Mathematics of the Decision Sciences, Part I (Seminar, Stanford, Calif., 1967), pages 335–345. Amer. Math. Soc., Providence, R.I., 1968.
- [10] J. Edmonds. Submodular functions, matroids, and certain polyhedra. In Combinatorial Structures and their Applications (Proc. Calgary Internat. Conf., Calgary, Alta., 1969), pages 69–87. Gordon and Breach, New York, 1970.

- [11] A. Galluccio and M. Loebl. On the theory of Pfaffian orientations. I. Perfect matchings and permanents. *Electron. J. Combin.*, 6:Research Paper 6, 18 pp. (electronic), 1999.
- [12] J. F. Geelen. Maximum rank matrix completion. Linear Algebra Appl., 288(1-3):211-217, 1999.
- [13] J. F. Geelen. An algebraic matching algorithm. *Combinatorica*, 20(1):61–70, 2000.
- [14] A. M. H. Gerards. A short proof of Tutte's characterization of totally unimodular matrices. *Linear Algebra Appl.*, 114/115:207–212, 1989.
- [15] B. Gerards. Unpublished.
- [16] C. D. Godsil. Algebraic combinatorics. Chapman & Hall, New York, 1993.
- [17] G. Hetyei. Rectangular configurations which can be covered by  $2 \times 1$  rectangles. *Pe'sci Tan. Foisk. Kozl.*, 8:351–367, 1964.
- [18] W. Hochstättler and J. Rieder. A note on the lattice of 2-matroid-intersections. Arch. Math. (Basel), 57(5):514–517, 1991.
- [19] A. J. Hoffman and J. B. Kruskal. Integral boundary points of convex polyhedra. In *Linear inequalities and related systems*, Annals of Mathematics Studies, no. 38, pages 223–246. Princeton University Press, Princeton, N. J.
- [20] P. W. Kasteleyn. Dimer statistics and phase transitions. J. Mathematical Phys., 4:287–293, 1963.
- [21] P. W. Kasteleyn. Graph theory and crystal physics. In Graph Theory and Theoretical Physics, pages 43–110. Academic Press, London, 1967.
- [22] C. H. C. Little. A characterization of convertible (0,1) matrices. J. Combin. Theory Ser. B, 18:187–208, 1975.
- [23] C. H. C. Little, F. Rendl, and I. Fischer. Towards a characterisation of Pfaffian near bipartite graphs. *Discrete Math.*, 244(1-3):279–297, 2002. Algebraic and topological methods in graph theory (Lake Bled, 1999).
- [24] L. Lovász. Ear-decompositions of matching-covered graphs. Combinatorica, 3(1):105–117, 1983.
- [25] L. Lovász. Some algorithmic problems on lattices. In Theory of algorithms (Pécs, 1984), volume 44 of Colloq. Math. Soc. János Bolyai, pages 323–337. North-Holland, Amsterdam, 1985.

- [26] L. Lovász. Matching structure and the matching lattice. J. Combin. Theory Ser. B, 43(2):187–222, 1987.
- [27] L. Lovász and M. D. Plummer. Matching theory, volume 121 of North-Holland Mathematics Studies. North-Holland Publishing Co., Amsterdam, 1986. Annals of Discrete Mathematics, 29.
- [28] W. McCuaig. Brace generation. J. Graph Theory, 38(3):124–169, 2001.
- [29] G. Nemhauser and L. Wolsey. Integer and combinatorial optimization. Wiley-Interscience Series in Discrete Mathematics and Optimization. John Wiley & Sons Inc., New York, 1999. Reprint of the 1988 original, A Wiley-Interscience Publication.
- [30] S. Norine, C. H. C. Little, and K. L. Teo. A new proof of a characterisation of Pfaffian bipartite graphs. J. Combin. Theory Ser. B, 91(1):123–126, 2004.
- [31] J. G. Oxley. *Matroid theory*. The Clarendon Press Oxford University Press, New York, 1992.
- [32] A. Recski. Matroid theory and its applications in electric network theory and in statics, volume 6 of Algorithms and Combinatorics. Springer-Verlag, Berlin, 1989.
- [33] J. Rieder. The lattices of matroid bases and exact matroid bases. Arch. Math. (Basel), 56(6):616-623, 1991.
- [34] J. Rieder. A note on the *f*-factor-lattice of bipartite graphs. J. Combin. Theory Ser. B, 56(1):90–96, 1992.
- [35] N. Robertson, P. D. Seymour, and R. Thomas. Permanents, Pfaffian orientations, and even directed circuits. Ann. of Math. (2), 150(3):929–975, 1999.
- [36] A. Schrijver. Combinatorial optimization. Polyhedra and efficiency. Vol. A,B,C, volume 24. Springer-Verlag, Berlin, 2003.
- [37] P. Seymour. On multi-colourings of cubic graphs and the conjectures of fulkerson and tutte. Proc. London Math. Soc., 38:423–460, 1979.
- [38] P. Seymour and C. Thomassen. Characterization of even directed graphs. J. Combin. Theory Ser. B, 42(1):36–45, 1987.
- [39] P. D. Seymour. Decomposition of regular matroids. J. Combin. Theory Ser. B, 28(3):305–359, 1980.
- [40] C. Thomassen. Even cycles in directed graphs. European J. Combin., 6(1):85–89, 1985.

- [41] K. Truemper. *Matroid decomposition*. Academic Press Inc., Boston, MA, 1992.
- [42] W. T. Tutte. Connectivity in matroids. Canad. J. Math., 18, 1966.
- [43] W.T. Tutte. Lectures on matroids. J. Res. Nat. Bur. Standards Sect. B, 69:1–47, 1965.
- [44] L. G. Valiant. The complexity of computing the permanent. *Theoret. Comput. Sci.*, 8(2):189–201, 1979.
- [45] V. V. Vazirani and M. Yannakakis. Pfaffian orientations, 0-1 permanents, and even cycles in directed graphs. *Discrete Appl. Math.*, 25(1-2):179–190, 1989. Combinatorics and complexity (Chicago, IL, 1987).
- [46] H. Whitney. On the abstract properties of linear dependence. American Journal of Mathematics, 1935.